

# ANNALES DE L'INSTITUT FOURIER

Ursula Ludwig

A Morse-Bott type complex and the Bismut-Zhang torsion for intersection cohomology

Tome 75, nº 2 (2025), p. 655-722.

https://doi.org/10.5802/aif.3648

Article mis à disposition par son auteur selon les termes de la licence CREATIVE COMMONS ATTRIBUTION - PAS DE MODIFICATION 3.0 FRANCE







### A MORSE-BOTT TYPE COMPLEX AND THE BISMUT-ZHANG TORSION FOR INTERSECTION COHOMOLOGY

by Ursula LUDWIG (\*)

ABSTRACT. — In the first part of this article we establish, for a compact pseudomanifold and a given perversity in the sense of Goresky and MacPherson a Morse theoretical cochain complex, which computes the intersection cohomology of the space. In the second part we use this cochain complex as well as the model Witten Laplacian to define the Bismut–Zhang torsion of a pseudomanifold. Conjecturally the Bismut–Zhang torsion will serve as the "topological" side in a Cheeger–Müller theorem for spaces with iterated conical singularities.

RÉSUMÉ. — Dans la première partie de cet article on construit, en utilisant la théorie de Morse, pour une pseudovariété stratifiée et une perversité au sens de la théorie de Goresky et MacPherson un complexe cohomologique. Ce complexe calcule la cohomologie d'intersection de la pseudovariété. Dans la deuxième partie on utilise ce complexe ainsi que le Laplacien de Witten pour définir la torsion de Bismut–Zhang, qui, conjecturellement, va servir dans un théorème de Cheeger–Müller pour des pseudovariétés à singularités coniques.

#### 1. Introduction

The famous Morse inequalities establish a relation between the singular cohomology of a smooth compact manifold M and the number of critical points of a smooth Morse function  $f: M \to \mathbb{R}$ . A way of proving the Morse inequalities is to show the existence of a combinatorial complex, which is generated by the critical points of f and whose cohomology is isomorphic to the singular cohomology of M. Such a complex has first been established

Keywords: Topological and analytic torsion, Morse–Bott complex, intersection cohomology.

<sup>2020</sup> Mathematics Subject Classification: 58J52, 37B30, 55N33.

<sup>(\*)</sup> The author is partly funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation) under Germany's Excellence Strategy EXC 2044–390685587, Mathematics Münster: Dynamics–Geometry–Structure.

by Thom [79] and Smale [78], using the unstable cell decomposition of M for the negative gradient flow of f with respect to a generic Riemannian metric. An analytic counterpart of the Morse–Thom–Smale complex has been suggested in [84]. Rigorous proofs of Witten's ideas have been given by Helffer and Sjöstrand [38] using semi-classical analysis. The Morse–Thom–Smale complex has seen a revival in the 1990s, starting with the work of Floer [22], where the idea of counting trajectories between critical points, has been exploited in an infinite dimensional setting.

An important generalisation of the Morse–Thom–Smale complex is to smooth Morse–Bott functions on a smooth compact manifold. Several approaches to the Morse–Bott complex can be found in the literature, most of them having Floer theoretical applications in mind [6, 7, 24, 29, 39, 41] (see [7, Section 1.2] for a comparison of the different approaches). In this paper we use the Morse–Bott cochain complex of Austin and Braam [6], defined in terms of the de Rham complexes of the critical submanifolds of the Morse–Bott function and the trajectory spaces between them.

The first aim of this article is to generalise the Morse–Thom–Smale complex to compact stratified pseudomanifolds and their intersection cohomology. First we introduce the notion of a stratified anti-radial gradient-like vector field on a compact abstract stratified space X. An important property of a stratified anti-radial gradient-like vector field  $\xi$  is that it can be lifted to a smooth vector field  $\tilde{\xi}$  on the smooth unfolding  $\tilde{X}$  of X. Singularities of a stratified anti-radial gradient-like vector field  $\xi$  are isolated, but the smooth vector field  $\widetilde{\xi}$  on  $\widetilde{X}$  has Morse–Bott singularities. This will allow us to adapt the construction in [6] and to construct, for any given perversity  $\bar{p}$  in the sense of Goresky and MacPherson [30, 31], a Morse–Bott type cochain complex  $(C^{\bullet}_{\bar{p}}(X,\xi),\partial^{\bullet})$  associated to the stratified anti-radial gradient-like vector field  $\xi$  on X, and which computes the intersection cohomology of X with perversity  $\bar{p}$ . The main idea of the construction is to replace the de Rham complex of critical submanifolds of  $\xi$  in the construction of the smooth Morse–Bott cochain complex of Austin and Braam [6] with the truncated complex of liftable intersection differential forms. The latter has been introduced by Brasselet, Hector and Saralegi in [11] and is a subcomplex of the more notorious complex of intersection differential forms of Brylinski, Goresky and MacPherson [14]. The main result regarding the Morse–Bott type complex, constructed in the first part of this article, is the following

MAIN THEOREM 1. — The Morse–Bott type complex  $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$  computes the intersection cohomology of X with perversity  $\bar{p}$ ,

$$H^{\bullet}(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet}) \simeq IH_{\bar{p}}^{\bullet}(X).$$

The Morse–Bott type complex, constructed in this article, generalises the constructions in [48] and in [49, 50]: In [48] a (homological) Morse–Thom–Smale complex computing singular homology (i.e., for a topologically normal space, intersection homology for the top perversity  $\bar{t}$ ) has been constructed for compact abstract stratified spaces. In [49, 50] a (homological) Morse–Thom–Smale complex computing intersection homology has been constructed for spaces with isolated singularities and any perversity  $\bar{p}$ . This latter complex (for the middle perversity) is an essential ingredient in the generalisation of the Cheeger–Müller theorem to singular spaces with isolated conical singularities achieved in [51].

While the Morse–Bott type complex constructed in the first part of this article may be interesting in itself, as already hinted above, our main motivation is its application to the comparison between analytic and topological torsion, aka Cheeger-Müller theorem, for spaces with conical singularities. The Cheeger-Müller theorem for a smooth compact manifold, proved independently by Cheeger [16] and Müller [60], is one of the most important comparison theorems in global analysis. It states the equality between topological and analytic torsion of a smooth compact manifold equipped with a unitary flat vector bundle. In [61] Müller extended the result to the case of odd dimensional manifolds and unimodular flat vector bundles. In the same time, in [9], Bismut and Zhang combined the Witten deformation and local index techniques to generalise the result of Cheeger and Müller to arbitrary flat vector bundles with arbitrary Hermitian metrics, thus giving the most general version of the theorem on smooth compact manifolds. A crucial role in Bismut and Zhang's proof is played by the duality between the Morse-Thom-Smale complex and the Witten complex.

The Cheeger–Müller theorem has also been extended to the equivariant setting ([10, 45, 46]) and for manifolds with boundary ([12, 13, 46, 83]). In recent years the study of analytic torsion and of the Cheeger–Müller theorem for spaces with cuspidal singularities and locally symmetric spaces has seen a lot of activity (see e.g. [2, 3, 15, 55, 56, 62, 63, 64, 65, 68, 69, 70, 72]), development which has been in part motivated by the study of the torsion of the cohomology of locally symmetric spaces associated to arithmetic groups [8].

Let us now review in more detail the history of the study of analytic and topological torsion for spaces with *conical singularities*: It started with work of Dar [19], who proved that the analytic torsion is well-defined for spaces with isolated conical singularities. Dar also defined the (topological) Reidemeister torsion in the context of intersection homology. In case of an oriented even dimensional space, by Poincaré duality, both the analytic torsion as well as the intersection Reidemeister torsion with middle perversity are trivial.

Apart from the easy even dimensional case, the Cheeger–Müller theorem has resisted attempts of generalisation to spaces with conical singularities for a while. Note that, since the analytic torsion of a singular space is not a topological invariant in general, while, by a result of Dar, the intersection Reidemeister torsion is, an equality between the two cannot be expected in general.

An approach to the Cheeger–Müller theorem for spaces with isolated conical singularities suggested by Lesch [43, Problem 5.3] (see also [44], where this strategy has been explained in more generality) is to reduce the problem via the gluing formula for analytic torsion (see [12, 13, 44, 83]) to a comparison of torsions on a truncated cone. The analytic torsion of a truncated cone has been studied in [35, 36, 66, 82]. The recent preprint [37] seems to complete Lesch's programme, without however giving an interpretation of the analytic correction terms coming from the singularities.

The well-definedness of analytic torsion for spaces with wedge singularities, i.e. conical spaces with a single singular stratum of dimension  $\geq 1$ , is due to Mazzeo and Vertman [57]. In [4], Albin, Rochon and Sher establish an extension of the Cheeger–Müller theorem to spaces with wedge singularities with even codimensional singular stratum and unimodular flat vector bundles satisfying a strong acyclicity condition at infinity; this excludes the trivial bundle. The strategy in [4] is to study the analytic torsion via degeneration of smooth metrics into conical metrics. This strategy has been applied previously with success by the three authors for singular spaces with cuspidal singularities [2, 3].

In [51] the author has followed yet a third strategy for extending the Cheeger–Müller theorem to singular spaces with conical singularities, namely by adapting the approach of Bismut and Zhang to the singular situation. The main result in [51] is a comparison between the analytic torsion of a singular space with isolated conical singularities and a torsion, which, in the present article, will be called the Bismut–Zhang torsion.

The present article is part of the author's programme to extend the approach in [51] to general stratified spaces with iterated conical singularities. In the second part of this article we define the Bismut–Zhang torsion, which conjecturally will serve as the "topological side" in a Cheeger–Müller theorem for stratified spaces. There are two ingredients entering the definition of the Bismut–Zhang torsion: the Morse–Bott type cochain complex for the lower middle perversity  $\overline{m}$  constructed in the first part of this article and the model Witten Laplacian appearing in the generalisation of the Witten deformation to stratified spaces in [49]. For clarity of the presentation, we only define the Bismut–Zhang torsion for spaces with two strata here. However, since both ingredients entering the definition of the Bismut–Zhang torsion are available in general, this definition can be extended directly to stratified pseudomanifolds of arbitrary depth (see Remark 6.4(b)).

The Bismut–Zhang torsion is not a topological invariant in general. However, we can prove the following

MAIN THEOREM 2. — Let X be an oriented compact pseudomanifold with two strata Y < Z such that  $\operatorname{codim}_X Y$  is even. Then the Bismut–Zhang torsion is a topological invariant. Moreover, the Bismut–Zhang torsion equals the intersection Reidemeister torsion defined by Aparna Dar [19].

The Main Theorem 2 can be seen as the topological counterpart of a result in [57], which states that the analytic torsion on a wedge space with both strata of odd dimension is independent of the choice of a conical metric.

The article is organised as follows: In Section 2, to keep the article self-contained, we collect those basics on abstract stratified spaces, aka Thom–Mather stratified spaces, and on the intersection (co-)homology of pseudomanifolds as introduced by Goresky and MacPherson, which are used in the course of the article. In Section 3 we introduce the notion of a stratified anti-radial gradient-like vector field on an abstract stratified space. In Section 4 we define, for a stratified pseudomanifold X, a stratified anti-radial gradient-like vector field and a given perversity  $\bar{p}$  in the sense of Goresky and MacPherson, the Morse–Bott type cochain complex advertised in the title, and prove the Main Theorem 1. In Section 5, adapting ideas from smooth Morse and Morse–Bott theory, we construct a perturbation of the Morse–Bott complex for a space with two strata. In the last section, Section 6, we define the Bismut–Zhang torsion for a stratified pseudomanifold with two strata, we recall the definition of intersection Reidemeister torsion from [19], and prove the Main Theorem 2.

#### Acknowledgements

I would like to thank Claire Debord, Andrew du Plessis, Clint McCrory, Jörg Schürmann and David Trotman for discussions on stratified spaces. I would also like to express my gratitude to Jean-Michel Bismut, for supporting from the start my project on the study of torsion on singular spaces via the Witten deformation. I thank the anonymous referee for their helpful comments. Final corrections of this article have been done during the authors stay at the Max-Planck Institute for Mathematics in Bonn and the author thanks the institute for its hospitality and the excellent working conditions.

#### 2. Preliminaries and Notation

In this article we use the language of abstract stratified spaces and control data as introduced by Thom [80] and Mather [53] (see also the reprint [54] of Mather's notes). A useful notion for carrying out analysis on singular spaces is that of a manifold with corners and iterated fibration structures as introduced by Melrose. For a dictionary between the two languages we refer the reader to [1, Section 6] and [21, Section 1].

Section 2 is organised as follows: In Section 2.1, for convenience of the reader, we recall some basic definitions from the theory of abstract stratified spaces. We also recall the notion of "déplissage élémentaire" (translated here as "unfolding") of a stratified space as introduced in [11, Section A.II].

In Section 2.2 we recall the definition and the local calculation for the intersection (co-)homology of a stratified pseudomanifold as introduced by Goresky and MacPherson [30, 31]. We also recall the complex of intersection differential forms introduced by Brylinksi in [14, Section 1.2], where it is attributed to Goresky and MacPherson. In this article we use the subcomplex of liftable intersection differential forms, which has been introduced in [11, Section B.IV] and which we also recall.

#### 2.1. Stratified spaces and unfoldings

## 2.1.1. Abstract stratified spaces (aka Thom–Mather stratified spaces). Controlled maps

Let X be a topological space, Hausdorff, locally compact and with countable basis for its topology. A stratification S of the topological space X is

a locally finite family of pairwise disjoint locally closed subsets  $Z\subset X$ , called strata, such that  $X=\bigcup_{Z\in\mathcal{S}}Z$ . The strata Z are smooth manifolds without boundary in the induced topology. We assume that the *frontier condition* is satisfied: For each pair of strata  $(Z,R),\,Z\cap\overline{R}\neq\emptyset$  implies that  $Z\leqslant R$ , i.e.  $Z\subset\overline{R}$ .

A tubular system for the stratification S is a family of tuples  $\{(T_Z, \pi_Z, \rho_Z)\}_{Z \in S}$ , where  $T_Z$  is an open neighbourhood of Z in X,  $\pi_Z$ :  $T_Z \to Z$  is a continuous retraction and  $\rho_Z : T_Z \to \mathbb{R}_{\geqslant 0}$  is continuous and satisfies  $\rho_Z^{-1}(0) = Z$ . We call a tubular system  $\{(T_Z, \pi_Z, \rho_Z)\}_{Z \in S}$  control data for the stratification S if the following conditions hold:

- (a) For each pair of strata (Z, R),  $T_Z \cap R \neq \emptyset$  implies that  $Z \leqslant R$ .
- (b) For each pair of strata (Z,R) with Z < R (i.e.  $Z \le R, Z \ne R$ ) the map

$$(2.1) (\pi_Z, \rho_Z): R \cap T_Z \longrightarrow Z \times \mathbb{R}_{>0}$$

is smooth and submersive.

(c) The tubular system is controlled, i.e. for each pair of strata (Z, R) with Z < R and for all  $x \in T_Z \cap T_R$  the following conditions are satisfied:  $\pi_R(x) \in T_Z$  and

(C1) 
$$\pi_Z(\pi_R(x)) = \pi_Z(x),$$

(C2) 
$$\rho_Z(\pi_R(x)) = \rho_Z(x).$$

DEFINITION 2.1. — An abstract stratified space is a topological space X as above with a stratification S and control data  $\{(T_Z, \pi_Z, \rho_Z)\}_{Z \in S}$ .

Let X, Y be abstract stratified spaces with stratifications  $\mathcal{S}$  resp.  $\mathcal{S}'$ . A continuous map  $g: X \to Y$  is called a stratified map, if for every  $Z \in \mathcal{S}$  there exists  $W \in \mathcal{S}'$  such that  $g(Z) \subset W$  and moreover  $g(T_Z) \subset T_W$ . We consider the following control conditions:

(C3) 
$$\pi_W(g(x)) = g(\pi_Z(x)), \quad x \in T_Z,$$

(C4) 
$$\rho_Z(x) = \rho_W(g(x)), \qquad x \in T_Z.$$

A stratified map  $g: X \to Y$  is called weakly controlled (resp. controlled) if for every  $Z \in \mathcal{S}$  the restriction  $g_{|Z}$  is smooth and the control condition (C3) (resp. the control conditions (C3) and (C4)) hold. A bijective controlled map  $g: X \to Y$  is an isomorphism of stratified spaces if g sends each stratum Z of X diffeomorphically to a stratum of Y and moreover  $g(T_Z) = T_{g(Z)}$ . A controlled map  $g: X \to M$  into a smooth manifold M is called a controlled submersion, if the restriction to each stratum is submersive.

The depth of a stratum Z, denoted by depth(Z), is the integer

(2.2) depth(Z) := 
$$\sup\{l \mid \exists \text{ a sequence of strata } Z = R_0 < \dots < R_l\}.$$

The depth of the stratified space X is defined as

(2.3) 
$$\operatorname{depth}(X) := \sup \{ \operatorname{depth}(Z) \mid Z \in \mathcal{S} \}.$$

The reader is warned that, for the notion of depth, there are different conventions in the literature.

Let X be a stratified space with  $\operatorname{depth}(X) = d$ . By shrinking the tubes  $\{T_Z\}_{Z \in \mathcal{S}}$ , we can assume that two strata of the same depth have disjoint tubular neighbourhoods (see [54, page 492]). For inductive arguments it is convenient to work with the stratification of X by depth (associated to  $\mathcal{S}$ ): The strata are given by  $Z_i := \bigcup_{\operatorname{depth}(Z) = d - i} Z$ ,  $i = 0, \ldots, d$ . By taking the union of tubes, we get control data  $\{(T_i, \pi_i, \rho_i)\}_{i=0,\ldots,d-1}$ .

In the following we assume that X is connected. Hence  $Z_0 < \cdots < Z_d$  and  $X = \overline{Z_d}$ .

#### 2.1.2. Local triviality of the tubular neighbourhood $T_i$

Let  $i \in \{0, ..., d-1\}$ . For  $x \in Z_i$  and  $\epsilon > 0$  small enough, the spaces  $T_i^{\epsilon} := \{w \in T_i \mid \rho_i(w) < \epsilon\}$ ,  $S_i := \rho_i^{-1}(\epsilon)$  and  $L_x := \pi_i^{-1}(x) \cap \rho_i^{-1}(\epsilon)$  are abstract stratified spaces, inheriting stratification and control data from those of X, and not depending (up to isomorphism of stratified spaces) on the choice of  $\epsilon$ . The space  $L_x$  is called the *link* of X at x, it only depends on the connected component of  $Z_i$  containing x. We denote by  $cL_x := (L_x \times [0,\infty))/_{(y,0)\sim (w,0)}$  the cone over  $L_x$ , it is also a stratified space.

The maps  $\pi_i: T_i^{\epsilon} \to Z_i$  and  $\pi_{i|S_i}: S_i \to Z_i$  are proper controlled submersions and therefore, by Thom's First Isotopy Lemma (see e.g. [81, Theorem 2.6]), locally trivial stratified fibre bundles. We denote by

$$(2.4) c_{\epsilon}(S_i, \pi_i) := \left( (S_i \times [0, \epsilon)) \sqcup (Z_i \times \{0\}) \right) / (c, 0) \sim (\pi_i(c), 0),$$

the topological mapping cylinder of  $\pi_i$ . The mapping cylinder  $c_{\epsilon}(S_i, \pi_i)$  inherits the structure of an abstract stratified space (see [81, Section 5.3.6] for more details) and there is an isomorphism of stratified spaces

(2.5) 
$$\alpha_i: c_{\epsilon}(S_i, \pi_i) \longrightarrow T_i^{\epsilon}.$$

By composition of  $\alpha_i$  with the canonical quotient map  $S_i \times (-\epsilon, \epsilon) \rightarrow c_{\epsilon}(S_i, \pi_i), (c, r) \mapsto [c, |r|],$  we get a map

(2.6) 
$$\beta_i: S_i \times (-\epsilon, \epsilon) \longrightarrow T_i^{\epsilon}.$$

We denote by  $(-\epsilon, \epsilon)^* := (-\epsilon, \epsilon) \setminus \{0\}$ . The restriction  $\beta_{i|S_i \times (-\epsilon, \epsilon)^*} : S_i \times (-\epsilon, \epsilon)^* \to T_i^{\epsilon} \setminus Z_i$  is a twofold covering of stratified spaces.

#### 2.1.3. Unfolding of X

In this section we recall the notions of déplissage élémentaire and déplissage as introduced in [11, Section A.II], and which are closely related to the notions of decomposition and total decomposition introduced by Verona (see [81, Sections 6.1.1 and 6.6.1]). Following [20] we translate déplissage élémentaire by unfolding in this article; other common translations in the literature are "doubling construction", "resolution" or "blow-up".

We assume that X as well as all links are connected.

To a fixed isomorphism of stratified spaces  $\alpha_0: c_{\epsilon}(S_0, \pi_0) \to T_0^{\epsilon}$  as in (2.5) (and the associated map  $\beta_0: S_0 \times (-\epsilon, \epsilon) \to T_0^{\epsilon}$  as in (2.6)) we define the unfolding  $\widetilde{X}^0$  of X along its minimal stratum  $Z_0$  as follows:  $\widetilde{X}^0$  is the quotient of

$$(2.7) \qquad ((X \setminus Z_0) \times \{-1, 1\}) \sqcup (S_0 \times (-\epsilon, \epsilon))$$

by the equivalence relation:

(2.8) 
$$(x,j) \sim (c,r) \text{ if } |r| = jr \text{ and } x = \beta_0(c,r).$$

There is a natural folding map  $\theta_0: \widetilde{X}^0 \to X$ . The space  $\widetilde{X}^0$  inherits the structure of an abstract stratified space of  $\operatorname{depth}(\widetilde{X}^0) = \operatorname{depth}(X) - 1$  (see [11, Proposition 3.1]). The restriction of the map  $\theta_0$  to  $\widetilde{X}^0 \setminus \theta_0^{-1}(Z_0)$  is a twofold stratified covering of  $X \setminus Z_0$ . Moreover, we have a natural isomorphism of stratified spaces  $\theta_0^{-1}(Z_0) \cong S_0$ .

Iterating the above unfolding procedure we get a sequence of unfoldings:

$$(2.9) \qquad \widetilde{X} := \widetilde{X}^{d-1} \xrightarrow{\theta_{d-1,d-2}} \widetilde{X}^{d-2} \xrightarrow{\theta_{d-2,d-3}} \cdots \xrightarrow{\theta_{1,0}} \widetilde{X}^0 \xrightarrow{\theta_0} X.$$

We denote by  $S^l := \{Z_{l+1}^l < \dots < Z_d^l\}$  the stratification by depth of  $\widetilde{X}^l$ ,  $l = 0, \dots, d-1$ .

We call  $\widetilde{X} := \widetilde{X}^{d-1}$  the total unfolding of X (déplissage in [11]), it is a smooth manifold.

We also get unfoldings of  $T_i$ , i = 0, ..., d-1, as well as of all links  $L_x$ ,  $x \in \overline{Z_{d-1}}$ , which are compatible with the unfoldings of X, i.e. the inclusion  $T_i \subset X$  induces an inclusion  $\widetilde{T}_i \subset \widetilde{X}$ , etc.

For  $l \in \{1, ..., d-1\}$ , we denote by  $\theta_{l,l-1} : \widetilde{X}^l \to \widetilde{X}^{l-1}$  the *l*-th step in the sequence of unfoldings (2.9) and by

(2.10) 
$$\theta_l: \widetilde{X}^l \longrightarrow X$$
 the composition  $\theta_l = \theta_{l,l-1} \circ \ldots \circ \theta_{1,0} \circ \theta_0$ .

For  $i, j \in \{0, \dots, d-1\}$ , i < j, we denote by

(2.11) 
$$\theta_{j,i}: \widetilde{X}^j \longrightarrow \widetilde{X}^i$$
 the composition  $\theta_{j,i}:=\theta_{j,j-1} \circ \cdots \circ \theta_{i+1,i}$ .

We denote by  $\theta = \theta_{d-1} : \widetilde{X} \to X$  the map from the total unfolding  $\widetilde{X}$  to X. By construction  $\widetilde{X} \setminus \theta^{-1}(\overline{Z_{d-1}})$  has  $2^d$  connected components and  $\theta_{|\widetilde{X}\setminus \theta^{-1}(\overline{Z_{d-1}})} : \widetilde{X} \setminus \theta^{-1}(\overline{Z_{d-1}}) \to X \setminus \overline{Z_{d-1}} = Z_d$  is a covering map. We fix one of these connected components and denote its closure in  $\widetilde{X}$  by  $\widehat{X}$ . We call  $\widehat{X}$  the marked leaf of  $\widetilde{X}$ , it is a smooth manifold with corners and iterated fibration structure (see [21, Section 1]).

Let  $i \in \{0, ..., d\}$ ,  $l \in \{0, ..., d-1\}$  and  $x \in Z_i$ . The fibre  $\theta_l^{-1}(x)$  is one of the following:

- a discrete set of  $2^{l+1}$  points, if l < i,
- a disjoint union of  $2^l$  connected components isomorphic to  $L_x$ , if l=i.
- a disjoint union of  $2^i$  connected components isomorphic to the (l-i-1)-th unfolding of  $L_x$ , if l>i.

#### 2.1.4. Distinguished neighbourhoods

For each point  $x \in Z_i$ , i = 0, ..., d - 1, there is an open neighbourhood  $U_x$  and a homeomorphism of stratified spaces

(2.12) 
$$\psi_x: U_x \simeq \mathbb{R}^{\dim Z_i} \times cL_x,$$

which is compatible with the isomorphisms (2.5), (2.6); here dim  $Z_i$  denotes the dimension of the connected component of  $Z_i$  containing x. We call  $U_x$  a distinguished neighbourhood and denote by  $(z, \varphi, r)$  the local coordinates in  $U_x$ . The coordinate change between two distinguished charts is of the form  $(z, \varphi, r) \mapsto (z, h(z)(\varphi), r)$ , where h(z) is a stratified automorphism of the link  $L_x$  (see [11, Remark A.I.3]).

### 2.2. Intersection (co-)homology and intersection differential forms

In this article the intersection (co-)homology introduced by Goresky and MacPherson [30, 31] plays an important role. To be able to define it, we will assume in Sections 4-6, that the stratified space X is moreover a pseudomanifold of dimension n, i.e. if we denote by  $X_i := \bigcup_{\dim Z \leq i} Z$ , then  $X = X_n, X_{n-1} = X_{n-2}$  and  $X^{sm} := X \setminus X_{n-2}$  is open and dense in X. We

call  $\Sigma := X_{n-2}$  the singular set of X. Note that for X connected,  $X^{sm} = Z_d$  and  $\Sigma = \overline{Z_{d-1}}$ .

When speaking of a stratified pseudomanifold X, in this article, we will always mean an abstract stratified space in the sense of Definition 2.1, which we assume in addition to be an orientable, compact, connected pseudomanifold. We moreover assume that all links are connected.

If X has only two strata and the singular set  $\Sigma$  is a manifold of dimension 0, we call X a space with isolated singularities.

#### 2.2.1. Intersection (co-)homology

Let  $\bar{p} = (p_2, \ldots, p_n)$  be a perversity function in the sense of Goresky and MacPherson, i.e.  $p_i$  are integers with  $p_2 = 0$  and  $p_j \leqslant p_{j+1} \leqslant p_j + 1$  for all  $2 \leqslant j \leqslant n-1$ . A subspace  $Y \subset X$  is  $(\bar{p}, i)$ -allowable if  $\dim(Y) \leqslant i$  and  $\dim(Y \cap X_{n-k}) \leqslant i - k + p_k$  for  $k \geqslant 2$ .

Let  $(IC^{\bar{p}}_{\bullet}(X,\mathbb{Z}),\partial_{\bullet})$  be the subcomplex of the complex of PL geometric chains of X consisting of intersection chains of perversity  $\bar{p}$ , i.e.  $\sigma \in IC^{\bar{p}}_i(X,\mathbb{Z})$  iff  $|\sigma|$  is  $(\bar{p},i)$ -allowable and  $|\partial\sigma|$  is  $(\bar{p},i-1)$ -allowable. The intersection homology of X with perversity  $\bar{p}$  is defined as the homology of the intersection chain complex  $(IC^{\bar{p}}_{\bullet}(X,\mathbb{Z}),\partial_{\bullet})$ ,

$$(2.13) IH^{\bar{p}}_{\bullet}(X,\mathbb{Z}) := H_{\bullet}(IC^{\bar{p}}_{\bullet}(X,\mathbb{Z}), \partial_{\bullet}).$$

The intersection cohomology  $IH_{\bar{p}}^{\bullet}(X,\mathbb{Z})$  is defined as the cohomology of the dual complex. The intersection homology  $IH_{\bullet}^{\bar{p}}(X)$  (resp. the intersection cohomology  $IH_{\bar{p}}^{\bullet}(X)$ ) with real coefficients is defined in the same way, working with intersection chains with real coefficients instead. For  $U \subset X$  open, one can also define the relative intersection homology  $IH_{\bullet}(X,U)$  (see e.g. [40, Section 4.6] for more details). For a stratified pseudomanifold X with boundary  $\partial X$ ,  $IH_{\bullet}(X,\partial X)$  denotes the intersection homology of the interior of X modulo a collared neighbourhood of  $\partial X$  (see [32, Section 1.4]).

Let L be a stratified pseudomanifold of dim L = n - 1. For convenience of the reader we recall the local calculation for the intersection homology resp. for the relative intersection homology of a cone cL (see [31, Section 2.2] and also [40, Proposition 4.7.2]):

(2.14) 
$$IH_{i}^{\bar{p}}(cL) = \begin{cases} IH_{i}^{\bar{p}}(L) & \text{for } i \leq n - p_{n} - 2, \\ 0 & \text{for } i > n - p_{n} - 2, \end{cases}$$

resp.

(2.15) 
$$IH_i^{\bar{p}}(cL, L) := IH_i^{\bar{p}}(cL, cL \setminus \{0\}) = \begin{cases} IH_{i-1}^{\bar{p}}(L) & \text{for } i \geqslant n - p_n, \\ 0 & \text{for } i < n - p_n. \end{cases}$$

The formulas (2.14) and (2.15) also hold with coefficients in  $\mathbb{Z}$ . Two perversities  $\bar{p}$ ,  $\bar{q}$  are called complementary if

$$(2.16) \bar{p} + \bar{q} = \bar{t} := (0, 1, 2, \dots, n-2).$$

In Section 6 of this article we will mainly focus on the lower middle perversity  $\overline{m}$ ,  $m_k = \lfloor k/2 \rfloor - 1$ . Its complementary perversity is the upper middle perversity  $\overline{n}$ ,  $n_k = \lfloor (k-1)/2 \rfloor$ .

#### 2.2.2. Intersection differential forms

Let M and N be smooth manifolds with dim M=m, dim N=n and let  $\pi:M\to N$  be a smooth submersion. A smooth j-form  $\omega\in\Omega^j(M)$  has perversity  $q\in\{0,\ldots,m-n\}$  with respect to  $\pi$  if for every (q+1)-tuple of smooth vector fields  $v_0,\ldots,v_q$  tangent to the fibres of  $\pi$  (i.e. smooth sections of  $\ker d\pi$ ), we have

$$(2.17) \iota_{v_0} \cdots \iota_{v_q} \omega = 0.$$

In other words, if the submersion  $\pi: M \to N$  is represented locally as the projection  $(z_1, \ldots, z_n, \varphi_1, \ldots, \varphi_{m-n}) \mapsto (z_1, \ldots, z_n)$ , then a form of perversity q is a linear combination of forms  $dz_I \wedge d\varphi_J$ , where I and J are multi-indices with  $|J| \leq q$ .

Let now X be a stratified pseudomanifold of dimension n according to our convention at the beginning of Section 2.2. We denote by  $(\Omega^{\bullet}(X), d)$  the de Rham complex of smooth forms on  $X^{sm} = X \setminus \Sigma$ . Let  $\bar{p}$ ,  $\bar{q}$  be two complementary perversities.

#### Definition 2.2.

- (a) A smooth differential j-form  $\omega \in \Omega^j(X)$  has perversity  $\bar{q}$  if for all strata Z the form  $\omega_{|T_Z\setminus\Sigma}$  has perversity  $q_{\operatorname{codim} Z}$  with respect to the submersion  $\pi_Z: T_Z\setminus\Sigma\to Z$ .
- (b) A form  $\omega \in \Omega^{j}(X)$  is an intersection differential form of perversity  $\bar{q}$  if both  $\omega$  and  $d\omega$  have perversity  $\bar{q}$ . We denote by  $(\Omega^{\bullet}_{\bar{q}}(X), d)$  the complex of intersection differential forms of perversity  $\bar{q}$ .

The complex  $(\Omega_{\bar{q}}^{\bullet}(X), d)$  computes the intersection cohomology of X with perversity  $\bar{p}$  (see [14, Section 1.2]),  $H^{\bullet}(\Omega_{\bar{q}}^{\bullet}(X), d) \simeq IH_{\bar{p}}^{\bullet}(X)$ .

#### 2.2.3. Intersection differential forms admitting a lift

We denote by  $(\Omega^{\bullet}(\widetilde{X}), d)$  the de Rham complex of smooth differential forms on the smooth manifold  $\widetilde{X}$ . We denote by  $(\Omega^{\bullet}_{lift}(X), d) \subset (\Omega^{\bullet}(X), d)$ 

the subcomplex of smooth forms  $\omega$  on the regular stratum  $X^{sm}$  admitting a lift  $\widetilde{\omega} \in \Omega^{\bullet}(\widetilde{X})$ , i.e.  $(\theta_{|\theta^{-1}(X^{sm})})^*\omega = \widetilde{\omega}_{|\theta^{-1}(X^{sm})}$ , where by  $\widetilde{\omega}_{|\theta^{-1}(X^{sm})}$  we denote the pull back of  $\widetilde{\omega}$  via the inclusion map  $\theta^{-1}(X^{sm}) \subset \widetilde{X}$ . Note that, by continuity of  $\widetilde{\omega}$  and density of  $\theta^{-1}(X^{sm})$  in  $\widetilde{X}$ , the lift  $\widetilde{\omega}$  of a form  $\omega \in \Omega^{\bullet}_{\text{lift}}(X)$  is unique.

DEFINITION 2.3. — We denote by  $(\mathcal{K}^{\bullet}_{\bar{q}}(X), d) \subset (\Omega^{\bullet}_{\bar{q}}(X), d)$  the subcomplex of intersection differential forms  $\omega$  of perversity  $\bar{q}$  which admit a lift, i.e.  $\omega \in \Omega^{\bullet}_{\mathrm{lift}}(X)$ .

By associating to each form  $\omega \in \mathcal{K}^{\bullet}_{\overline{q}}(X)$  its lift  $\widetilde{\omega} \in \Omega^{\bullet}(\widetilde{X})$ , the complex  $(\mathcal{K}^{\bullet}_{\overline{q}}(X), d)$  can be seen as a subcomplex of the de Rham complex  $(\Omega^{\bullet}(\widetilde{X}), d)$ . The complex  $(\mathcal{K}^{\bullet}_{\overline{q}}(X), d)$  computes the intersection cohomology of X with perversity  $\overline{p}$  (see [11, page 212]),

$$(2.18) H^{\bullet}(\mathcal{K}_{\bar{q}}^{\bullet}(X), d) \simeq IH_{\bar{p}}^{\bullet}(X).$$

Let L be a stratified pseudomanifold of dimension n-1. We denote by  $(\mathcal{K}^{\bullet}_{\overline{q},tr}(L),d)$  the following subcomplex of  $(\mathcal{K}^{\bullet}_{\overline{q}}(L),d)$ :

(2.19) 
$$\mathcal{K}_{\bar{q},tr}^{i}(L) = \begin{cases} \mathcal{K}_{\bar{q}}^{i}(L) & \text{if } i < q_{n}, \\ \text{ker } d & \text{if } i = q_{n}, \\ 0 & \text{if } i > q_{n}. \end{cases}$$

By [11, Proposition C.IV.2.3 and page 212],

(2.20) 
$$H^{\bullet}(\mathcal{K}_{\bar{q},tr}^{\bullet}(L),d) \simeq IH_{\bar{p}}^{\bullet}(cL).$$

LEMMA 2.4. — Let  $\omega \in \Omega^{\bullet}(X)$  be a differential form admitting a lift  $\omega_0 \in \Omega^{\bullet}(\widetilde{X}^0)$  to  $\widetilde{X}^0$ , i.e.  $\omega_{0|\theta_0^{-1}(X^{sm})} = (\theta_{0|\theta_0^{-1}(X^{sm})})^*\omega$ .

- (a) If  $\omega$  has perversity  $\bar{q}$ , then  $\omega_0$  also has perversity  $\bar{q}$ .
- (b) If  $\omega_0$  has perversity  $\bar{q}$ , then  $\omega$  satisfies the perversity condition for the perversity  $\bar{q}$  and all non-minimal strata of X.

Proof. — To prove that a form  $\alpha \in \Omega^{\bullet}(\widetilde{X}^0)$  has perversity  $\overline{q}$ , by continuity, it is enough to check the perversity condition (2.17) on the dense subset  $\widetilde{X}^0 \setminus \theta^{-1}(Z_0)$  of  $\widetilde{X}^0$ . The claim of the proposition follows from the fact that the restriction of the folding map  $\theta_0$  to a connected component of  $\widetilde{X}_0 \setminus \theta_0^{-1}(Z_0)$  is an isomorphism of stratified spaces.

#### 3. Stratified anti-radial gradient-like vector fields

In this section we introduce the notion of a stratified anti-radial gradient-like vector field on an abstract stratified space X with unfolding  $\widetilde{X}$ . We

assume in this section, that the abstract stratified space X is compact and connected.

Section 3 is organised as follows: In Section 3.1 we first recall some standard notions on stratified vector fields: the notion of a controlled stratified vector field is an important tool in Thom and Mather's theory; the notion of a totally anti-radial vector field is inspired from the radial vector fields as introduced by Marie-Hélène Schwartz [75] (for Whitney stratified spaces) in her study of characteristic classes for singular spaces. In Section 3.2 we introduce the notion of unfolding of a stratified vector field. In Section 3.3 we introduce the notion of a stratified anti-radial gradient-like vector field and study its main properties. The weaker notion of a stratified gradient-like vector field has been introduced and studied in [48]. The main additional feature of a stratified anti-radial gradient-like vector field is that it can be unfolded to a Morse–Bott vector field on the unfolding  $\widetilde{X}$ . This unfolding property will be crucial for the construction of the "singular" Morse–Bott type cochain complex figuring in the title of this article.

#### 3.1. Controlled and totally anti-radial stratified vector fields

A stratified vector field  $\xi$  on an abstract stratified space X is a family  $\{\xi_Z: Z \to TZ\}_{Z \in \mathcal{S}}$  of smooth vector fields.

Definition 3.1.

(a) A stratified vector field on X is called controlled if for all pairs of strata (Z,R) with Z < R and all  $x \in T_Z \cap R$  the following control conditions are satisfied:

(C5) 
$$d\pi_Z \xi_R(x) = \xi_Z(\pi_Z(x)),$$

(C6) 
$$d\rho_Z \xi_R(x) = 0.$$

A stratified vector field satisfying only the control condition (C5) is called weakly controlled.

(b) A stratified vector field is called totally anti-radial if for each stratum Z there exists a bounded non-negative function  $A_Z: Z \to [0,\infty)$ , such that for all pairs of strata (Z,R) with Z < R and all  $x \in T_Z$ ,

$$(3.1) |d\rho_Z \xi_R(x)| \leqslant A_Z(\pi_Z(x))\rho_Z(x) \text{ and } d\rho_Z \xi_R(x) \leqslant 0.$$

There is no canonical notion of continuity for stratified vector fields on an abstract stratified space. However using the control conditions (C5) and (C6) one can show that a controlled stratified vector field is locally integrable and its (local) flow is continuous, strata-wise smooth and controlled (see [81, Lemma 2.3], [54, Proposition 10.1]). Weakly controlled, totally anti-radial vector fields are locally integrable to a (local) weakly controlled flow (see [71, Proposition 2.5.1]). Moreover total anti-radiality ensures that trajectories do not leave a stratum in finite time, and moreover, in positive infinite time flow lines can only go from a larger into a smaller stratum.

#### 3.2. Stratified vector fields admitting a sequence of unfoldings

From now on, we fix a sequence of unfoldings (2.9). We denote by  $E_0 := \theta_0^{-1}(Z_0) \simeq S_0$  the exceptional set of the folding map  $\theta_0 : \widetilde{X}^0 \to X$ .

Let  $\xi = \{\xi_i\}_{i=0,\dots,d}$  be a stratified vector field on X. A stratified vector field  $\widetilde{\xi}^0 = \{\widetilde{\xi}_i^0\}_{i=1,\dots,d}$  on  $\widetilde{X}^0$  is called an unfolding of the vector field  $\xi$  under the folding map  $\theta_0$  if

- For i = 1, ..., d,  $\widetilde{\xi}_i^0$  is a smooth vector field on the stratum  $Z_i^0$ , tangent to the exceptional set  $E_0 \cap Z_i^0$  and satisfying  $d\theta_0 \widetilde{\xi}_i^0 = \xi_i \circ \theta_0$  over  $Z_i^0 \setminus E_0$ .
- $\widetilde{\xi}_{|E_0}^0$  is a controlled lift of  $\xi_0$  under the proper controlled submersion  $\theta_{0|E_0}: E_0 \to Z_0$ , i.e.  $\widetilde{\xi}_{|E_0}^0$  is a controlled vector field on the stratified space  $E_0$  and, for each stratum W of  $E_0$ ,  $d\theta_{0|W}\widetilde{\xi}_W^0 = \xi_0 \circ \theta_{0|W}$ .

Not every stratified vector field on X admits an unfolding. However in case it exists, by density, the unfolding is unique. Moreover, if  $\xi$  satisfies the control condition (C5) (resp. (C6), resp. (3.1)) so does the unfolding  $\widetilde{\xi}^0$ .

DEFINITION 3.2. — We say that a stratified vector field  $\xi$  admits a sequence of unfoldings if there exist stratified vector fields  $\tilde{\xi}^l$  over  $\tilde{X}^l$ ,  $l = 0, \ldots, d-1$ , such that  $\tilde{\xi}^l$  is an unfolding of  $\tilde{\xi}^{l-1}$  under the folding map  $\theta_{l,l-1}: \tilde{X}^l \to \tilde{X}^{l-1}$ .

Remark 3.3. — Let  $\lambda: X \to \mathbb{R}$  be a controlled cut-off function. By [11, C.II.Lemma 2.1], for  $l=0\ldots,d-1$ , the composition  $\lambda\circ\theta_l:\widetilde{X}^l\to\mathbb{R}$  is also controlled. Let  $\xi$  be a weakly controlled vector field on X admitting a sequence of unfoldings  $\{\widetilde{\xi}^l\}_l$ . Then  $\lambda\cdot\xi$  is also a weakly controlled vector field admitting a sequence of weakly controlled unfoldings, namely  $\{(\lambda\circ\theta_l)\cdot\widetilde{\xi}^l\}_l$ .

#### 3.3. Stratified anti-radial gradient-like vector fields

Let  $\xi = \{\xi_Z\}_{Z \in \mathcal{S}}$  be a stratified vector field on an abstract stratified space X. A point  $x \in Z$  with  $\xi_Z(x) = 0$  is called a singularity (or singular point) of  $\xi$ . We denote by  $\text{Crit}(\xi)$  the set of singular points of  $\xi$ .

DEFINITION 3.4. — Let  $\xi$  be a stratified vector field and  $x \in Z \cap \text{Crit}(\xi)$ . The point x is called a singularity of strong standard form of  $\xi$  if there exists an open distinguished neighbourhood  $U_x$  of x in X and local coordinates  $z_1, \ldots, z_{\dim Z}$  of Z such that the vector field  $\xi$  has normal form

(3.2) 
$$\xi_{|U_x} = -r \frac{\partial}{\partial r} + \sum_{i=1}^m z_i \frac{\partial}{\partial z_i} - \sum_{i=m+1}^{\dim Z} z_i \frac{\partial}{\partial z_i}$$

in the coordinates  $(z, \varphi, r)$ . We call  $m = \operatorname{ind}(x) = \operatorname{ind}(x, \xi)$  the index of the singular point  $x \in X$ .

Let  $\xi$  be a stratified weakly controlled, totally anti-radial vector field with singularities of strong standard form. We denote by  $\mathrm{Crit}_m(\xi), m=0,\ldots,n$ , the set of singular points of  $\xi$  of index m. Since X is compact and since singularities of strong standard form are isolated, the vector field  $\xi$  has only finitely many singularities. Using the results mentioned in Section 3.1 and the compactness of X, one can show that by integrating the vector field  $\xi$  one gets even a globally defined, continuous, strata-wise smooth and weakly controlled flow  $\Phi: X \times \mathbb{R} \to X$ .

A point of X is called a wandering point of the flow  $\Phi$  if there exists an open neighbourhood V of the point as well as  $t_0 \in \mathbb{R}$  such that  $\Phi(t, V) \cap V = \emptyset$  for  $|t| > t_0$ . We denote by  $\Omega_{\Phi}$  the set of non-wandering points of  $\Phi$ . Clearly  $\mathrm{Crit}(\xi) \subset \Omega_{\Phi}$ , but the converse does not hold in general.

Let us assume that  $\operatorname{Crit}(\xi) = \Omega_{\Phi}$ . Then, for  $x \in Z \cap \operatorname{Crit}(\xi)$ , the unstable set

$$(3.3) W^u(x) \coloneqq \left\{ y \in X \, \middle| \, \lim_{t \to -\infty} \Phi(y, t) = x \right\} \subset Z$$

is an embedded submanifold (see [48, Proposition 5.1]). The stable set

$$(3.4) W^s(x) \coloneqq \left\{ y \in X \, \middle| \, \lim_{t \to \infty} \Phi(y, t) = x \right\} \subset \bigcup_{Z < R} R$$

is an abstract stratified space with stratification and control data inherited from those of X (see [48, Proposition 5.2]).

It follows from the total anti-radiality of the vector field  $\xi$  that the strata of X are invariant for the flow  $\Phi$  and that moreover in positive infinite time

flow lines can only go from a larger into a smaller stratum. Therefore, for  $x, y \in \text{Crit}(\xi), x \in Z, y \in R$ ,

$$(3.5) W^u(x) \cap W^s(y) \neq \emptyset \Longrightarrow R \leqslant Z \text{ and } W^u(x) \cap W^s(y) \subset Z.$$

In view of (3.5) it is possible to define the Morse–Smale transversality condition, also in this stratified context: We say that  $\xi$  satisfies the Morse–Smale condition, if for all singular points  $x \in Z \cap \operatorname{Crit}(\xi)$  and  $y \in R \cap \operatorname{Crit}(\xi)$   $(R \leqslant Z)$  the manifolds  $W^u(x)$  and  $W^s(y) \cap Z$  intersect transversally, i.e. we have

$$T_wW^u(x)+T_w\left(W^s(y)\cap Z\right)=T_wZ$$
 for all points  $w\in W^u(x)\cap W^s(y)\subset Z$ .

We denote by  $\mathcal{M}(x,y) := (W^u(x) \cap W^s(y))/\mathbb{R}$  the set of unparameterised trajectories starting in x and ending in y. The Morse–Smale transversality condition implies that  $\mathcal{M}(x,y)$  is a smooth manifold of dimension  $\operatorname{ind}(x) - \operatorname{ind}(y) - 1$ . In particular  $\mathcal{M}(x,y) = \emptyset$  if  $\operatorname{ind}(x) \leq \operatorname{ind}(y)$ .

DEFINITION 3.5. — A stratified vector field  $\xi$  on the abstract stratified space X is called anti-radial gradient-like if it is weakly controlled, totally anti-radial, with singularities of strong standard form and moreover the following three conditions hold:

- (a)  $\Omega_{\Phi} = \operatorname{Crit}(\xi)$ .
- (b)  $\xi$  satisfies the Morse–Smale condition.
- (c)  $\xi$  admits a sequence of unfoldings  $\{\widetilde{\xi}^l\}_{l\in\{0,\dots,d-1\}}$ .

PROPOSITION 3.6. — Let  $\xi$  be a stratified anti-radial gradient-like vector field. We denote by  $\{\widetilde{\xi}^l\}_{l\in\{0,\dots,d-1\}}$  the sequence of unfoldings of  $\xi$ . Then

- (a) For  $l \in \{0, ..., d-1\}$  the weakly controlled, totally anti-radial vector field  $\tilde{\xi}^l$  induces a weakly controlled flow  $\tilde{\Phi}^l : \tilde{X}^l \times \mathbb{R} \to \tilde{X}^l$ , satisfying
- (3.6)  $\theta_l(\widetilde{\Phi}^l(x,t)) = \Phi(\theta_l(x),t)$  for all  $x \in \widetilde{X}^l$  and all  $t \in \mathbb{R}$ .

Moreover for  $i, j \in \{0, \dots, d-1\}, i < j$ ,

(3.7) 
$$\theta_{j,i}(\widetilde{\Phi}^j(x,t)) = \widetilde{\Phi}^i(\theta_{j,i}(x),t)$$
 for all  $x \in \widetilde{X}^j$  and all  $t \in \mathbb{R}$ .

(b) For  $l \in \{0, ..., d-1\}$  and  $x \in \text{Crit}(\xi)$ , set  $B_l(x) := \theta_l^{-1}(x)$ . We have

(3.8) 
$$\operatorname{Crit}(\widetilde{\xi}^l) = \bigcup_{x \in \operatorname{Crit}(\xi)} B_l(x).$$

(c) For  $l \in \{0, ..., d-1\}$  and  $x \in \text{Crit}(\xi)$  the stable resp. unstable set for  $B_l(x)$ ,

$$(3.9) \widetilde{W}_l^{s/u}(x) := \left\{ w \in \widetilde{X}^l \, \middle| \, \lim_{t \to \pm \infty} \widetilde{\Phi}^l(w, t) \in B_l(x) \right\},$$

are abstract stratified spaces.

Proof.

- (a). The unfolding  $\tilde{\xi}^0$  is weakly controlled and totally anti-radial with respect to the induced control data on  $\widetilde{X}^0$ , hence  $\widetilde{\xi}^0$  is locally integrable to a weakly controlled, strata-wise smooth flow. Since the restriction of  $\theta_0$  to the complement of the exceptional set  $E_0$  is a stratified covering and the flow  $\Phi$  is globally defined, the flow  $\widetilde{\Phi}^0$  induced from  $\widetilde{\xi}^0$  is also globally defined on  $\widetilde{X}^0 \setminus E_0$ . On  $E_0$  the vector field  $\widetilde{\xi}^0$  is the controlled lift under a proper controlled submersion of a smooth vector field with global flow. By [81, Lemma 2.4] the flow of  $\widetilde{\Phi}^0$  on  $E_0$  then also exists for all time  $t \in \mathbb{R}$ . Thus  $\widetilde{\xi}^0$  induces a globally defined stratified flow  $\widetilde{\Phi}^0: \widetilde{X} \times \mathbb{R} \to \widetilde{X}$  and  $E_0$  is invariant under the flow  $\widetilde{\Phi}^0$ . The argument above can be iterated to show global existence of the flow  $\widetilde{\Phi}^0: \widetilde{X}^l \times \mathbb{R} \to \widetilde{X}^l$  induced from  $\widetilde{\xi}^l$ ,  $l = 0, \ldots, d-1$ . By definition of the unfolding of a stratified vector field, the vector fields  $\xi, \widetilde{\xi}^i, \widetilde{\xi}^j$  for  $i, j \in \{0, \ldots, d-1\}$  are related, hence (3.6) and (3.7) hold.
- (b). Singular points of  $\tilde{\xi}^l$  are precisely the fixed points of the flow  $\tilde{\Phi}^l$ . Therefore, using the commutativity (3.6),  $\operatorname{Crit}(\tilde{\xi}^l) \subset \cup_{x \in \operatorname{Crit}(\xi)} B_l(x)$ . Let  $x \in \operatorname{Crit}(\xi) \cap Z_i$ , i > l. For a small enough open neighbourhood  $U_x \subset X$  of x the preimage  $\theta_l^{-1}(U_x)$  consists of  $2^{l+1}$  disjoint copies of  $U_x$ ,  $\theta_{l|\theta_l^{-1}(U_x)}$  is a covering map, and  $d\theta_l \tilde{\xi}^l = \xi \circ \theta_l$  on  $\theta_l^{-1}(U_x)$ . Therefore  $B_l(x) \subset \operatorname{Crit}(\tilde{\xi}^l)$ . In case  $x \in \operatorname{Crit}(\xi) \cap Z_i$ , i < l (resp. i = l), we have that for a distinguished neighbourhood  $U_x$  the preimage  $\theta_l^{-1}(U_x)$  is a disjoint union of  $2^i$  connected components, which are isomorphic (as stratified spaces) to  $\mathbb{R}^{\dim Z_i} \times (-\epsilon, \epsilon) \times \tilde{L}_x^{l-i-1}$  (resp. to  $\mathbb{R}^{\dim Z_i} \times (-\epsilon, \epsilon) \times L_x$ ). It follows from the definition of a singularity of strong standard form that the restriction of  $\tilde{\xi}^l$  to these connected components can be written as in the normal form (3.2). Hence we deduce  $B_l(x) \subset \operatorname{Crit}(\tilde{\xi}^l)$ .
- (c). The claim follows using the normal form (3.2) of a singularity of strong standard form and arguing as in the proof of [48, Propositions 5.1 and 5.2].

Let  $\xi$  be a stratified anti-radial gradient-like vector field. For  $x, y \in \text{Crit}(\xi)$  we denote by  $\widetilde{\mathcal{M}}_l(x,y) := (\widetilde{W}_l^u(x) \cap \widetilde{W}_l^s(y))/\mathbb{R}$  the trajectory space

of unparameterised trajectories of  $\widetilde{\Phi}^l$  starting in  $B_l(x)$  and ending in  $B_l(y)$ . Following the flow as  $t \to \pm \infty$ , we get endpoint maps

(3.10) 
$$\widetilde{\pi}_{x,y,l}^{s}: \widetilde{\mathcal{M}}_{l}(x,y) \longrightarrow B_{l}(y),$$

$$\widetilde{\pi}_{x,y,l}^{u}: \widetilde{\mathcal{M}}_{l}(x,y) \longrightarrow B_{l}(x).$$

Similarly we have natural maps

(3.11) 
$$\widetilde{\pi}_{x,l}^{s/u}: \widetilde{W}_l^{s/u}(x) \longrightarrow B_l(x).$$

PROPOSITION 3.7. — Let  $\xi$  be a stratified anti-radial gradient-like vector field on a compact abstract stratified space X. For  $x, y \in \text{Crit}(\xi)$  and all  $w \in B_l(x)$ , the intersection of  $\widetilde{W}_l^s(y)$  and  $(\widetilde{\pi}_{x,l}^u)^{-1}(w)$  is transverse (in the stratum, where it takes place).

*Proof.* — Since the flow preserves transverse intersections, it is enough to check the transversality condition in a neighbourhood of  $B_l(x)$ , where the vector field can be written in normal form (3.2). The latter follows easily using the Morse–Smale property for  $\xi$ .

Similarly to the smooth situation (see [6, page 140]), the transversality condition in Proposition 3.7 together with the properties of the unfolding process, has the following important consequences

COROLLARY 3.8. — The trajectory spaces  $\widetilde{\mathcal{M}}_l(x,y)$  are stratified spaces with stratification and control data inherited from those of  $\widetilde{X}^l$ . Moreover, the endpoint maps  $\widetilde{\pi}^u_{x,y,l}$  are locally trivial stratified fibre bundles.

### 3.4. Existence of stratified anti-radial gradient-like vector fields. Morse and Morse-Bott function

An important tool in the theory of abstract stratified spaces is the existence of controlled lifts of smooth vector fields under controlled submersions (see [81, Lemma 2.4]). One can easily prove a stronger version of this lifting property, namely the existence of controlled lifts which in addition admit a sequence of unfoldings. Using this stronger lifting property and Remark 3.3 one can modify the inductive proof of the existence of stratified gradient-like vector fields [48, Proposition 6.4] to get

Proposition 3.9. — Let X be a compact abstract stratified space. Then there exists a stratified anti-radial gradient-like vector field on X.

For a strata-wise smooth function  $f: X \to \mathbb{R}$  and  $x \in Z$ , we denote by  $df_x := d(f_{|Z})_x$ . Generalising [78, Theorem B] one can prove the existence of a self-indexing "Morse function"  $f: X \to \mathbb{R}$  with negative pseudo-gradient vector field  $\xi$ :

PROPOSITION 3.10. — There exists a continuous strata-wise smooth function  $f: X \to \mathbb{R}$  with the following properties:

(a) Let  $x \in Z$ . Then

(3.12) 
$$df_x(\xi_Z) < 0 \quad \text{if } x \notin \text{Crit}(\xi_Z), \\ df_x = 0 \quad \text{if } x \in \text{Crit}(\xi_Z).$$

- (b) f is self-indexing, i.e. f(x) = k for  $x \in Crit_k(\xi)$ .
- (c)  $\widetilde{f} := f \circ \theta : \widetilde{X} \to \mathbb{R}$  is a smooth Morse–Bott function.

### 4. The Morse–Bott type cochain complex computing intersection cohomology

In this section X denotes an n-dimensional compact stratified pseudomanifold according to our conventions at the beginning of Section 2.2;  $\xi$  is a stratified anti-radial gradient-like vector field on X as in Definition 3.5. Moreover  $\bar{p}$  and  $\bar{q}$  denote complementary perversities.

The aim of this section is to construct the Morse–Bott type cochain complex  $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$  associated to  $\xi$  and the perversity  $\bar{p}$ , and which computes the intersection cohomology of X.

Section 4 is organised as follows: In view of the results in Sections 3.3 and 3.4 the unfolding vector field  $\widetilde{\xi} := \widetilde{\xi}_{d-1}$  on  $\widetilde{X}$  is a smooth Morse–Bott vector field satisfying the Morse–Bott transversality condition of [6]. In Section 4.1 we recall the definition of the Morse–Bott cochain complex  $(C^{\bullet}(\widetilde{X},\widetilde{\xi}),\partial^{\bullet})$  of Austin and Braam [6]: the complex  $(C^{\bullet}(\widetilde{X},\widetilde{\xi}),\partial^{\bullet})$  is generated by the de Rham complexes of the critical submanifolds of  $\widetilde{\xi}$ .

In Section 4.2, we construct the Morse–Bott type cochain complex  $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$ . The main idea is to replace the de Rham complex of the critical submanifolds in Austin and Braam's construction with the complex of liftable intersection differential forms  $(\mathcal{K}_{\bar{q},tr}^{\bullet}(L_x),d), x \in \text{Crit}(\xi)$ . The main technical difficulty is to show the well-definedness of the boundary operator  $\partial^{\bullet}$ . For this we need to show that the boundary operator maps liftable intersection differential forms of perversity  $\bar{q}$  to liftable intersection differential forms of perversity  $\bar{q}$ .

In Section 4.3 we prove the Main Theorem 1, i.e. that the Morse–Bott type cochain complex  $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$  computes the intersection cohomology  $IH_{\bar{p}}^{\bullet}(X)$ .

# 4.1. The smooth Morse–Bott cochain complex associated to $(\widetilde{X},\widetilde{\xi})$

The vector field  $\widetilde{\xi} := \widetilde{\xi}_{d-1}$  is a smooth vector field on the smooth manifold  $\widetilde{X}$ . We denote by  $\widetilde{\Phi} := \widetilde{\Phi}^{d-1} : \widetilde{X} \times \mathbb{R} \to \widetilde{X}$  the smooth flow induced from  $\widetilde{\xi}$ . For  $x \in \operatorname{Crit}(\xi)$  we write shortly  $B(x) := B_{d-1}(x)$ . We denote by  $\widetilde{W}^{u/s}(x) := \widetilde{W}^{u/s}_{d-1}(x)$  the unstable/stable manifold of B(x). For  $x, y \in \operatorname{Crit}(\xi)$  we denote by  $\widetilde{\mathcal{M}}(x,y) := \widetilde{\mathcal{M}}_{d-1}(x,y) = (\widetilde{W}^u(x) \cap \widetilde{W}^s(y))/\mathbb{R}$  the space of unparameterised trajectories of the flow  $\widetilde{\Phi}$  starting in B(x) and ending in B(y) and by  $\widetilde{\pi}^{u/s}_{x,y}$  the endpoint maps:

(4.1) 
$$\widetilde{\pi}_{x,y}^u \coloneqq \widetilde{\pi}_{x,y,d-1}^u : \widetilde{\mathcal{M}}(x,y) \longrightarrow B(x),$$

$$\widetilde{\pi}_{x,y}^s \coloneqq \widetilde{\pi}_{x,y,d-1}^s : \widetilde{\mathcal{M}}(x,y) \longrightarrow B(y).$$

In view of the results in Sections 3.3 and 3.4, all assumptions needed in [6] for the construction of the Morse–Bott cochain complex are satisfied (see [6, pages 140-141]), more precisely:

PROPOSITION 4.1. — The following holds for the flow  $\widetilde{\Phi}: \widetilde{X} \times \mathbb{R} \to \widetilde{X}$  induced from  $\widetilde{\mathcal{E}}$ :

(a) The set of singular points of  $\tilde{\xi}$  is given by

(4.2) 
$$\operatorname{Crit}(\widetilde{\xi}) := \left\{ w \in \widetilde{X} \,\middle|\, \widetilde{\xi}(w) = 0 \right\} = \bigcup_{x \in \operatorname{Crit}(\xi)} B(x).$$

For  $x \in \operatorname{Crit}(\xi)$ , the critical submanifold B(x) is of Morse–Bott type with trivial unstable normal bundle of rank  $\operatorname{ind}(x)$ . The critical submanifolds B(x),  $x \in \operatorname{Crit}(\xi)$ , and their unstable normal bundles are orientable.

- (b) For  $x, y \in \text{Crit}(\xi)$  with  $\text{ind}(x) \leqslant \text{ind}(y)$ , we have  $\widetilde{\mathcal{M}}(x, y) = \emptyset$ .
- (c) For all  $x, y \in \text{Crit}(\xi)$  and all  $w \in B(x)$ ,  $(\widetilde{\pi}_x^u)^{-1}(w)$  and  $\widetilde{W}^s(y)$  intersect transversally.

As a consequence of the transversality condition in Proposition 4.1(c) the trajectory space  $\widetilde{\mathcal{M}}(x,y), x,y \in \mathrm{Crit}(\xi)$ , is a smooth manifold admitting a compactification (in the topology of convergence up to broken trajectories) to a manifold with corners (see [6, Lemma 3.3]) and moreover, the endpoint map  $\widetilde{\pi}_{x,y}^u : \widetilde{\mathcal{M}}(x,y) \to B(x)$  is a locally trivial fibre bundle (see [6, page 140]). We fix orientations on all B(x)'s and their unstable normal bundles. Together with the flow, this induces orientations on the trajectory manifolds  $\widetilde{\mathcal{M}}(x,y)$ . Integration of smooth forms along the fibres of  $\widetilde{\pi}_{x,y}^u$  is well-defined and will be denoted by  $(\widetilde{\pi}_{x,y}^u)_*$ .

We now recall the construction of Austin and Braam [6]: For  $x \in \text{Crit}(\xi)$ , we denote by  $(\Omega^{\bullet}(B(x)), d)$  the de Rham complex of the smooth compact manifold B(x). Let  $x, y \in \text{Crit}(\xi)$ . Let  $\widetilde{\sigma}_{xy}$  be the linear map defined by

(4.3) 
$$\widetilde{\sigma}_{xy}: \Omega^{\bullet}(B(y)) \longrightarrow \Omega^{\bullet}(B(x)) \\ \omega \longmapsto \widetilde{\sigma}_{xy}(\omega) := (\widetilde{\pi}_{x,y}^{u})_{*} (\widetilde{\pi}_{x,y}^{s})^{*} \omega.$$

For  $\omega \in \Omega^{j}(B(y))$ , we have that  $\widetilde{\sigma}_{xy}(\omega) \in \Omega^{j-\operatorname{ind}(x)+\operatorname{ind}(y)+1}(B(x))$ .

Definition 4.2. — The Morse–Bott cochain complex  $(C^{\bullet}(\widetilde{X}, \widetilde{\xi}), \partial^{\bullet})$  associated to  $\widetilde{\xi}$  is defined by

$$(4.4) C^k(\widetilde{X},\widetilde{\xi}) \coloneqq \bigoplus_{i+j=k} C^{i,j}(\widetilde{X},\widetilde{\xi}) \coloneqq \bigoplus_{i+j=k \text{ ind}(x)=i} \Omega^j(B(x)).$$

The boundary operator  $\partial^{\bullet} = \bigoplus \partial_t^{\bullet}$  is defined by

$$(4.5) \partial_t: C^{i,j}(\widetilde{X}, \widetilde{\xi}) \longrightarrow C^{i+t,j-t+1}(\widetilde{X}, \widetilde{\xi}),$$

where for  $y \in \operatorname{Crit}_i(\xi)$ ,  $\omega \in \Omega^j(B(y))$ ,

(4.6) 
$$\partial_t \omega = \begin{cases} d\omega & \text{for } t = 0, \\ (-1)^j \sum_{\text{ind}(x) = i+t} \widetilde{\sigma}_{xy}(\omega) & \text{for } t > 0. \end{cases}$$

To prove that  $(C^{\bullet}(\widetilde{X},\widetilde{\xi}),\partial^{\bullet})$  is indeed a complex, one has to show that  $\partial^2 = 0$ , see [6, Proposition 3.5]. More precisely using the compactification of the spaces of gradient lines, one can show that for  $k \geq 0$ ,

$$(4.7) \sum_{t=0}^{k} \partial_{k-t} \partial_t = 0.$$

Remark 4.3. — In case that all critical submanifolds of a smooth Morse–Bott vector field are just isolated singular points, i.e. the vector field is gradient-like, the Morse–Bott cochain complex is just the usual Morse–Thom–Smale cochain complex. In particular, in our situation, we have that, for  $x, y \in \text{Crit}(\xi) \cap Z_d$  the map (4.3) is the trivial map unless ind(x) = ind(y) + 1. In the latter case,  $\widetilde{\sigma}_{xy}$  is just given by counting with signs trajectories of  $\xi$  starting in x and ending in y. More precisely, let us denote by  $\widehat{B}(x) \simeq \{x\}$  (resp. by  $\widehat{B}(y) \simeq \{y\}$ ) the connected component of B(x) (resp. of B(y)) included in  $\widehat{X}$ . Then

(4.8) 
$$\sigma_{xy} := \widetilde{\sigma}_{xy|\Omega^{\bullet}(\widehat{B}(y))} : \Omega^{\bullet}(\widehat{B}(y)) \simeq \mathbb{R} \longrightarrow \Omega^{\bullet}(\widehat{B}(x)) \simeq \mathbb{R}$$
$$1 \longmapsto \widetilde{\sigma}_{xy}(1) = \pm n(x, y),$$

where n(x, y) is the number of trajectories between x and y counted with signs (see e.g. [42, Section c] for more details).

The Morse–Bott cochain complex  $(C^{\bullet}(\widetilde{X}, \widetilde{\xi}), \partial^{\bullet})$  is filtered by the index of singular points of  $\xi$ ,

$$(4.9) 0 \subset F_k C^k(\widetilde{X}, \widetilde{\xi}) \subset \cdots \subset F_0 C^k(\widetilde{X}, \widetilde{\xi}) = C^k(\widetilde{X}, \widetilde{\xi}),$$

where

(4.10) 
$$F_m C^k(\widetilde{X}, \widetilde{\xi}) := \bigoplus_{i+j=k, i \geqslant m} C^{i,j}(\widetilde{X}, \widetilde{\xi}).$$

Let  $f: X \to \mathbb{R}$  be a self-indexing function, s.th.  $\widetilde{f} = f \circ \theta : \widetilde{X} \to \mathbb{R}$  is a smooth self-indexing Morse–Bott function as in Proposition 3.10. For  $m \in \mathbb{N}_0$ , set  $\widetilde{Y}_m := \widetilde{f}^{-1}((m-1/2,\infty)) \subset \widetilde{X}$ . We denote by  $(\Omega_c^{\bullet}(\widetilde{Y}_m),d)$  the de Rham complex of smooth compactly supported forms on  $\widetilde{Y}_m$ . The de Rham complex  $(\Omega^{\bullet}(\widetilde{X}),d)$  computes the de Rham cohomology  $H_{dR}^{\bullet}(\widetilde{X})$  and is filtered by

$$(4.11) 0 \subset \Omega_c^{\bullet}(\widetilde{Y}_n) \subset \Omega_c^{\bullet}(\widetilde{Y}_{n-1}) \subset \cdots \subset \Omega_c^{\bullet}(\widetilde{Y}_0) = \Omega^{\bullet}(\widetilde{X}).$$

For  $x \in \operatorname{Crit}(\xi)$ , the endpoint map  $\widetilde{\pi}^u_x \coloneqq \widetilde{\pi}^u_{x,d-1} : \widetilde{W}^u(x) \to B(x)$ ,  $\widetilde{\pi}^u_x(w) = \lim_{t \to -\infty} \widetilde{\Phi}(w,t)$ , is a locally trivial fibre bundle. Actually in our situation it follows from the normal form (3.2) of a singularity of strong standard form and the unfolding process that the unstable fibre bundle is a trivial bundle. By [6, Lemma 3.3] the unstable manifold  $\widetilde{W}^u(x)$  can be compactified to a manifold with corners. Integration of smooth forms along the fibres of the fibre bundle  $\widetilde{\pi}^u_x$  is well-defined and will be denoted by  $(\widetilde{\pi}^u_x)_*$ . For  $\omega \in \Omega^k(\widetilde{X})$ , we denote by  $\omega_{|\widetilde{W}^u(x)}$  the pullback of  $\omega$  via the inclusion map  $\widetilde{W}^u(x) \subset \widetilde{X}$ . We define the linear map  $\widetilde{\Psi}^i$  by

$$(4.12) \qquad \widetilde{\Psi}^{i}: \Omega^{k}(\widetilde{X}) \longrightarrow \widetilde{C}^{i,k-i}(\widetilde{X},\widetilde{\xi})$$

$$\omega \longmapsto \widetilde{\Psi}^{i}(\omega) = \sum_{x \in \operatorname{Crit}_{i}(\xi)} (\widetilde{\pi}_{x}^{u})_{*}(\omega_{|\widetilde{W}^{u}(x)}).$$

Note that for  $\omega \in \Omega_c^{\bullet}(\widetilde{Y}_m)$  and i < m, we have  $\widetilde{\Psi}^i(\omega) = 0$ . Hence  $(\bigoplus_i \widetilde{\Psi}^i)(\omega) \in F_m C^{\bullet}(\widetilde{X}, \widetilde{\xi})$ .

Theorem 4.4 ([6, Theorems 3.1 and 3.8, Lemma 3.6]). — The map

$$(4.13) \qquad \widetilde{\Psi} = \bigoplus_{i} \widetilde{\Psi}^{i} : (\Omega^{\bullet}(\widetilde{X}), d) \longrightarrow (C^{\bullet}(\widetilde{X}, \widetilde{\xi}), \partial^{\bullet})$$

is a map of filtered cochain complexes inducing an isomorphism

$$(4.14) H_{dR}^{\bullet}(\widetilde{X}) = H^{\bullet}(\Omega^{\bullet}(\widetilde{X}), d) \simeq H^{\bullet}(C^{\bullet}(\widetilde{X}, \widetilde{\xi}), \partial^{\bullet}).$$

#### **4.2.** The Morse–Bott type cochain complex $(C^{\bullet}_{\bar{n}}(X,\xi),\partial^{\bullet})$

#### 4.2.1. Definition of the Morse–Bott type cochain complex $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$

Before defining the complex  $(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$  in Definition 4.8, we need several preparatory results (Lemmas 4.5 and 4.6, Proposition 4.7) to tackle the well-definedness of the boundary operator  $\partial^{\bullet}$ .

Let  $x, y \in \text{Crit}(\xi)$ , with  $x \neq y$ ,  $x \in Z_l$  and  $y \in Z_k$  and such that  $\mathcal{M}(x,y) \neq \emptyset$ . By total anti-radiality of the vector field  $\xi$ , we have  $k \leq l$ . We assume that l < d.

We consider the l-th unfolding  $\widetilde{X}^l$  of X. Recall that by definition of unfoldings,  $B_l(x)$  is a disjoint union of copies of  $L_x$  (see end of Section 2.1.3). If k = l, then  $B_l(y)$  is a disjoint union of copies of  $L_y$ . If k < l, then  $B_l(y)$  is a disjoint union of copies of the (l - k - 1)-th unfolding of  $L_y$ . Let us denote by  $\widehat{X}^l := \theta_{d-1,l}(\widehat{X})$  the marked leaf of  $\widetilde{X}^l$ . We denote by  $\widehat{B}_l(x) \simeq L_x$  (resp. by  $\widehat{B}_l(y)$ ) the connected component of  $B_l(x)$  (resp. of  $B_l(y)$ ) with  $\widehat{B}_l(x) \subset \widehat{X}^l$  (resp. with  $\widehat{B}_l(y) \cap \widehat{X}^l \neq \emptyset$ ). We have used the notation  $\widehat{B}(x)$ ,  $\widehat{B}(y)$  for the corresponding connected components of B(x), B(y) in  $\widehat{X}$ .

Recall that for a link L, we have an injective map of cochain complexes  $(\Omega^{\bullet}_{\mathrm{lift}}(L), d) \to (\Omega^{\bullet}(\widetilde{L}), d)$  (resp.  $(\mathcal{K}^{\bullet}_{\overline{q}}(L), d) \to (\Omega^{\bullet}(\widetilde{L}), d)$ ) by sending a liftable form  $\omega$  to its unique lift  $\widetilde{\omega}$ . In the following, we will often tacitly identify  $(\Omega^{\bullet}_{\mathrm{lift}}(L), d)$  (resp.  $(\mathcal{K}^{\bullet}_{\overline{q}}(L), d)$ ) with the image subcomplex in  $(\Omega^{\bullet}(\widetilde{L}), d)$ . In particular,  $(\Omega^{\bullet}_{\mathrm{lift}}(\widehat{B}_{l}(x)), d) = (\Omega^{\bullet}_{\mathrm{lift}}(L_{x}), d) \subset (\Omega^{\bullet}(\widehat{B}(x)), d) = (\Omega^{\bullet}(\widetilde{L}_{x}), d)$ . Note also that, by uniqueness of lifts, we can identify liftable forms on a link L with liftable forms on some unfolding of L. Hence  $(\Omega^{\bullet}_{\mathrm{lift}}(\widehat{B}_{l}(y)), d) = (\Omega^{\bullet}_{\mathrm{lift}}(L_{y}), d)$ .

LEMMA 4.5. — The map  $\tilde{\sigma}_{xy}$  defined in (4.3) restricts to a linear map

$$(4.15) \qquad \widehat{\sigma}_{xy}: \Omega_{\text{lift}}^{\bullet}(L_y) = \Omega_{\text{lift}}^{\bullet}(\widehat{B}_l(y)) \longrightarrow \Omega_{\text{lift}}^{\bullet}(L_x) = \Omega_{\text{lift}}^{\bullet}(\widehat{B}_l(x)).$$

Proof. — The result is essentially a consequence of the construction of the unfolding (see Section 2.1.3) and of the commutativity of flows (3.7): The stratified spaces  $\widetilde{X}^l$ ,  $\widetilde{\mathcal{M}}_l(x,y)$ ,  $B_l(x)$ ,  $B_l(y)$  all carry stratifications and control data which are induced from the stratification of X and are therefore compatible. Denoting by  $E_l$  the exceptional set of the folding map  $\theta_{l,l-1}: \widetilde{X}^l \to \widetilde{X}^{l-1}$ , we have  $B_l(x) \subset E_l$ . We can also identify  $\widetilde{\mathcal{M}}_l(x,y)$  with a stratified subset of  $E_l \cap \widetilde{X}^l$ , namely  $\widetilde{\mathcal{M}}_l(x,y) \simeq \widetilde{W}_l^u(x) \cap \widetilde{W}_l^s(y) \cap (\theta_l \circ f)^{-1}(a)$ , where f is a Morse function as in Section 3.4 and  $a \in (f(y), f(x))$  is a regular level. The exceptional set  $E_l$  is flow invariant, the flow on  $E_l$  is controlled (not only weakly controlled). The endpoint

maps  $\widetilde{\pi}_{x,y,l}^{s/u}$  defined in (3.10) are strata-preserving and strata-wise smooth. Moreover, by Corollary 3.8,  $\widetilde{\pi}_{x,y,l}^u$  is a locally trivial stratified fibre bundle.

We denote by  $\widehat{\mathcal{M}}_l(x,y) := \widetilde{\mathcal{M}}_l(x,y) \cap \widehat{X}^l$ , i.e. the space of trajectories between  $\widehat{B}_l(x)$  and  $\widehat{B}_l(y)$ . We denote by  $\widehat{\mathcal{M}}_l(x,y)^{sm} \subset \widehat{\mathcal{M}}_l(x,y)$  the top stratum of the stratified space  $\widehat{\mathcal{M}}_l(x,y)$ ;  $\widehat{\mathcal{M}}_l(x,y)^{sm}$  parameterises the trajectories between  $\widehat{B}_l(x)$  and  $\widehat{B}_l(y)$  which also lie in  $\widetilde{X}^{l,sm}$ . We denote by

$$\widehat{\pi}_{x,y,l}^s: \widehat{\mathcal{M}}_l(x,y)^{sm} \longrightarrow \widehat{B}_l(y) \cap \widetilde{X}^{l,sm}$$

resp. by

$$\widehat{\pi}_{x,y,l}^{u}: \widehat{\mathcal{M}}_{l}(x,y)^{sm} \longrightarrow \widehat{B}_{l}(x) \cap \widetilde{X}^{l,sm}$$

the restriction of the endpoint maps  $\widetilde{\pi}_{x,y,l}^{s/u}$ .

Similarly, we denote by  $\widehat{\mathcal{M}}(x,y)$  the subset of  $\widetilde{\mathcal{M}}(x,y)$  parameterising trajectories between  $\widehat{B}(x)$  and  $\widehat{B}(y)$ ; these trajectories lie in  $\widehat{X}$ . We denote by  $\widehat{\pi}_{x,y}^u: \widehat{\mathcal{M}}(x,y) \to \widehat{B}(x)$ ,  $\widehat{\pi}_{x,y}^s: \widehat{\mathcal{M}}(x,y) \to \widehat{B}(y)$  the restriction of the smooth endpoint maps  $\widetilde{\pi}_{x,y}^{u/s}$ . From (3.7), we get a commutative diagram

(4.18) 
$$\widehat{\pi}_{x,y}^{u}: \qquad \widehat{\mathcal{M}}(x,y) \longrightarrow \widehat{B}(x) \\
\downarrow \theta_{d-1,l} \qquad \qquad \downarrow \theta_{d-1,l} \\
\widetilde{\pi}_{x,y,l|\widehat{\mathcal{M}}_{l}(x,y)}^{u}: \widehat{\mathcal{M}}_{l}(x,y) \longrightarrow \widehat{B}_{l}(x),$$

and similarly for the stable endpoint maps. From the commutativity of the diagram (4.18) and by the well-definedness of integration along fibres of  $\widehat{\pi}_{x,y}^u$ , integration along fibres of  $\widehat{\pi}_{x,y,l}^u$  is well-defined on forms on  $\widehat{\mathcal{M}}_l(x,y)^{sm}$  admitting a smooth lift to  $\widehat{\mathcal{M}}(x,y)$ .

Let  $\alpha \in \Omega^{\bullet}_{lift}(\widehat{B}_{l}(y)) = \Omega^{\bullet}_{lift}(L_{y})$  and denote by  $\widetilde{\alpha} \in \Omega^{\bullet}(\widehat{B}(y))$  its unique lift. By the commutativity of the diagram (4.18) and its stable counterpart, the form  $(\widehat{\pi}^{u}_{x,y})_{*}(\widehat{\pi}^{s}_{x,y})^{*}\widetilde{\alpha} \in \Omega^{\bullet}(\widehat{B}(x))$  is the lift of the form  $\widehat{\sigma}_{xy}(\alpha) := (\widehat{\pi}^{u}_{x,y,l})_{*}(\widehat{\pi}^{s}_{x,y,l})^{*}\alpha \in \Omega^{\bullet}(\widehat{B}_{l}(x)) = \Omega^{\bullet}(L_{x}).$ 

Let W be a stratum of  $\widetilde{X}^l$  with tubular neighbourhood  $T_W$  and projection  $\pi_W: T_W \to W$ . Set  $T_W^{sm} := T_W \cap \widetilde{X}^{l,sm}$ . Since, the restriction of the flow to  $E_l$  is controlled, the sets  $E_l \cap T_W$  and  $E_l \cap T_W^{sm}$  are also preserved by the flow. Hence, again using the identification explained in the proof of Lemma 4.5 above, we have commutative diagrams

$$(4.19) T_W^{sm} \cap \widehat{\mathcal{M}}_l(x,y) \xrightarrow{\pi_W} W \cap \widehat{\mathcal{M}}_l(x,y) \downarrow \widehat{\pi}_{x,y,l}^s \qquad \qquad \downarrow \widetilde{\pi}_{x,y,l}^s T_W^{sm} \cap \widehat{B}_l(y) \xrightarrow{\pi_W} W \cap \widehat{B}_l(y)$$

and

$$(4.20) T_W^{sm} \cap \widehat{\mathcal{M}}_l(x,y) \xrightarrow{\pi_W} W \cap \widehat{\mathcal{M}}_l(x,y) \downarrow \widehat{\pi}_{x,y,l}^u \qquad \qquad \downarrow \widetilde{\pi}_{x,y,l}^u T_W^{sm} \cap \widehat{B}_l(x) \xrightarrow{\pi_W} W \cap \widehat{B}_l(x).$$

All maps in the unstable diagram (4.20) are locally trivial fibre bundles.

LEMMA 4.6. — Let  $q \in \{0, \ldots, \operatorname{codim} W\}$  and let  $\alpha \in \Omega^{\bullet}_{\operatorname{lift}}(\widehat{B}_l(y)) = \Omega^{\bullet}_{\operatorname{lift}}(L_y)$  be a form of perversity q with respect to the smooth submersion  $\pi_{W_{|\widehat{B}_l(y)\cap T_W^{sm}}}$ . Then  $\widehat{\sigma}_{xy}(\alpha) \in \Omega^{\bullet}_{\operatorname{lift}}(\widehat{B}_l(x)) = \Omega^{\bullet}_{\operatorname{lift}}(L_x)$  is a form of perversity q with respect to the smooth submersion  $\pi_{W_{|\widehat{B}_l(x)\cap T_W^{sm}}}$ .

Proof. — We denote by  $\Gamma(\ker(d\pi_W))$  the vector space of smooth sections of the vector bundle  $\ker(d\pi_W)$ . By commutativity of the diagram (4.19), we have  $\pi_W \circ \widehat{\pi}_{x,y,l}^s = \widetilde{\pi}_{x,y,l}^s \circ \pi_W$ . Therefore for  $v_0, \ldots, v_q \in \Gamma(\ker d\pi_W)$ , also  $d\widehat{\pi}_{x,y,l}^s(v_0), \ldots d\widehat{\pi}_{x,y,l}^s(v_q) \in \Gamma(\ker(d\pi_W))$ . By definition of the pullback of forms, and since the form  $\alpha \in \Omega^{\bullet}(\widehat{B}_l(y))$  has perversity q with respect to  $\pi_{W_{\widehat{B}_l(y)} \cap T_{\mathcal{E}_l^m}}$  we get

$$(4.21) \qquad \iota_{v_0} \dots \iota_{v_q} (\widehat{\pi}_{x,y,l}^s)^* \alpha = \iota_{\widehat{d\pi}_{x,y,l}^s(v_0)} \dots \iota_{\widehat{d\pi}_{x,y,l}^s(v_q)} \alpha = 0.$$

Hence  $(\widehat{\pi}_{x,y,l}^s)^* \alpha \in \Omega_{\mathrm{lift}}^{\bullet}(\widehat{\mathcal{M}}_l(x,y))$  is a form of perversity q with respect to the submersion  $\pi_{W_{|\widehat{\mathcal{M}}_l(x,y)\cap T_W^{sm}}}$ . By commutativity of the diagram (4.20), integration along the fibres of  $\widehat{\pi}_{x,y,l}^u$  preserves the perversity of a form. Hence  $(\widehat{\pi}_{x,y,l}^u)_*(\widehat{\pi}_{x,y,l}^s)^* \alpha \in \Omega_{\mathrm{lift}}^{\bullet}(\widehat{B}_l(x))$  is a form of perversity q with respect to the submersion  $\pi_{W_{|\widehat{B}_l(x)\cap T_W^{sm}}}$ .

Proposition 4.7. — The map (4.15) restricts to linear maps

(4.22) 
$$\sigma_{xy}: \mathcal{K}_{\bar{q}}^{\bullet}(L_y) \longrightarrow \mathcal{K}_{\bar{q}}^{\bullet}(L_x)$$

resp.

(4.23) 
$$\sigma_{xy}: \mathcal{K}_{\bar{q},tr}^{\bullet}(L_y) \longrightarrow \mathcal{K}_{\bar{q},tr}^{\bullet}(L_x).$$

*Proof.* — The first claim follows from Lemmas 2.4 and 4.6, [6, Proposition 3.5] (more precisely using (4.7)) and the anti-radiality of  $\xi$ .

We now prove the second claim: We can assume w.l.o.g. that the strata  $Z_k$  and  $Z_l$  are connected. We have to distinguish two cases.

Case 1. — k = l, i.e. the critical points x and y lie in the same (connected) stratum. Therefore  $L_x \simeq L_y$ . Let  $0 \neq \alpha \in \mathcal{K}^{\bullet}_{\overline{q},tr}(L_y)$ . By definition (2.19) of  $\mathcal{K}^{\bullet}_{\overline{q},tr}(L_y)$  we have that

(4.24)  $\deg(\alpha) \leq q_{\dim L_y+1}$  and moreover  $d\alpha = 0$  in case  $\deg(\alpha) = q_{\dim L_y+1}$ .

From (4.22) we know already that  $\sigma_{xy}(\alpha) \in \mathcal{K}^{\bullet}_{\overline{q}}(L_x)$ . In case  $\operatorname{ind}(x) - \operatorname{ind}(y) = 1$ , by [6, Proposition 3.5], the map  $\sigma_{xy}$  is a cochain map. Therefore condition (4.24) for the form  $\alpha$  does imply that the same condition holds for  $\sigma_{xy}(\alpha)$  as well, hence  $\sigma_{xy}(\alpha) \in \mathcal{K}^{\bullet}_{\overline{q},tr}(L_x)$ . In case  $\operatorname{ind}(x) - \operatorname{ind}(y) > 1$ ,  $\operatorname{deg}(\sigma_{xy}(\alpha)) < \operatorname{deg}(\alpha)$ , and hence again  $\sigma_{xy}(\alpha) \in \mathcal{K}^{\bullet}_{\overline{q},tr}(L_x)$ .

Case 2. — k < l. Let  $\widetilde{X}^k$  be the k-th unfolding of X and let  $\widehat{B}_k(y) \simeq L_y$ be the marked connected component of  $B_k(y)$ . We denote by  $\widehat{W}_k^{s/u}(y)$  the stable resp. unstable set of  $\widehat{B}_k(y)$  w.r.t. the flow  $\widetilde{\Phi}^k$ . Since  $y \in \operatorname{Crit}(\xi)$  is a singularity of strong standard form, there exists an open neighbourhood  $U \simeq \mathbb{R}^{\operatorname{ind}(y)} \times \mathbb{R}^{\dim Z_k - \operatorname{ind}(y)} \times L_y \times \mathbb{R} \text{ of } \widehat{B}_k(y) \text{ in } \widetilde{X}^k \text{ such that } \widehat{W}_k^s(y) \cap U \simeq \mathbb{R}^{\operatorname{ind}(y)} \times \mathbb{R}^{\operatorname{ind}(y)}$  $\{0\} \times \mathbb{R}^{\dim Z_k - \operatorname{ind}(y)} \times L_y \times \mathbb{R} \text{ and } \widehat{B}_k(y) \simeq \{0\} \times \{0\} \times L_y \times \{0\}. \text{ Let } W :=$  $L_{\nu} \cap Z_{l}$  be the stratum of  $L_{\nu}$  corresponding to the stratum  $Z_{l}$  of X. For a point in W there exists a distinguished neighbourhood  $V \subset L_y$ , with  $V \simeq$  $\mathbb{R}^{\dim Z_l - \dim Z_k - 1} \times cL_x$ . We denote by  $(z^u, z^s, \varphi, r)$  the local coordinates in  $U' \subset U, U' \simeq \mathbb{R}^{\operatorname{ind}(y)} \times \mathbb{R}^{\dim Z_k - \operatorname{ind}(y)} \times V \times \mathbb{R}$ , where  $z = (z^u, z^s)$  are the coordinates in  $\mathbb{R}^{\operatorname{ind}(y)} \times \mathbb{R}^{\dim Z_k - \operatorname{ind}(y)}$ , r is the distance from the stratum  $Z_k$  and  $\varphi = (z^l, \varphi^l, r^l)$  are the local coordinates in V. A form of perversity  $\bar{q}$  on the top stratum of  $\widehat{W}_k^s(y)$  can be written (locally in  $U' \cap \widehat{W}_k^s(y)$ ) as a linear combination of forms  $dz_I^s \wedge d\varphi_J \wedge dr$  and  $dz_I^s \wedge d\varphi_J$ ; where  $d\varphi_J$  is of the form  $dz_{J_1}^l \wedge d\varphi_{J_2}^l \wedge dr^l$  (resp.  $dz_{J_1}^l \wedge d\varphi_{J_2}^l$ ) and  $I, J, J_1, J_2$  are multi-indices. The perversity condition imposes restrictions on the multi-indices  $J, J_1, J_2$ . In particular, the fact that the form  $dz_I^s \wedge d\varphi_J$  with  $d\varphi_J = dz_{J_1}^l \wedge d\varphi_{J_2}^l \wedge dr^l$ (resp.  $d\varphi_J=dz_{J_1}^l\wedge d\varphi_{J_2}^l$ ) has perversity  $q_{\operatorname{codim} Z_l^k}=q_{\operatorname{codim} W}=q_{\dim L_x+1}$ with respect to the projection  $T_{Z_l^k} \cap \widehat{W}_k^s(y) \cap \widetilde{X}^{k,sm} \to Z_l^k \cap \widehat{W}_k^s(y)$ , implies that

$$(4.25) |J_2| + 1 \leqslant q_{\text{codim } W} = q_{\dim L_x + 1} \text{ (resp. } |J_2| \leqslant q_{\text{codim } W} = q_{\dim L_x + 1}).$$

Note that under integration of  $dz_I^s \wedge d\varphi_J \wedge dr$  (resp. of  $dz_I^s \wedge d\varphi_J$ ) along the fibres of  $\widehat{\pi}_{x,y,l}^u$ , only the (lift of the) form  $d\varphi_{J_2}^l$  survives.

Let now  $\alpha \in \mathcal{K}_{\bar{q}}^{\bullet}(L_y)$ ; we still denote by the same letter its lift to  $\widehat{B}_l(y)$ . As in the proof of Lemma 4.6, one can show that  $(\widehat{\pi}_{x,y,l}^s)^*(\alpha)$ ,  $(\widehat{\pi}_{x,y,l}^s)^*(d\alpha)$ ,  $\sigma_{xy}(\alpha)$  and  $\sigma_{xy}(d\alpha)$  also have perversity  $\bar{q}$ . From the above discussion (in particular from (4.25)) and the definition of  $\sigma_{xy}$ , we deduce

that  $\deg \sigma_{xy}(\alpha) \leqslant q_{\dim L_x+1}$  and  $\deg d\sigma_{xy}(\alpha) = \deg \sigma_{xy}(d\alpha) \leqslant q_{\dim L_x+1}$ , which implies  $\sigma_{xy}(\alpha) \in \mathcal{K}^{\bullet}_{\overline{q},tr}(L_x)$ .

For  $x \in \operatorname{Crit}(\xi) \cap Z_d$  we have used the notation  $\widehat{B}(x) := \theta^{-1}(x) \cap \widehat{X} \simeq \{x\}$ . We set  $\mathcal{K}^{\bullet}_{\overline{q},tr}(L_x) = \mathcal{K}^{0}_{\overline{q},tr}(L_x) := \Omega^{0}(\widehat{B}(x)) = \mathbb{R}$ . For  $x,y \in \operatorname{Crit}(\xi) \cap Z_d$  we have defined  $\sigma_{xy} : \mathcal{K}^{\bullet}_{\overline{q},tr}(L_y) = \Omega^{\bullet}(\widehat{B}(y)) \to \mathcal{K}^{\bullet}_{\overline{q},tr}(L_x) = \Omega^{\bullet}(\widehat{B}(x))$  as the restriction of the map  $\widetilde{\sigma}_{xy}$  (see (4.8)). By Remark 4.3,  $\sigma_{xy}$  is just given by counting trajectories of  $\widetilde{\Phi}$  between x and y with sign.

DEFINITION 4.8. — To the stratified anti-radial gradient-like vector field  $\xi$  on X and the perversity  $\bar{p}$  we associate the well-defined complex  $(C^{\bullet}_{\bar{p}}(X,\xi),\partial^{\bullet})$ , defined by

$$(4.26) C_{\bar{p}}^{k}(X,\xi) := \bigoplus_{i+j=k} C_{\bar{p}}^{i,j}(X,\xi) := \bigoplus_{i+j=k} \bigoplus_{\mathrm{ind}(x)=i} \mathcal{K}_{\bar{q},tr}^{j}(L_{x}),$$

for k = 0, ..., n. The boundary operator  $\partial^{\bullet} = \oplus \partial_t^{\bullet}$  is defined by

$$(4.27) \partial_t: C^{i,j}_{\bar{p}}(X,\xi) \longrightarrow C^{i+t,j-t+1}_{\bar{p}}(X,\xi),$$

where for  $\alpha \in \mathcal{K}^{j}_{\bar{q},tr}(L_y)$ ,  $\operatorname{ind}(y) = i$ ,

(4.28) 
$$\partial_t \alpha = \begin{cases} d\alpha & \text{for } t = 0, \\ (-1)^j \sum_{\text{ind}(x) = i+t} \sigma_{xy}(\alpha) & \text{for } t > 0. \end{cases}$$

Proof. — Let us denote by  $(\widetilde{C}_{\overline{p}}^{\bullet}, \partial^{\bullet})$  the subcomplex of  $(C^{\bullet}(\widetilde{X}, \widetilde{\xi}), \partial^{\bullet})$  generated by  $\mathcal{K}_{\overline{q},tr}^{\bullet}(L_x) \subset \Omega^{\bullet}(B'(x))$ , where  $x \in \operatorname{Crit}(\xi)$  and B'(x) runs over all connected components of B(x). The fact, that  $(\widetilde{C}_{\overline{p}}^{\bullet}, \partial^{\bullet})$  is a subcomplex, follows arguing as in Lemmas 4.5 and 4.6 and Proposition 4.7. By total anti-radiality of  $\xi$ , there is a subcomplex  $(D^{\bullet}, \partial^{\bullet})$  of  $(\widetilde{C}_{\overline{p}}^{\bullet}, \partial^{\bullet})$ , which is generated by  $\mathcal{K}_{\overline{q},tr}^{\bullet}(L_x) \subset \Omega^{\bullet}(B'(x))$ , where  $x \in \operatorname{Crit}(\xi)$ , and B'(x) now only runs over all connected components of B(x) in  $\widetilde{X} \setminus \widehat{X}$ . The complex  $(C_{\overline{p}}^{\bullet}(X, \xi), \partial^{\bullet})$  can be identified with the quotient complex  $\widetilde{C}_{\overline{p}}^{\bullet}/D^{\bullet}$ .

Remark 4.9. — As mentioned in the introduction, for spaces with isolated singularities, a (homological) complex  $(C^{\bar{q}}_{\bullet}(X, -\xi), \partial_{\bullet})$  associated to the radial gradient-like vector field  $-\xi$  and a perversity  $\bar{q}$  has already been defined in [49, 50]. The complex  $(C^{\bar{q}}_{\bullet}(X, -\xi), \partial_{\bullet})$  defined in [49, 50] is a finite dimensional subcomplex of the complex constructed here,  $(C^{\bar{q}}_{\bullet}(X, -\xi), \partial_{\bullet}) \subset (C^{n-\bullet}_{\bar{v}}(X, \xi), \partial^{n-\bullet})$ .

4.2.2. Filtration of the complex 
$$(C_{\bar{p}}^{\bullet}(X,\xi),\partial^{\bullet})$$

The complex  $(C_{\overline{p}}^{\bullet}(X,\xi),\partial^{\bullet})$  inherits a filtration by the index of singular points of  $\xi$ ,

$$(4.29) 0 \subset F_k C_{\overline{p}}^k(X,\xi) \subset \cdots \subset F_0 C_{\overline{p}}^k(X,\xi) = C_{\overline{p}}^k(X,\xi),$$

where

(4.30) 
$$F_m C^k(X,\xi) := \bigoplus_{i+j=k, i \geqslant m} C^{i,j}_{\bar{p}}(X,\xi).$$

The graded complex associated to the filtered complex  $(C^{\bullet}_{\bar{p}}(X,\xi),\partial^{\bullet})$  is

$$(4.31) G_m^k := F_m C_{\bar{p}}^k(X,\xi) / F_{m+1} C_{\bar{p}}^k(X,\xi) = C_{\bar{p}}^{m,k-m}(X,\xi)$$

$$= \bigoplus_{x \in \operatorname{Crit}_m(\xi)} \mathcal{K}_{\bar{q},tr}^{k-m}(L_x),$$

with boundary given by the de Rham differential. From (2.20) and (4.31), we get for the  $E_1$ -term of the spectral sequence associated to the filtered chain complex  $(C^{\bullet}_{\overline{p}}(X,\xi),\partial^{\bullet})$ 

(4.32) 
$$E_{1}^{m,k} = H^{m+k}(G_{m}^{\bullet}, d) = \left(\bigoplus_{x \in \operatorname{Crit}_{m}(\xi)} H^{k}(\mathcal{K}_{\bar{q}, tr}^{\bullet}(L_{x}), d)\right)$$

$$\simeq \left(\bigoplus_{x \in \operatorname{Crit}_{m}(\xi) \cap \Sigma} IH_{\bar{p}}^{k}(cL_{x})\right) \oplus \left(\bigoplus_{x \in \operatorname{Crit}_{m}(\xi) \cap X^{sm}} H^{k}(\{x\})\right).$$

## 4.3. The Morse–Bott type cochain complex $(C^{\bullet}_{\bar{p}}(X,\xi), \partial^{\bullet})$ computes intersection cohomology

In this section, by adapting the proof of [6, Theorems 3.8] to the singular situation and taking into account the properties of intersection cohomology, we prove the Main Theorem 1, i.e. that the Morse–Bott type cochain complex  $(C^{\bullet}_{\bar{p}}(X,\xi),\partial^{\bullet})$  computes the intersection cohomology  $IH^{\bullet}_{\bar{p}}(X)$ .

4.3.1. Filtration of the complex 
$$(\mathcal{K}_{\bar{q}}^{\bullet}(X), d)$$

For  $m \in \mathbb{N}_0$ , we denote by  $N_m := f^{-1}((-\infty, m - 1/2))$  and by  $Y_m := f^{-1}((m - 1/2, \infty)) = X \setminus \overline{N_m}$ . The complex of liftable intersection forms  $(\mathcal{K}^{\bullet}_{\overline{\sigma}}(X), d)$  is filtered by

$$(4.33) 0 \subset \mathcal{K}_{\bar{a},c}^k(Y_n) \subset \mathcal{K}_{\bar{a},c}^k(Y_{n-1}) \subset \cdots \subset \mathcal{K}_{\bar{a},c}^k(Y_0) = \mathcal{K}_{\bar{a}}^k(X),$$

where  $(\mathcal{K}_{\bar{q},c}^{\bullet}(Y_m),d) \subset (\mathcal{K}_{\bar{q}}^{\bullet}(X),d)$  denotes the subcomplex of the complex of intersection forms, whose lifts are compactly supported in  $\theta^{-1}(Y_m)$ . We denote by  $G'^{\bullet}_{m} = \mathcal{K}_{\bar{q},c}^{\bullet}(Y_m)/\mathcal{K}_{\bar{q},c}^{\bullet}(Y_{m+1}), m \in \mathbb{N}_0$ , the graded complex associated to the filtration (4.33). The short exact sequence of cochain complexes

$$(4.34) 0 \longrightarrow (\mathcal{K}_{\bar{q},c}^{\bullet}(Y_{m+1}), d) \longrightarrow (\mathcal{K}_{\bar{q},c}^{\bullet}(Y_m), d) \longrightarrow (G'_m, d) \longrightarrow 0$$

induces a long exact sequence in cohomology related to the exact sequence in intersection cohomology of the triple  $(X, N_{m+1}, N_m)$  as follows

$$(4.35) \quad \dots H^{i}(\mathcal{K}_{\bar{q},c}^{\bullet}(Y_{m+1}), d) \longrightarrow H^{i}(\mathcal{K}_{\bar{q},c}^{\bullet}(Y_{m}), d) \longrightarrow H^{i}(G'_{m}, d) \quad \dots$$

$$\downarrow \qquad \qquad \downarrow \qquad \qquad \downarrow$$

$$\dots IH_{\bar{p}}^{i}(X, N_{m+1}) \longrightarrow IH_{\bar{p}}^{i}(X, N_{m}) \longrightarrow IH_{\bar{p}}^{i}(N_{m+1}, N_{m}) \dots$$

The first two vertical maps in (4.35) are isomorphisms and hence, by Five Lemma,  $H^i(G_m^{\prime \bullet}, d) \simeq IH^i_{\bar{\eta}}(N_{m+1}, N_m)$ .

Let  $x \in \operatorname{Crit}(\xi)$  and Z be the stratum containing x. Let  $D_x \subset Z$  be a small open ball in Z centred around x, such that  $\pi_{Z|\pi_Z^{-1}(D_x)}$  is trivial, i.e. we have an isomorphism of stratified spaces  $\pi_Z^{-1}(D_x) \simeq D_x \times cL_x$  and  $\xi_{|\pi_Z^{-1}(D_x)}$  has normal form (3.2). We denote by  $D_x^u := W^u(x) \cap D_x \subset Z$  the local unstable disc of x. Using the stratified flow  $\Phi$  we get a stratified deformation retract of the pair  $(N_{m+1}, N_m)$  into the pair

$$(4.36) (N_m \cup H_m, N_m),$$

where

$$(4.37) H_m \simeq \left( \bigcup_{x \in \operatorname{Crit}_m(\xi) \cap \Sigma} (D_x^u \times cL_x) \right) \cup \left( \bigcup_{x \in \operatorname{Crit}_m(\xi) \cap X^{sm}} D_x^u \right).$$

Hence, by the excision formula for intersection cohomology (see [32, Section 1.5]),

$$(4.38) \quad H^{\bullet}(G_{m}^{\bullet}, d) \simeq IH_{\bar{p}}^{\bullet}(N_{m+1}, N_{m})$$

$$\simeq \left(\bigoplus_{x \in \operatorname{Crit}_{m}(\xi) \cap \Sigma} IH_{\bar{p}}^{\bullet}(D_{x}^{u} \times cL_{x}, \partial D_{x}^{u} \times cL_{x})\right)$$

$$\oplus \left(\bigoplus_{x \in \operatorname{Crit}_{m}(\xi) \cap X^{sm}} H^{\bullet}(D_{x}^{u}, \partial D_{x}^{u})\right).$$

4.3.2. The map of filtered complexes  $\Psi: (\mathcal{K}^{\bullet}_{\overline{q}}(X), d) \to (C^{\bullet}_{\overline{p}}(X, \xi), \partial^{\bullet})$ 

Let  $x \in \operatorname{Crit}(\xi)$ . Denote by  $\widehat{W}^u(x)$  the unstable set of  $\widehat{B}(x) \subset \widehat{X}$ . The endpoint map  $\widehat{\pi}^u : \widehat{W}^u(x) \to \widehat{B}(x)$ ,  $\widehat{\pi}^u(w) := \lim_{t \to -\infty} \Phi(w, t)$ , is a locally trivial fibration. Actually in our situation this fibration is even trivial, due to the normal form (3.2), compatible with the unfolding process.

LEMMA 4.10. — Let  $x \in \text{Crit}(\xi)$ . Let  $\omega \in \mathcal{K}_{\overline{q}}^{\bullet}(X)$  with lift  $\widetilde{\omega} \in \Omega^{\bullet}(\widetilde{X})$ . Then

$$(4.39) \qquad (\widehat{\pi}_x^u)_* \widetilde{\omega}_{|\widehat{W}^u(x)} \in \mathcal{K}_{\bar{q},tr}^{\bullet}(L_x).$$

*Proof.* — The claim follows with similar arguments as in the proof of Proposition 4.7.  $\Box$ 

PROPOSITION 4.11. — The map of filtered complexes  $\widetilde{\Psi}$  defined in (4.12), (4.13) induces a map of filtered complexes

$$(4.40) \Psi := \bigoplus_{i} \Psi^{i} : (\mathcal{K}_{\bar{q}}^{\bullet}(X), d) \longrightarrow (C_{\bar{p}}^{\bullet}(X, \xi), \partial^{\bullet}),$$

where

(4.41) 
$$\Psi^{i}: \mathcal{K}_{\bar{q}}^{k}(X) \longrightarrow C_{\bar{p}}^{i,k-i}(X,\xi)$$

$$\omega \longmapsto \Psi^{i}(\omega) = \sum_{x \in \operatorname{Crit}_{i}(\xi)} (\widehat{\pi}_{x}^{u})_{*} \widetilde{\omega}_{|\widehat{W}^{u}(x)}.$$

*Proof.* — The well-definedness of the map follows from Lemma 4.10.  $\Box$ 

#### 4.3.3. Proof of Main Theorem 1

Proof of Main Theorem 1. — We follow closely the proof of [6, Theorem 3.8], whose statement has been recalled in Theorem 4.4. We will prove that the map induced from the map of filtered cochain complexes  $\Psi: (\mathcal{K}^{\bullet}_{\overline{q}}(X), d) \to (C^{\bullet}_{\overline{p}}(X, \xi), \partial^{\bullet})$  on the  $E_1$ -terms of the spectral sequences associated to the two filtered complexes is an isomorphism. Then, by [58, Theorem 3.2], the induced morphism of cohomologies is an isomorphism,  $H^{\bullet}(\mathcal{K}^{\bullet}_{\overline{q}}(X), d) \simeq H^{\bullet}(C^{\bullet}_{\overline{p}}(X, \xi), \partial^{\bullet})$ . The claim then follows using the isomorphism (2.18).

The map induced from the map of filtered cochain complexes  $\Psi$  on the  $E_1$ -terms of the associated spectral sequences, factors as

$$E_{1}^{\prime m,k} \simeq H^{m+k}(G_{m}^{\prime \bullet},d)$$

$$\longrightarrow \left(\bigoplus_{\substack{x \in \operatorname{Crit}_{m}(\xi) \\ x \in \Sigma}} H^{m+k}(\mathcal{K}_{\overline{q}}^{\bullet}(D_{x}^{u} \times cL_{x}, \partial D_{x}^{u} \times cL_{x}), d)\right)$$

$$\oplus \left(\bigoplus_{\substack{x \in \operatorname{Crit}_{m}(\xi) \\ x \in X^{sm}}} H^{m+k}(D_{x}^{u}, \partial D_{x}^{u})\right)$$

$$\longrightarrow E_{1}^{m,k} \simeq \left(\bigoplus_{\substack{x \in \operatorname{Crit}_{m}(\xi) \\ x \in \Sigma}} H^{k}(\mathcal{K}_{\overline{q},tr}^{\bullet}(L_{x}), d)\right)$$

$$\simeq \left(\bigoplus_{\substack{x \in \operatorname{Crit}_{m}(\xi) \\ x \in X^{sm}}} IH_{\overline{p}}^{k}(cL_{x})\right) \oplus \left(\bigoplus_{\substack{x \in \operatorname{Crit}_{m}(\xi) \\ x \in X^{sm}}} H^{k}(\{x\})\right).$$

The first map in (4.42) is induced from the restriction of forms, the second map is induced from integration along the fibres of the unstable fibration. More precisely, for  $x \in \text{Crit}(\xi) \cap \Sigma$  it is given by the composition

$$(4.43) \qquad \qquad \omega \longmapsto \widetilde{\omega} \longmapsto \widetilde{\omega}_{|\widehat{W}^u(x)} \longmapsto (\widehat{\pi}^u_x)_* \widetilde{\omega}_{|\widehat{W}^u(x)},$$

where  $\widetilde{\omega}$  denotes the lift of  $\omega$  to the total unfolding  $D_x^u \times \widetilde{L}_x \times \mathbb{R}$  of  $D_x^u \times cL_x$ . It is not difficult to prove by mimicking the smooth proofs, that

$$(4.44) \quad H^{\bullet}(\mathcal{K}_{\bar{q}}^{\bullet}(D_x^u \times cL_x, \partial D_x^u \times cL_x), d) \simeq IH_{\bar{p}}^{\bullet}(D_x^u \times cL_x, \partial D_x^u \times cL_x).$$

Hence, from (4.38) and (4.44) we conclude that the first map in the factorisation (4.42) is an isomorphism. By (4.44) and the Künneth formula for intersection cohomology (see [32, Section 1.6]),

$$(4.45) IH_{\bar{p}}^{m+k}(D_x^u \times cL_x, \partial D_x^u \times cL_x) \simeq H^m(D_x^u, \partial D_x^u) \otimes IH_{\bar{p}}^k(cL_x)$$

$$\simeq IH_{\bar{p}}^k(cL_x),$$

the second map in (4.42) is also an isomorphism.

# 5. Perturbation of the local Morse–Bott type cochain complex

In this section, let X be an n-dimensional oriented pseudomanifold with 2 strata  $\{Y, Z := X^{sm}\}$ , Y < Z. For simplicity we assume that the singular stratum Y is connected. We denote by L its link and by  $T := \{x \in T_Y \mid \rho_Y(x) \leq \epsilon\}$  its (closed) tubular neighbourhood with boundary  $S_Y = \{x \in T_Y \mid \rho_Y(x) = \epsilon\}$  and projection  $\pi = \pi_Y : T \to Y$ . We denote by  $\widetilde{T} \simeq S_Y \times [-\epsilon, \epsilon]$  the unfolding of the tubular neighbourhood T and by  $\theta : \widetilde{T} \to T$  the folding map. Set  $c := \operatorname{codim}_X Y$ .

Let  $\xi = \{\xi_Y, \xi_Z\}$  be a stratified anti-radial gradient-like vector field on X. Let  $\bar{p}$  and  $\bar{q}$  be complementary perversities.

Since here the link L is a smooth manifold of dimension c-1, the complex  $(\mathcal{K}_{\overline{q},tr}^{\bullet}(L), \partial^{\bullet})$  defined in (2.19) and used in the definition of the Morse–Bott type cochain complex  $(C_{\overline{p}}^{\bullet}(X,\xi), \partial^{\bullet})$ , is just the truncated de Rham complex  $(\Omega_{\overline{q},tr}^{\bullet}(L), d) \subset (\Omega^{\bullet}(L), d)$ ,

(5.1) 
$$\Omega_{\bar{q},tr}^{k}(L) := \begin{cases} \Omega^{k}(L) & \text{if } k < q_{c}, \\ \ker d & \text{if } k = q_{c}, \\ 0 & \text{if } k > q_{c}. \end{cases}$$

Let  $(C^{\bullet}(Z, \xi_Z), \partial^{\bullet})$  be the subcomplex of  $(C^{\bullet}_{\overline{p}}(X, \xi), \partial^{\bullet})$  generated by the singular points of  $\xi_Z$ . We denote by  $(C^{\bullet}_{\overline{p}, loc}(T, \xi), \partial^{\bullet})$  the quotient complex,

(5.2) 
$$C_{\bar{p},\operatorname{loc}}^{k}(T,\xi) = \sum_{i+j=k} \sum_{x \in \operatorname{Crit}_{i}(\xi_{Y})} \Omega_{\bar{q},tr}^{j}(L_{x}), \text{ for } k = 0,\ldots,n,$$

and call it the local Morse–Bott type cochain complex. The complex  $(C^{\bullet}_{\bar{p},\text{loc}}(T,\xi),\partial^{\bullet})$  is filtered by the index of critical points of  $\xi_Y$ .

In this section, by adapting a construction in smooth Morse and Morse–Bott theory, we construct a perturbation of the local Morse–Bott type cochain complex  $(C^{\bullet}_{\overline{p},\text{loc}}(T,\xi),\partial^{\bullet})$ . We can assume w.l.o.g. that in the tubular neighbourhood T of Y the stratified anti-radial gradient-like vector field  $\xi$  is of the form  $-r\partial_r + \xi_Y$ . The idea of the construction presented in this section, is to produce a perturbation  $\widetilde{\xi}_{\eta}$  of the unfolded vector field  $\widetilde{\xi}$  on  $\widetilde{T}$ , which has isolated singularities only. The perturbed complex is a subcomplex of the Morse–Thom–Smale complex associated to  $\widetilde{\xi}_{\eta}$ , quasi-isomorphic to  $(C^{\bullet}_{\overline{p},\text{loc}}(T,\xi),\partial^{\bullet})$ .

The results of this section will be used in the proof of the Main Theorem 2 (in particular in Sections 6.3.2 and 6.4.2). A further application of the construction in this section, handled in a sequel paper, is to the comparison

between the Morse–Bott type complex  $(C^{\bullet}_{\overline{m}}(X,\xi), \partial^{\bullet})$  and the "singular" Witten complex established in [49].

Section 5 is organised as follows: In Section 5.1 we explain the construction for spaces with isolated singularities, in Section 5.2 we extend the construction to the case dim  $Y \ge 1$ . In Section 5.3 we prove that the perturbed complex and the local Morse–Bott cochain complex  $(C^{\bullet}_{\bar{p},\text{loc}}(T,\xi),\partial^{\bullet})$  are quasi-isomorphic; both complexes compute the intersection cohomology  $IH^{\bullet}_{\bar{p}}(T)$ . Note that the argument given in Section 5.3 can be used to give an alternative proof of the Main Theorem 1 in the case of a pseudomanifold with two strata.

#### 5.1. Perturbation of the local Morse–Bott type complex: Isolated singularities

In this section we explain the construction of the perturbed complex for a space X with isolated singularities. Let L be the smooth compact connected oriented link manifold of dimension dim L=c-1=n-1. Here  $T \simeq L \times [0,\epsilon]/_{(y,0)\sim(w,0)}$  and  $\xi=-r\partial_r$ .

Let  $\eta$  be a smooth gradient-like vector field on L. We define the smooth gradient-like vector field

(5.3) 
$$\widetilde{\xi}_{\eta} := -r\partial_r + \eta$$

on the manifold with boundary  $\widetilde{T}\simeq L\times [-\epsilon,\epsilon]$ . The singularities of the vector field  $\widetilde{\xi}_{\eta}$  on  $L\times [-\epsilon,\epsilon]$  are in one-to-one correspondence with the singularities of the vector field  $\eta$  on L. We orient all unstable manifolds of  $\widetilde{\xi}_{\eta}$ . Using the orientations of  $L\times [-\epsilon,\epsilon]$  and of the unstable manifolds of  $\widetilde{\xi}_{\eta}$ , we get induced orientations on all stable manifolds of  $\widetilde{\xi}_{\eta}$ ; note that the stable manifolds of  $\widetilde{\xi}_{\eta}$  are the unstable manifolds of  $-\widetilde{\xi}_{\eta}$ . We denote by  $(C_{\bullet}(L\times [-\epsilon,\epsilon],-\widetilde{\xi}_{\eta},\mathbb{Z}),\partial_{\bullet})$  the homological Morse–Thom–Smale complex with integer coefficients, associated to the manifold with boundary  $L\times [-\epsilon,\epsilon]$  and the vector field  $-\widetilde{\xi}_{\eta}$ , generated by the unstable manifolds of the singular points of the vector field  $-\widetilde{\xi}_{\eta}$  (see e.g. [67, Section 3.4]); it computes the relative homology  $H_{\bullet}(L\times [-\epsilon,\epsilon],\partial(L\times [-\epsilon,\epsilon]),\mathbb{Z})$ . We denote by  $(C_{\bullet}^{\overline{q},\mathrm{pert}}(cL,-\xi,-\eta,\mathbb{Z}),\partial_{\bullet})\subset (C_{\bullet}(L\times [-\epsilon,\epsilon],-\widetilde{\xi}_{\eta},\mathbb{Z}),\partial_{\bullet})$  the subcomplex defined by

$$(5.4) C_k^{\bar{q},\mathrm{pert}}(cL, -\xi, -\eta, \mathbb{Z}) := \{ \sigma \mid \theta(\sigma) \in IC_k^{\bar{q}}(cL, L, \mathbb{Z}) \}.$$

In the following we tacitly identify  $\sigma \in C_k^{\overline{q}, \mathrm{pert}}(cL, -\xi, -\eta, \mathbb{Z})$  with the cone  $\theta(\sigma)$ .

PROPOSITION 5.1. — The complex  $(C^{\bar{q}, \text{pert}}_{\bullet}(cL, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet})$  computes the relative intersection homology of the cone cL with integer coefficients,

$$(5.5) H_{\bullet}(C_{\bullet}^{\bar{q}, \text{pert}}(cL, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet}) \simeq IH_{\bullet}^{\bar{q}}(cL, L, \mathbb{Z}).$$

Moreover

$$(5.6) H_{\bullet}(C_{\bullet}^{\bar{q},\mathrm{pert}}(cL, -\xi, -\eta, \mathbb{Z}) \otimes_{\mathbb{Z}} \mathbb{R}, \partial_{\bullet}) \simeq IH_{\bullet}^{\bar{q}}(cL, L).$$

Proof. — We denote by  $(C_{\bullet}(L, -\eta, \mathbb{Z}), \partial_{\bullet})$  the homological Morse–Thom–Smale complex with integer coefficients associated to the smooth manifold L and the vector field  $-\eta$ , generated by the unstable manifolds of singular points of  $-\eta$  (see e.g. [42, Section (c)]). Recall, that the complex  $(C_{\bullet}(L, -\eta, \mathbb{Z}), \partial_{\bullet})$  computes the (singular) homology of L,

(5.7) 
$$H_{\bullet}(C_{\bullet}(L, -\eta, \mathbb{Z}), \partial_{\bullet}) \simeq H_{\bullet}(L, \mathbb{Z}).$$

We denote by  $(C^{\bar{q},tr}_{\bullet}(L,-\eta,\mathbb{Z}),\partial_{\bullet}) \subset (C_{\bullet}(L,-\eta,\mathbb{Z}),\partial_{\bullet})$  the subcomplex defined by

(5.8) 
$$C_k^{\bar{q},tr}(L,-\eta,\mathbb{Z}) := \begin{cases} C_k(L,-\eta,\mathbb{Z}) & \text{for } k \geqslant c - q_c, \\ \ker \partial_k & \text{for } k = c - q_c - 1, \\ 0 & \text{else.} \end{cases}$$

There is an isomorphism of chain complexes

$$(5.9) \quad (C^{\bar{q}, tr}_{\bullet}(L, -\eta, \mathbb{Z}), \partial_{\bullet}) \longrightarrow (C^{\bar{q}, pert}_{\bullet+1}(cL, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet+1}), \ \tau \longmapsto c\tau,$$

where for  $\tau \subset L$  we denote by  $c\tau \subset cL$  the cone over  $\tau$ .

From (5.7), (5.8), (5.9) and the local calculation for relative intersection homology (recalled in (2.15)), we have

(5.10) 
$$H_{\bullet}(C_{\bullet}^{\bar{q}, \text{pert}}(cL, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet}) \simeq H_{\bullet}(C_{\bullet-1}^{\bar{q}, tr}(L, -\eta, \mathbb{Z}), \partial_{\bullet-1})$$
$$\simeq IH_{\bullet}^{\bar{q}}(cL, L, \mathbb{Z}).$$

Since  $\mathbb{R}$  is a field, the isomorphism (5.6) follows from (5.5) and the universal coefficient theorem.

Set  $(C_{\overline{p},\text{pert}}^{\bullet}(cL,\xi,\eta),\partial^{\bullet}) := (C_{c-\bullet}^{\overline{q},\text{pert}}(cL,-\xi,-\eta,\mathbb{Z}) \otimes_{\mathbb{Z}} \mathbb{R},(-1)^{\bullet+1}\partial_{c-\bullet}).$ By Poincaré duality for intersection (co-)homology (see [30, Section 3.3]) and Proposition 5.1,

(5.11) 
$$H^{\bullet}(C_{\bar{p},\mathrm{pert}}^{\bullet}(cL,\xi,\eta),\partial^{\bullet}) \simeq IH_{\bar{p}}^{\bullet}(cL).$$

# 5.2. Perturbation of the local Morse–Bott type complex: General case

Let now dim  $Y \geqslant 1$ . We start by constructing a perturbation of the unfolded vector field  $\widetilde{\xi}$  on  $\widetilde{T}$ . We essentially follow the construction in [67, Section 3]: We choose pairwise disjoint, open neighbourhoods  $V_x \subset Y$ ,  $x \in \operatorname{Crit}(\xi_Y)$ . The sets  $\widetilde{U}_x := \theta^{-1}(\pi^{-1}(V_x)) \subset \widetilde{T}$  are pairwise disjoint with  $\widetilde{U}_x \simeq V_x \times L_x \times [-\epsilon, \epsilon]$ . We denote by  $\pi_{L_x} : \widetilde{U}_x \to L_x$  the canonical projection into the second factor. Let  $\lambda_x : Y \to \mathbb{R}_{\geqslant 0}$  be a bump-function with compact support in  $V_x$  which is equal to 1 in a neighbourhood of x. Let  $\eta$  be a gradient-like vector field on L, and denote by  $\eta_x$  the corresponding vector field on  $\theta^{-1}(x) \simeq L_x \simeq L$ . We define a smooth vector field  $\widetilde{\xi}_{\eta}$  on  $\widetilde{T}$  as follows

(5.12) 
$$\widetilde{\xi}_{\eta}(w) := \widetilde{\xi}(w) + \sum_{x \in \text{Crit}(\xi_Y)} \lambda_x(\pi(\theta(w))) \eta_x(\pi_{L_x}(w)), \ w \in \widetilde{T}.$$

We have

therefore flow lines for  $\widetilde{\xi}_{\eta}$  project to flow lines for  $\xi$  (under the folding map  $\theta$ ). The singular points of  $\widetilde{\xi}_{\eta}$  are precisely the singular points of  $\eta$  in the fibres of  $\theta$  lying over the singular points  $x \in \text{Crit}(\xi_Y)$ :

$$(5.14) \quad \operatorname{Crit}(\widetilde{\xi}_{\eta}) = \{ \alpha \in \widetilde{T} \mid \theta(\alpha) = x \in \operatorname{Crit}(\xi_{Y}) \text{ and } \pi_{L_{x}}(\alpha) \in \operatorname{Crit}(\eta_{x}) \}.$$

All singular points of the smooth vector field  $\widetilde{\xi}_{\eta}$  on  $\widetilde{T}$  are of standard form. By genericity of the Morse–Smale condition, after possibly perturbing the pair  $(\widetilde{\xi}, \eta)$  we can achieve that the vector field  $\widetilde{\xi}_{\eta}$  is gradient-like.

We choose orientations of the unstable manifolds of  $\eta_x$ ,  $x \in \text{Crit}(\xi_Y)$ . Together with the orientations on the unstable normal bundles of  $L_x$ ,  $x \in \text{Crit}(\xi_Y)$ , we get orientations on all unstable manifolds of  $\widetilde{\xi}_{\eta}$ . Using the orientations of  $\widetilde{T}$  and of the unstable manifolds, we get induced orientations on all stable manifolds of  $\widetilde{\xi}_{\eta}$  as well.

For a singular point  $\alpha \in \operatorname{Crit}(\widetilde{\xi}_{\eta})$  with  $\theta(\alpha) = x$ , we denote by  $W^u(\alpha, -\widetilde{\xi}_{\eta}) \subset \widetilde{T}$  the unstable manifold of  $\alpha$  w.r.t. the flow induced from  $-\widetilde{\xi}_{\eta}$ . We denote by  $W^u(\alpha, \pm \eta_x) \subset L_x$  the unstable manifold of  $\pi_{L_x}(\alpha)$  w.r.t. the flow induced from  $\pm \eta_x$  on  $L_x$ .

We denote by  $(C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet})$  the homological Morse–Thom–Smale complex with integer coefficients associated to the manifold with boundary  $\widetilde{T}$  and the vector field  $-\widetilde{\xi}_{\eta}$ , generated by the unstable manifolds of  $-\widetilde{\xi}_{\eta}$  on  $\widetilde{T}$  (see e.g. [67, Section 3.4]). The complex  $(C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet})$  computes

the relative homology  $H_{\bullet}(\widetilde{T}, \partial \widetilde{T}, \mathbb{Z})$ . It is filtered by the index of critical points of  $-\xi_Y$ , more precisely

$$(5.15) F_m(C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet}) = \bigoplus_{\operatorname{ind}(\theta(\alpha), -\xi_Y) \leq m} \mathbb{Z} \cdot [W^u(\alpha, -\widetilde{\xi}_{\eta})].$$

Let  $\alpha \in \operatorname{Crit}_k(-\widetilde{\xi}_\eta)$  with  $\theta(\alpha) \in \operatorname{Crit}_m(-\xi_Y)$ , and let  $\beta \in \operatorname{Crit}_{k-1}(-\widetilde{\xi}_\eta)$ . By Morse–Smale transversality for the vector field  $\widetilde{\xi}_\eta$  there is only a finite number of trajectories  $\gamma$  for the flow induced from the vector field  $-\widetilde{\xi}_\eta$ , which start in  $\alpha$  and end in  $\beta$ . Set  $n(\alpha,\beta) := \sum_{\gamma} n_{\gamma}(\alpha,\beta)$ , where  $n_{\gamma}(\alpha,\beta) \in \{\pm 1\}$  is the sign obtained by comparison of orientations (see e.g. [42, Section (c)] for more details). Note that, if  $\operatorname{ind}(\theta(\beta), -\xi_Y) \geq m$ , by (5.13) and Morse–Smale transversality for  $\xi$ , there are no trajectories between  $\alpha$  and  $\beta$  unless  $\theta(\alpha) = \theta(\beta) = x$ . In the latter case, all trajectories between  $\alpha$  and  $\beta$  lie in the fibre  $\theta^{-1}(x)$ . From the above discussion we get the following formula for the boundary operator of the complex  $(C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet})$  (see e.g. [67, Section 3.3.3]):

$$\partial[W^{u}(\alpha, -\widetilde{\xi}_{\eta})] = \sum_{\substack{i \text{ ind}(\beta, -\widetilde{\xi}_{\eta}) = k-1 \\ \text{ ind}(\beta, -\widetilde{\xi}_{\eta}) = k-1}} n(\alpha, \beta)[W^{u}(\beta, -\widetilde{\xi}_{\eta})]$$

$$= \sum_{\substack{i \leq m \text{ ind}(\beta, -\widetilde{\xi}_{\eta}) = k-1 \\ \text{ ind}(\theta(\beta), -\xi_{Y}) = i}} n(\alpha, \beta)[W^{u}(\beta, -\widetilde{\xi}_{\eta})]$$

$$= \sum_{\substack{\text{ ind}(\beta, -\widetilde{\xi}_{\eta}) = k-1 \\ \theta(\beta) = \theta(\alpha)}} n(\alpha, \beta)[W^{u}(\beta, -\widetilde{\xi}_{\eta})]$$

$$+ \sum_{\substack{i < m \text{ ind}(\beta, -\widetilde{\xi}_{\eta}) = k-1 \\ \text{ ind}(\theta(\beta), -\xi_{Y}) = i}} n(\alpha, \beta)[W^{u}(\beta, -\widetilde{\xi}_{\eta})].$$

From (5.16) we get, that the graded complex associated to the filtered chain complex  $(C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet})$  (together with its boundary) can be identified with the sum of the Morse–Thom–Smale complexes "in the fibre"  $(C_{\bullet-m}(L_x \times [-\epsilon, \epsilon], r\partial_r - \eta_x, \mathbb{Z}), \partial_{\bullet-m}), x \in \mathrm{Crit}_m(-\xi_Y).$ 

We denote by  $(C^{\overline{q},\mathrm{pert}}_{\bullet}(T,-\xi,-\eta,\mathbb{Z}),\partial_{\bullet})$  the subcomplex of the complex  $(C_{\bullet}(\widetilde{T},-\widetilde{\xi}_{\eta},\mathbb{Z}),\partial_{\bullet})$  defined by

$$(5.17) \quad C_k^{\overline{q}, \mathrm{pert}}(T, -\xi, -\eta, \mathbb{Z})$$

$$\coloneqq \{ \sigma \in C_k(\widetilde{T}, -\widetilde{\xi}_n, \mathbb{Z}) \mid \theta(\sigma) \in IC_k^{\overline{q}}(T, \partial T, \mathbb{Z}) \}.$$

Note that, for  $\alpha \in \operatorname{Crit}_k(-\widetilde{\xi}_{\eta})$  with  $\theta(\alpha) = x \in \operatorname{Crit}(\xi_Y)$ ,

(5.18) 
$$\theta(W^u(\alpha, -\tilde{\xi}_{\eta}))$$
 is  $(\bar{q}, k)$ -allowable  $\iff \operatorname{ind}(\pi_{L_x}(\alpha), \eta_x) \leqslant q_c$ .

Let  $\sigma = \sum_{\alpha, \text{ind}(\theta(\alpha), -\xi_Y) = m} a_{\alpha}[W^u(\alpha, -\widetilde{\xi}_{\eta})] \in C^{\overline{q}, \text{pert}}_{m+c-q_c}(T, -\xi, -\eta, \mathbb{Z})$ . Denote by  $(\partial \sigma)_m$  the part of  $\partial \sigma$  lying in  $\bigoplus_{\text{ind}(\theta(\beta), -\xi_Y) = m} \mathbb{Z} \cdot [W^u(\beta, -\widetilde{\xi}_{\eta})]$ . Using (5.16) and (5.18) we get that

$$(5.19) Y \cap \theta((\partial \sigma)_m) = \emptyset$$

is equivalent to

(5.20)

for all 
$$x \in \operatorname{Crit}_m(-\xi_Y) : \{x\} \cap \theta \left( \sum_{\alpha, \beta \in \theta^{-1}(x)} a_{\alpha} n(\alpha, \beta) [W^u(\beta, -\widetilde{\xi_{\eta}})] \right) = \emptyset$$

$$\iff \text{ for all } x \in \operatorname{Crit}_m(-\xi_Y) : \partial \left( \sum_{\theta(\alpha) = x} a_{\alpha} [W^u(\alpha, -\eta_x)] \right) = 0.$$

Let us now consider the subcomplex

$$(5.21) (C_{\bullet}^{\overline{q}, \text{pert}}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet}) \subset (C_{\bullet}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}), \partial_{\bullet})$$

with its filtration induced from the filtration (5.15). Using (5.8), (5.9), (5.16), (5.18), (5.19), (5.20), we conclude for the associated graded complex,

$$(5.22) \quad F_m(C_{\bullet}^{\bar{q}, \mathrm{pert}}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet}) / F_{m-1}(C_{\bullet}^{\bar{q}, \mathrm{pert}}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet})$$

$$\simeq \bigoplus_{x \in \mathrm{Crit}_m(-\xi_Y)} C_{\bullet - m}^{\bar{q}, \mathrm{pert}}(cL_x, r\partial_r, -\eta_x, \mathbb{Z});$$

i.e. it is the sum of the complexes "in the fibre" studied in Section 5.1,  $(C^{\bar{q}, \text{pert}}_{\bullet}(cL_x, r\partial_r, -\eta_x, \mathbb{Z}), \partial_{\bullet}).$ 

PROPOSITION 5.2. — The complex  $(C^{\overline{q}, \text{pert}}_{\bullet}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet})$  computes the relative intersection homology of the tubular neighbourhood T with integer coefficients,

$$(5.23) H_{\bullet}(C_{\bullet}^{\bar{q},\mathrm{pert}}(T,-\xi,-\eta,\mathbb{Z}),\partial_{\bullet}) \simeq IH_{\bullet}^{\bar{q}}(T,\partial T,\mathbb{Z}).$$

Moreover

$$(5.24) H_{\bullet}(C_{\bullet}^{\bar{q},\mathrm{pert}}(T,-\xi,-\eta,\mathbb{Z})\otimes_{\mathbb{Z}}\mathbb{R},\partial_{\bullet}) \simeq IH_{\bullet}^{\bar{q}}(T,\partial T).$$

*Proof.* — The idea of the proof is to adapt the Leray–Serre spectral sequence in smooth Morse homology (see e.g. [29, 39, 67]) to the singular situation; here we rely on the version in [67]. The second ingredient in this proof is the Leray–Serre spectral sequence for singular fibrations in

intersection homology due to Friedman (see [26, 27, 28]). Here we use the version in [26, Section 5.8], which we now recall: Let us fix a triangulation K of Y, which is sufficiently fine, such that each simplex is contained in a neighbourhood, where the fibre bundle  $T \to Y$  is trivial. We denote by  $Y^m \subset Y$  the open regular neighbourhood (in the barycentric subdivision of K) of the simplicial m-skeleton of Y. This provides a filtration of the space Y as well as an induced filtration of the space T by  $T^m :=$  $\pi^{-1}(Y^m)$ . The complex of PL intersection chains  $(IC^{\bar{q}}_{\bullet}(T,\partial T,\mathbb{Z}),\partial_{\bullet})$  is filtered by  $F_m(IC^{\bar{q}}_{\bullet}(T,\partial T,\mathbb{Z}),\partial_{\bullet}) := \operatorname{im}(IC^{\bar{q}}_{\bullet}(T^m,\partial T^m,\mathbb{Z}) \to IC^{\bar{q}}_{\bullet}(T,\partial T,\mathbb{Z})).$ The spectral sequence associated to the filtration described above abuts to  $IH^{\bar{q}}_{\bullet}(T,\partial T,\mathbb{Z})$ . We denote by  $I\mathcal{H}^{\bar{q}}_{\bullet}(cL,L,\mathbb{Z})$  the local system on Y with fibre  $I\mathcal{H}^{\bar{q}}_{\bullet}(cL,L,\mathbb{Z})_x = IH^{\bar{q}}_{\bullet}(cL_x,L_x,\mathbb{Z})$ . The  $E^1$ -term of the filtered complex described above can be identified with the cellular complex  $(C_{\bullet}(Y, I\mathcal{H}^{\bar{q}}_{\bullet}(cL, L, \mathbb{Z})), \partial_{\bullet}), E^{1}_{m,k} \simeq C_{m}(Y, I\mathcal{H}^{\bar{q}}_{k}(cL, L, \mathbb{Z})).$  For the  $E^{2}$ -term we get  $E_{m,k}^2 \simeq H_m(Y, I\mathcal{H}_k^{\bar{q}}(cL, L, \mathbb{Z}))$ , where  $H_{\bullet}(Y, I\mathcal{H}_k^{\bar{q}}(cL, L, \mathbb{Z}))$  denotes the singular homology of Y with values in the local system  $I\mathcal{H}_{k}^{\overline{q}}(cL,L,\mathbb{Z})$ .

We denote by  $(C_{\bullet}(Y, -\xi_Y, I\mathcal{H}^{\bar{q}}_{\bullet}(cL, L, \mathbb{Z})), \partial_{\bullet})$  the homological Morse–Thom–Smale complex on Y associated to the vector field  $-\xi_Y$  with values in the local system  $I\mathcal{H}^{\bar{q}}_{\bullet}(cL, L, \mathbb{Z})$  (see e.g. [67, Section 3.2.3], [9, Section I(c)]). Using (5.22) and following the arguments in [67, Sections 3.3 and 3.4] we can prove that the  $E^1$ -term in the spectral sequence associated to the filtered complex  $(C^{\bar{\bullet}, \text{pert}}_{\bullet}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet})$  can be identified with  $(C_{\bullet}(Y, -\xi_Y, I\mathcal{H}^{\bar{q}}_{\bullet}(cL, L, \mathbb{Z})), \partial_{\bullet}), E^1_{m,k} \simeq C_m(Y, -\xi_Y, I\mathcal{H}^{\bar{q}}_k(cL, L, \mathbb{Z}))$ . For the  $E^2$ -term we get again  $E^2_{m,k} \simeq H_m(Y, I\mathcal{H}^{\bar{q}}_k(cL, L, \mathbb{Z}))$ .

We now choose the triangulation K of Y such that in addition it is compatible with the stratification of Y by the unstable manifolds of the gradient-like vector field  $-\xi_Y$ . This is possible by the triangulability of abstract stratified spaces (see [81]) and since, by [42, Proposition 2], the manifold Y with the unstable cell decomposition is an abstract stratified space. We then have a natural map of filtered chain complexes

$$(5.25) (C^{\bar{q}, \mathrm{pert}}_{\bullet}(T, -\xi, -\eta, \mathbb{Z}), \partial_{\bullet}) \longrightarrow (IC^{\bar{q}}_{\bullet}(T, \partial T, \mathbb{Z}), \partial_{\bullet}).$$

From the above discussion, the induced map on the  $E^1$ -terms of the associated spectral sequences is the natural quasi-isomorphism

$$(5.26) \quad (C_{\bullet}(Y, -\xi_Y, I\mathcal{H}_k^{\bar{q}}(cL, L, \mathbb{Z})), \partial_{\bullet}) \longrightarrow (C_{\bullet}(Y, I\mathcal{H}_k^{\bar{q}}(cL, L, \mathbb{Z})), \partial_{\bullet}),$$

and the induced map on the  $E_2$ -term is an isomorphism.

The isomorphism (5.23) now follows using [58, Theorem 3.2]. The isomorphism (5.24) follows from (5.23) and the universal coefficient theorem.  $\square$ 

We denote by

$$(5.27) \quad (C^{\bullet}_{\bar{\nu},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet}) := (C^{\bar{q},\mathrm{pert}}_{n-\bullet}(T,-\xi,-\eta,\mathbb{Z}) \otimes_{\mathbb{Z}} \mathbb{R}, (-1)^{\bullet+1}\partial_{n-\bullet}).$$

By Poincaré duality for intersection (co-)homology (see [30, Section 3.3]) and Proposition 5.2,

(5.28) 
$$H^{\bullet}(C^{\bullet}_{\bar{p}, \mathrm{pert}}(T, \xi, \eta), \partial^{\bullet}) \simeq IH^{\bullet}_{\bar{p}}(T).$$

# 5.3. The cohomology of the local Morse–Bott type cochain complex

The situation is as in the beginning of Section 5. We denote by  $(C^{\bullet}(\widetilde{T},\widetilde{\xi}),\partial^{\bullet})$  the smooth Morse–Bott complex associated to the Morse–Bott vector field  $\widetilde{\xi}$  on  $\widetilde{T}$ ; it is generated by the de Rham complexes  $(\Omega^{\bullet}(L_x),d), x \in \text{Crit}(\xi_Y)$ . By [6, Proposition 3.10] we have a quasi-isomorphism of chain complexes

$$(5.29) F: (C^{\bullet}(\widetilde{T}, \widetilde{\xi}), \partial^{\bullet}) \longrightarrow (C^{\bullet}(\widetilde{T}, \widetilde{\xi}_{\eta}), \partial^{\bullet})$$

defined as follows: For  $x \in \operatorname{Crit}_i(\xi_Y)$  and  $\omega \in \Omega^j(L_x) \subset C^{i,j}(\widetilde{T}, \widetilde{\xi}) \subset C^{i+j}(\widetilde{T}, \widetilde{\xi})$  we define (5.29) by

(5.30) 
$$F(\omega) = \sum_{\substack{\alpha \in \operatorname{Crit}_{i+j}(\widetilde{\xi}_{\eta}) \\ \theta(\alpha) = x}} \left( \int_{\overline{W}^{u}(\alpha, \eta_{x})} \omega \right) [W^{u}(\alpha, \widetilde{\xi}_{\eta})]^{*}.$$

PROPOSITION 5.3. — The map (5.29) restricts to a map, still denoted by F,

$$(5.31) F: (C^{\bullet}_{\bar{p}, \text{loc}}(T, \xi), \partial^{\bullet}) \longrightarrow (C^{\bullet}_{\bar{p}, \text{pert}}(T, \xi, \eta), \partial^{\bullet}).$$

Proof. — Let  $x \in \operatorname{Crit}_i(\xi_Y)$  and  $\omega \in \Omega^j(L_x)$ . We denote by  $F(\omega)^s \in C_{n-(i+j)}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}) \otimes \mathbb{R}$  the element in  $C_{n-(i+j)}(\widetilde{T}, -\widetilde{\xi}_{\eta}, \mathbb{Z}) \otimes \mathbb{R}$  corresponding to  $F(\omega) \in C^{i+j}(\widetilde{T}, \widetilde{\xi}_{\eta})$ .

For  $deg(\omega) = j \leqslant q_c$  we have

$$(5.32) \qquad \dim(\theta(F(\omega)^s) \cap Y) \leqslant (n-c) - i \leqslant n - (i+j) - c + q_c.$$

Hence  $\theta(F(\omega)^s)$  is  $(\bar{q}, n-(i+j))$ -allowable. This shows that, for  $\deg(\omega) < q_c$ ,  $F(\omega) \in C^{\bullet}_{\bar{p},\mathrm{pert}}(T,\xi,\eta)$ .

Let now  $deg(\omega) = j = q_c$  and  $d\omega = 0$  (see (5.1), (5.2)). We have, using the definition of the boundary operator in the various complexes involved,

as well as Stokes' theorem,

$$\partial^{\bullet} F(\omega) = \pm \sum_{\substack{\alpha \in \operatorname{Crit}_{i+j}(\widetilde{\xi}_{\eta}) \\ \theta(\alpha) = x}} \left( \int_{\overline{W^{u}(\alpha, \eta_{x})}} \omega \right) \\ \sum_{\beta \in \operatorname{Crit}_{i+j+1}(\widetilde{\xi}_{\eta})} n(\alpha, \beta) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ = \pm \sum_{\beta, \theta(\beta) = x} \left( \int_{\sum_{\alpha} n(\alpha, \beta)} \overline{W^{u}(\alpha, \eta_{x})} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ \pm \sum_{\alpha, \beta} n(\alpha, \beta) \left( \int_{\overline{W^{u}(\alpha, \eta_{x})}} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ = \pm \sum_{\beta, \theta(\beta) = x} \left( \int_{\partial_{\bullet} \overline{W^{u}(\beta, \eta_{x})}} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ \pm \sum_{\beta, \theta(\beta) \neq x} n(\alpha, \beta) \left( \int_{\overline{W^{u}(\alpha, \eta_{x})}} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ = \pm \sum_{\beta, \theta(\beta) \neq x} n(\alpha, \beta) \left( \int_{\overline{W^{u}(\alpha, \eta_{x})}} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*} \\ = \pm \sum_{\alpha, \beta} n(\alpha, \beta) \left( \int_{\overline{W^{u}(\alpha, \eta_{x})}} \omega \right) [W^{u}(\beta, \widetilde{\xi}_{\eta})]^{*}.$$

Note that for all  $\beta$  in the above sum, by Morse–Smale transversality, we have  $\operatorname{ind}(\theta(\beta), \xi_Y) \geqslant i + 1$ . Therefore from (5.33)

$$(5.34) \dim(\partial \theta(F(\omega)^s) \cap Y) \leqslant (n-c) - (i+1) = n - (i+j+1) - c + q_c.$$

This shows that  $\partial \theta(F(\omega)^s)$  is  $(\bar{q}, n-(i+j+1))$ -allowable, hence  $F(\omega) \in C^{\bullet}_{\bar{p}, \mathrm{pert}}(T, \xi, \eta)$ .

COROLLARY 5.4. — The map of filtered cochain complexes (5.31) induces an isomorphism on the  $E_1$ -terms of the spectral sequences associated to the two filtered complexes. Hence F is a quasi-isomorphism and

$$(5.35) H^{\bullet}(C^{\bullet}_{\bar{p}, loc}(T, \xi), \partial^{\bullet}) \simeq H^{\bullet}(C^{\bullet}_{\bar{p}, pert}(T, \xi, \eta), \partial^{\bullet}) \simeq IH^{\bullet}_{\bar{p}}(T).$$

*Proof.* — The map induced from F on the graded complexes is essentially given by the sum of the "truncated" de Rham maps in smooth Morse theory for the gradient-like vector field  $\eta_x$  on the link  $L_x$ ,  $x \in \text{Crit}(\xi_Y)$ ,

(5.36) 
$$(\Omega_{\overline{q},tr}^{\bullet}(L_x),d) \longrightarrow (C_{\overline{p},\mathrm{pert}}^{\bullet}(cL_x, -r\partial r, \eta_x), \partial^{\bullet})$$

$$\omega \longmapsto \sum_{\alpha \in \mathrm{Crit}(\eta_x)} \left( \int_{\overline{W}^u(\alpha,\eta_x)} \omega \right) [W^u(\alpha,\eta_x)]^*.$$

It is easy to check using smooth Morse theory (see e.g. [9, Theorem 2.9]), that (5.36) is a quasi-isomorphism. Hence the map induced from F on the  $E_1$ -term of the spectral sequences associated to the two filtered complexes is an isomorphism. The isomorphism (5.35) follows from the first claim, (5.11), (5.28) and [58, Theorem 3.2].

# 6. Bismut-Zhang Torsion

The aim of this section is to define the  $Bismut-Zhang\ torsion$  of a compact stratified pseudomanifold for the lower middle perversity  $\overline{m}$ . The definition of the  $Bismut-Zhang\ torsion$  uses the Morse-Bott type cochain complex  $(C^{\bullet}_{\overline{m}}(X,\xi),\partial^{\bullet})$  for the lower middle perversity  $\overline{m}$  defined in Section 4.2. The second ingredient is the model Witten Laplacian appearing in the generalisation of the Witten deformation to stratified spaces in [49]. Conjecturally the  $Bismut-Zhang\ torsion\ will\ serve\ as\ the\ "topological"\ side\ in\ a Cheeger-Müller theorem for singular spaces with iterated conical singularities. Indeed, for Witt spaces with isolated conical singularities, the identity between analytic and <math>Bismut-Zhang\ torsion\ has\ been\ established\ in\ [51].$ 

The Bismut–Zhang torsion for the upper middle perversity  $\bar{n}$  can be defined analogously (see Remarks 6.2 and 6.4(a)). In case of a Witt space (see [77, Definition 2.1]), hence in particular for a space having strata of even codimension only, the Bismut–Zhang torsion for upper and lower middle perversity coincide.

Section 6 is organised as follows: In Section 6.1, for convenience of the reader, we recall some preliminaries on torsion. In Section 6.2 we define the Bismut–Zhang torsion for a stratified pseudomanifold with two strata. The Bismut–Zhang torsion is not a topological invariant in general. The aim of Sections 6.3 and 6.4 is the proof of Main Theorem 2, i.e. we prove that, in case of even-codimensional singular stratum, the Bismut–Zhang torsion is a topological invariant of the pseudomanifold X and moreover is equal to the intersection Reidemeister torsion defined by Aparna Dar [19].

Here, for the clarity of presentation, we give the definition of the Bismut–Zhang torsion only for pseudomanifolds with two strata. However, as mentioned in the introduction, the Bismut–Zhang torsion can be defined in a completely analogous way for stratified pseudomanifolds of arbitrary depth (see Remark 6.4(b)). The results in Sections 6.3 and 6.4 can not be extended beyond the case of two strata at the present: they rely on the Cheeger–Müller theorem for the smooth links.

# 6.1. Preliminaries on Torsion

In this section for convenience of the reader, we recall some basics about torsion: In Section 6.1.1 we recall the definition of the torsion of a finite Hilbert cochain complex, i.e. a cochain complex  $(C^{\bullet}, \partial^{\bullet})$  of finite dimensional Hilbert spaces, such that  $C^i = 0$  for  $|i| \ge N$  for some  $N \in \mathbb{N}$ . We also recall the definition of the Milnor- and Reidemeister torsion of a smooth manifold. Moreover, we recall the concept of a simple structure on an infinite dimensional cochain complex as introduced by Lück, Schick and Thielmann in [47, Section 2]. In Section 6.1.2 we recall results of [47] on the torsion of a filtered complex; they will be used in the definition of the Bismut–Zhang torsion in Section 6.2 and for the proof of its topological invariance in Sections 6.3 and 6.4.

# 6.1.1. The torsion of a cochain complex

For  $f:V\to W$  a linear isomorphism of real finite-dimensional Hilbert spaces, set

(6.1) 
$$[\![f]\!] := \frac{1}{2} \ln(|\det(f^*f)|) \in \mathbb{R}.$$

For an acyclic finite Hilbert cochain complex  $(C^{\bullet}, \partial^{\bullet})$ , define

(6.2) 
$$\rho(C^{\bullet}, \partial^{\bullet}) := [\![(\partial^{\bullet} + \Upsilon^{\bullet}) : C^{\operatorname{ev}} \longrightarrow C^{\operatorname{odd}}]\!] \in \mathbb{R},$$

where  $\Upsilon^{\bullet}$  is a cochain contraction. Definition (6.2) is independent of the choice of the cochain contraction  $\Upsilon^{\bullet}$ . For a cochain homotopy equivalence of finite Hilbert complexes  $f: C^{\bullet} \to D^{\bullet}$ , the cone (Cone $(f)^{\bullet}, \partial^{\bullet}$ ) is the acyclic cochain complex with n-th differential

(6.3) 
$$\begin{pmatrix} \partial_{C^{\bullet}}^{n} & 0 \\ f^{n} & -\partial_{D^{\bullet}}^{n-1} \end{pmatrix} : C^{n} \oplus D^{n-1} \longrightarrow C^{n+1} \oplus D^{n}.$$

Let  $(C^{\bullet}, \partial^{\bullet})$  be a finite Hilbert cochain complex such that its cohomology  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$  carries a Hilbert structure. We can consider  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$  as a cochain complex with trivial differential. There is up to homotopy precisely one cochain map  $i: H^{\bullet}(C^{\bullet}, \partial^{\bullet}) \to (C^{\bullet}, \partial^{\bullet})$  with  $H^{\bullet}(i) = \text{id}$ . Define

(6.4) 
$$\rho(C^{\bullet}, \partial^{\bullet}) := -\rho(\operatorname{Cone}(i)^{\bullet}, \partial^{\bullet}) \in \mathbb{R}.$$

The minus sign in (6.4) ensures that the definitions (6.2) and (6.4) coincide for acyclic complexes. If we fix an orthonormal basis of  $(C^{\bullet}, \partial^{\bullet})$  and of  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$ , then the logarithm of the torsion defined in [59, page 365] is equal to (6.4).

The Milnor and Reidemeister torsion of a smooth compact manifold Y are defined as follows: Let  $(\mathcal{F}, \nabla^{\mathcal{F}}, g^{\mathcal{F}})$  be a flat unimodular vector bundle, i.e.  $\mathcal{F}$  is a flat vector bundle,  $\nabla^{\mathcal{F}}$  is the canonical flat connection and  $g^{\mathcal{F}}$  is a Hermitian metric on  $\mathcal{F}$ , such that the induced metric on the flat line bundle det  $\mathcal{F}$  is flat. Let  $\xi_Y$  be a smooth gradient-like vector field on Y. We denote by  $(C^{\bullet}(Y, \xi_Y, \mathcal{F}), \partial^{\bullet})$  the Morse–Thom–Smale cochain complex (associated to the manifold Y and the vector field  $\xi_Y$ ) with values in the flat vector bundle  $\mathcal{F}$  (see e.g. [9, Section I (c)]). The Hermitian metric  $g^{\mathcal{F}}$  induces a Hilbert structure on the Morse–Thom–Smale complex  $(C^{\bullet}(Y, \xi_Y, \mathcal{F}), \partial^{\bullet})$  (see [9, Section I (d)]). We equip the cohomology  $H^{\bullet}(C^{\bullet}(Y, \xi_Y, \mathcal{F}), \partial^{\bullet}) \simeq H^{\bullet}(Y, \mathcal{F})$  with a metric h. The Milnor torsion of the manifold Y with coefficients in  $\mathcal{F}$  (see [59, Section 8]) is defined as

(6.5) 
$$\rho_M(Y, \mathcal{F}, h) := \rho(C^{\bullet}(Y, \xi_Y, \mathcal{F}), \partial^{\bullet}).$$

The Reidemeister torsion  $\rho_R(Y, \mathcal{F}, h)$  (see [23, 74]) can be defined analogously to the Milnor torsion, using the complex of simplicial cochains w.r.t. a triangulation of Y instead of the Morse–Thom–Smale complex. For a flat unimodular vector bundle  $(\mathcal{F}, \nabla^{\mathcal{F}}, g^{\mathcal{F}})$ , the Reidemeister torsion  $\rho_R(Y, \mathcal{F}, h)$  is a combinatorial invariant of Y (see [59, Theorem 7.1 and Section 8] and [9, Remark 1.10]). Moreover, Milnor and Reidemeister torsion are equal (see [59, Theorem 9.3] and [9, Remark 1.10]). Hence, the Milnor torsion  $\rho_M(Y, \mathcal{F}, h)$  is a combinatorial invariant of Y and in particular does not dependent on the choice of the gradient-like vector field  $\xi_Y$  (see also [9, Theorem 16.1]). In case  $\mathcal{F} = \mathbb{R}$ , i.e.  $\mathcal{F}$  is the trivial vector bundle of rank 1 equipped with the trivial connection and the canonical metric, we write simply  $\rho_M(Y, h)$  (resp.  $\rho_R(Y, h)$ ) for the Milnor torsion (resp. the Reidemeister torsion).

The concept of torsion of a cochain complex has been extended to infinite dimensional cochain complexes by Lück, Schick and Thielmann [47, Section 2] as follows: A simple structure on a cochain complex  $(C^{\bullet}, \partial^{\bullet})$  is an

equivalence class of cochain homotopy equivalences  $u:(\overline{C}^{\bullet},\partial^{\bullet})\to (C^{\bullet},\partial^{\bullet})$  with a finite Hilbert cochain complex as source. Hereby two cochain homotopy equivalences  $u:(\overline{C}^{\bullet},\partial^{\bullet})\to (C^{\bullet},\partial^{\bullet})$  and  $v:(\overline{\overline{C}}^{\bullet},\partial^{\bullet})\to (C^{\bullet},\partial^{\bullet})$  are equivalent if  $\rho(\operatorname{Cone}^{\bullet}(v^{-1}\circ u),\partial^{\bullet})=0$ .

Let  $(C^{\bullet}, \partial^{\bullet})$  be a cochain complex with a simple structure and such that  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$  carries a Hilbert structure. Define

$$\rho(C^{\bullet}, \partial^{\bullet}) := \rho(\overline{C}^{\bullet}, \partial^{\bullet}),$$

for any representative of the simple structure  $u:(\overline{C}^{\bullet},\partial^{\bullet})\to (C^{\bullet},\partial^{\bullet})$ , where we use the Hilbert structure on  $H^{\bullet}(\overline{C}^{\bullet},\partial^{\bullet})$  for which  $H^{\bullet}(u)$  is an isometry.

Let  $0 \to (C^{\bullet}, \partial^{\bullet}) \to (D^{\bullet}, \partial^{\bullet}) \to (E^{\bullet}, \partial^{\bullet}) \to 0$  be an exact sequence of finite Hilbert cochain complexes. Then  $C^n \to D^n \to E^n$ ,  $n \in \mathbb{Z}$ , can be seen as an acyclic finite Hilbert cochain complex and we define

(6.7) 
$$\rho(C^{\bullet} \longrightarrow D^{\bullet} \longrightarrow E^{\bullet}) := \sum_{n \in \mathbb{Z}} (-1)^n \rho(C^n \longrightarrow D^n \longrightarrow E^n).$$

The definition (6.7) can be extended to an exact sequence of cochain complexes with simple structures in the obvious way (see [47, (2.5)]).

# 6.1.2. Torsion and spectral sequences

The study of torsion for filtered complexes is due to Freed [25] and to Lück, Schick and Thielmann [47]; here we follow [47, Section 4]. In Theorem 6.1 below we recall a result of [47, Section 4] on the torsion of a filtered cochain complex, which will be useful in the sequel, more precisely in Proposition 6.8 and Section 6.4.2.

Let  $(C^{\bullet}, \partial^{\bullet})$  be a cochain complex with filtration

$$(6.8) 0 = F_N^{\bullet} \subset \dots \subset F_0^{\bullet} = C^{\bullet}.$$

The associated spectral cohomology sequence  $(E_{\bullet}^{\bullet,\bullet}, d_{\bullet}^{\bullet,\bullet})$  converges to the cohomology  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$ .

Assume that the following data are given

- (a) Simple structures on the graded complexes  $F_m^{\bullet}/F_{m+1}^{\bullet}, m \in \mathbb{N}_0$ ,
- (b) a Hilbert structure on the cohomology  $H^{\bullet}(C^{\bullet}, \partial^{\bullet})$ .

Using the data (a), we can define inductively simple structures on the complexes  $(F_m^{\bullet}, \partial^{\bullet}), m \in \mathbb{N}_0$ , by requiring that

(6.9) 
$$\rho(F_{m+1}^{\bullet} \longrightarrow F_m^{\bullet} \longrightarrow F_m^{\bullet}/F_{m+1}^{\bullet}) = 0,$$

(see [47, Section 4.1]). Hence we get in particular a simple structure on  $(C^{\bullet}, \partial^{\bullet})$  induced from the data (a). Using (a) and (b) we can therefore define  $\rho(C^{\bullet}, \partial^{\bullet})$ .

Set  $K^{m,k} := \operatorname{im}(H^{m+k}(F_m^{\bullet}, \partial^{\bullet}) \to H^{m+k}(C^{\bullet}, \partial^{\bullet}))$ . There is a natural isomorphism

$$(6.10) \Psi^{m,k}: K^{m,k}/K^{m+1,k-1} \longrightarrow E_{\infty}^{m,k}.$$

We equip  $K^{m,k}\subset H^{m+k}(C^{\bullet},\partial^{\bullet})$  with the Hilbert substructure and the quotient  $K^{m,k}/K^{m+1,k-1}$  with the Hilbert quotient structure. The same can be done iteratively for  $E^{m,k}_r$  and  $H^{\bullet}(E^{m+r\bullet,k-(r-1)\bullet}_r)$ .

Theorem 6.1. — In the situation described above, we have

$$(6.11) \quad \rho(C^{\bullet}, \partial^{\bullet}) = \sum_{m} \rho(F_{m}^{\bullet}/F_{m+1}^{\bullet}) + \sum_{k} (-1)^{k} \rho(E_{1}^{\bullet, k}) + \rho^{\geqslant 2}(C^{\bullet}, \partial^{\bullet}),$$

where

(6.12) 
$$\rho^{\geqslant 2}(C^{\bullet}, \partial^{\bullet}) := \sum_{r\geqslant 2} \sum_{m=0}^{r-1} \sum_{k} (-1)^{m+k} \rho(E_r^{m+r\bullet, k-(r-1)\bullet}) - \sum_{m,k} (-1)^{m+k} \llbracket \Psi^{m,k} \rrbracket.$$

# 6.2. Definition of the Bismut-Zhang torsion

# 6.2.1. The model Witten Laplacian

Next to the Morse–Bott type cochain complex constructed in Section 4.2, the second ingredient needed for the definition of the Bismut–Zhang torsion is the model Witten Laplacian, which we recall in this subsection.

The smooth model Witten Laplacian appears as model operator in the Witten deformation of the de Rham complex of a smooth compact manifold using a smooth Morse function (see [38] and [84]).

The generalisation of the "easy" part of Witten's programme, i.e. the analytic proof of the Morse inequalities, to spaces with iterated conical singularities and radial/anti-radial Morse functions has been studied in [5] and [49]; the main idea is to deform the complex of L²-forms (instead of the de Rham complex). In [49] also the "hard" part of Witten's programme, i.e. the comparison between the Witten complex and an appropriate singular Morse–Thom–Smale complex, has been addressed for spaces with isolated conical singularities.

The model Witten Laplacian and its torsion zeta function. The smooth case. We start by recalling the well known model Witten Laplacian  $\Delta_{W,d,m}$  for a smooth Morse critical point of index m on  $\mathbb{R}^d$  (see [84], [9, Proposition 8.2]). We equip the Euclidean space  $\mathbb{R}^d$  with the Morse function

(6.13) 
$$f_{d,m} = \frac{1}{2}(-z_1^2 - \dots - z_m^2 + z_{m+1}^2 + \dots + z_d^2)$$

and the Euclidean metric  $dz_1^2 + \cdots + dz_d^2$ . We denote by  $T^*\mathbb{R}^d$  the cotangent bundle of  $\mathbb{R}^d$ . The Euclidean metric induces an  $L^2$ -metric on sections of  $\Lambda^{\bullet}(T^*\mathbb{R}^d)$ ; we denote by  $L^2(\Lambda^{\bullet}(T^*\mathbb{R}^d))$  the vector space of  $L^2$ -sections of  $\Lambda^{\bullet}(T^*\mathbb{R}^d)$ .

Let  $0 \leqslant l \leqslant d$  and let  $I = (i_1, \ldots, i_l) \in \{1, \ldots, d\}^l$  be a multi-index of length l. Set  $dz_I := dz_{i_1} \wedge \cdots \wedge dz_{i_l}$ . We denote by  $N^+$  resp. by  $N^-$  the number operators defined by

(6.14) 
$$N^+ dz_I := \#\{s \mid m+1 \leqslant i_s \leqslant d\} \cdot dz_I, \quad N^- dz_I := (l-N^+) \cdot dz_I.$$

We denote by  $\Delta_{\mathbb{R}^d} = -\sum_{j=1}^d \frac{\partial^2}{\partial z_j^2}$  the (geometric) Laplacian acting on smooth compactly supported sections of  $\Lambda^{\bullet}(T^*\mathbb{R}^d)$ . For  $z \in \mathbb{R}^d$ , set  $|z| := (z_1^2 + \cdots + z_d^2)^{1/2}$ . The model Witten Laplacian  $\Delta_{W,d,m}$  is defined as the closure in  $L^2(\Lambda^{\bullet}(T^*\mathbb{R}^d))$  of the following operator acting on smooth compactly supported sections of  $\Lambda^{\bullet}(T^*\mathbb{R}^d)$ :

(6.15) 
$$\Delta_{\mathbb{R}^d} + |z|^2 - d + 2(m + N^+ - N^-)$$
$$= \sum_{j=1}^d \left( -\frac{\partial^2}{\partial z_j^2} + z_j^2 \right) - d + 2(m + N^+ - N^-).$$

We denote by  $\Delta_{W,d,m}^{(k)}$  the restriction of  $\Delta_{W,d,m}$  to k-forms,  $k = 0, \ldots, d$ . By [84] (see also [9, Proposition 8.2]), spec( $\Delta_{W,d,m}$ ) =  $2\mathbb{N}_0$  and

(6.16) 
$$\ker(\mathbf{\Delta}_{Wdm}^{(k)}) \simeq H^k(D^m, S^{m-1}), \ k = 0, \dots, d,$$

where  $D^m$  denotes an m-dimensional closed ball and  $S^{m-1}$  denotes its boundary. We explain the isomorphism (6.16) in more detail: Both sides of (6.16) are non-trivial in degree m only. An orthonormal basis of  $\ker(\Delta^{(m)}_{W,d,m})$  w.r.t. the L<sup>2</sup>-metric (induced from the Euclidean metric on  $\mathbb{R}^d$ ) is given by

(6.17) 
$$\alpha^{d,m} := \pi^{-d/4} e^{-|z|^2/2} dz_1 \wedge \dots \wedge dz_m.$$

The isomorphism  $\ker(\mathbf{\Delta}_{W,d,m}^{(m)}) \simeq H^m(D^m,S^{m-1}) \simeq \mathbb{R}$  is given by multiplication with  $e^{f_{d,m}}$  followed by integration (along the unstable manifold

of the critical point 0 w.r.t. the flow induced from the negative gradient vector field  $-\nabla f_{d,m}$ ):

(6.18) 
$$\alpha^{d,m} \longmapsto e^{f_{d,m}} \alpha^{d,m} \longmapsto \int_{\mathbb{R}^m} e^{f_{d,m}} \alpha^{d,m} = \pi^{(2m-d)/4}.$$

We denote by  $\Delta_{W,d,m}^{\perp}$  the restriction of  $\Delta_{W,d,m}$  to the orthogonal complement of  $\ker(\Delta_{W,d,m})$ . We denote by  $N := N^+ + N^-$  the number operator acting on sections of  $\Lambda^{\bullet}(T^*\mathbb{R}^d)$  by multiplication by the form degree. For  $\Re(s) > d/2$ , the zeta function

(6.19) 
$$s \longmapsto \zeta_{d,m}(s) := -\text{Tr}_{\mathbf{s}} \left[ N \left( \mathbf{\Delta}_{W,d,m}^{\perp} \right)^{-s} \right]$$

is a well-defined holomorphic function (see [9], [76, Section 2]). Moreover  $\zeta_{d,m}$  extends to a meromorphic function on the whole complex plane, which is holomorphic at s = 0. We have (see [9], [76, Section 2])

(6.20) 
$$\zeta'_{d,m}(0) = (-1)^m \left( m - \frac{d}{2} \right) \ln \pi.$$

The model Witten Laplacian and its torsion zeta function. The case of isolated singularities. In this paragraph we explain the model Witten Laplacian for an infinite cone cL over a smooth compact connected manifold L of dimension dim L = c - 1. We equip  $(cL)^{sm}$  with the radial Morse function  $f_{cL} = \frac{1}{2}r^2$  and the conical Riemannian metric

$$(6.21) dr^2 + r^2 g^{TL},$$

where  $g^{TL}$  is a Riemannian metric on the smooth link manifold L not depending on r. We denote by  $T^*cL$  the cotangent bundle of  $(cL)^{sm}$  and by  $L^2(\Lambda^{\bullet}(T^*cL))$  the vector space of  $L^2$ -sections of  $\Lambda^{\bullet}(T^*cL)$  w.r.t. the  $L^2$ -metric induced from the Riemannian metric (6.21).

Let  $\Delta_{cL}$  be the Laplacian acting on smooth compactly supported sections of  $\Lambda^{\bullet}(T^*cL)$ . Recall that, unlike in the smooth situation,  $\Delta_{cL}$  admits several closed self adjoint extensions in  $L^2(\Lambda^{\bullet}(T^*cL))$ . We denote by d (resp. by  $\delta$ ) the outer differential (resp. its adjoint w.r.t. the  $L^2$ -metric induced from the conical metric (6.21)) acting on smooth compactly supported sections of  $\Lambda^{\bullet}(T^*cL)$ . Their minimal resp. maximal extensions are denoted by  $d_{\min/\max}$ ,  $\delta_{\min/\max}$ . We denote by N the number operator acting on sections of  $\Lambda^{\bullet}(T^*cL)$  by multiplication by the form degree. The model Witten Laplacian on cL is defined as

(6.22) 
$$\Delta_{W,cL} := \Delta_{cL} - (c - 2N) + r^2,$$

with domain the following subspace of  $L^2(\Lambda^{\bullet}(T^*cL))$  (6.23)

$$\operatorname{dom}(\boldsymbol{\Delta}_{W,cL}) = \left\{ \omega \; \middle| \; \begin{aligned} \omega \in \operatorname{dom}(d_{\max}) \cap \operatorname{dom}(\delta_{\min}), d_{\max}\omega \in \operatorname{dom}\delta_{\min}, \\ \delta_{\min}\omega \in \operatorname{dom}d_{\max} \text{ locally near the cone point} \end{aligned} \right\}$$

(The reader is encouraged to compare the definition of the singular model Witten Laplacian (6.22) with the definition of the smooth model Witten Laplacian (6.15) in case m = 0.)

By [52, Section 4] the model Witten Laplacian has discrete spectrum. We denote by  $\Delta_{W,cL}^{(k)}$  the restriction of  $\Delta_{W,cL}$  to k-forms. We denote by Harm $^{\bullet}(L)$  the vector space of harmonic forms on  $(L, g^{TL})$ . By [49, Theorem 4.2], the local calculation for intersection (co-)homology with lower middle perversity (recalled in (2.14)) and the Hodge theorem for the smooth manifold  $(L, g^{TL})$ ,

(6.24) 
$$\ker(\boldsymbol{\Delta}_{W,cL}^{(k)}) \simeq IH_{\overline{m}}^{k}(cL_{x}) \simeq \begin{cases} H^{k}(L) & \text{for } k < \frac{c}{2}, \\ 0 & \text{else,} \end{cases}$$
$$\simeq \begin{cases} \operatorname{Harm}^{k}(L) & \text{for } k < \frac{c}{2}, \\ 0 & \text{else.} \end{cases}$$

For later use (see the proof of Proposition 6.6), we explain the isomorphism (6.24) in more detail: Let  $\{\beta_i^k\}_i$  be an ON-basis (w.r.t. the  $L^2$ -metric induced from  $g^{TL}$ ) of  $\operatorname{Harm}^k(L)$ . For  $0 \le k < c/2$ , an ON-basis of  $\ker(\Delta_{W,cL}^{(k)})$  w.r.t. the L<sup>2</sup>-metric induced from the conical metric (6.21) is given by

(6.25) 
$$\left\{ \gamma_i^k := \left( \Gamma(c/2 - k)/2 \right)^{-1/2} e^{-r^2/2} \beta_i^k \mid i = 1, \dots, \dim H^k(L) \right\}.$$

For  $0 \le k < c/2$ , the isomorphism (6.24) is given by multiplication with  $e^{f_{cL}}$  followed by evaluation at r = 0,

(6.26) 
$$\gamma_i^k \longmapsto e^{f_{cL}} \gamma_i^k \longmapsto (\Gamma(c/2 - k)/2)^{-1/2} \beta_i^k.$$

We denote by  $\Delta_{W,cL}^{\perp}$  the restriction of  $\Delta_{W,cL}$  to the orthogonal complement of  $\ker(\Delta_{W,cL})$ . By [52, Theorem I], for  $\Re(s) \gg 0$ , the torsion zeta function

$$(6.27) s \longmapsto \zeta_{cL}(s) := -\text{Tr}_{s} \left[ N \left( \Delta_{W,cL}^{\perp} \right)^{-s} \right]$$

is a well-defined holomorphic function, which can be described explicitly in terms of the spectrum of the Laplacian on L. Moreover  $\zeta_{cL}$  extends to a meromorphic function on the whole complex plane, which is holomorphic at s=0.

(In [52] the Witt condition and a spectral Witt condition are assumed, however the cited results also hold without these assumptions.)

Remark 6.2. — To define the Bismut–Zhang torsion for the upper middle perversity  $\bar{n}$  we have to use a different self-adjoint extension  $\Delta'_{W,cL}$  of (6.22), namely the operator with domain the following subset of  $L^2(T^*cL)$ :

$$(6.28) \quad \operatorname{dom}(\boldsymbol{\Delta}'_{W,cL}) \\ = \left\{ \omega \middle| \begin{array}{l} \omega \in \operatorname{dom}(d_{\min}) \cap \operatorname{dom}(\delta_{\max}), d_{\min}\omega \in \operatorname{dom}\delta_{\max}, \\ \delta_{\max}\omega \in \operatorname{dom}d_{\min} \text{ locally near the cone point} \end{array} \right\}.$$

The model Witten Laplacian and its torsion zeta function. The general case. Let L be a smooth compact connected manifold of dimension dim L=c-1. In this paragraph, we explain the model Witten Laplacian for a critical point of index m of a radial Morse function on  $\mathbb{R}^{n-c} \times cL$ , i.e. we equip  $\mathbb{R}^{n-c} \times cL$  with the Morse function

(6.29) 
$$f_{loc}(z,\varphi,r) = \frac{1}{2}(r^2 - z_1^2 - \dots - z_m^2 + z_{m+1}^2 + \dots + z_{n-c}^2)$$
$$= f_{cL}(r) + f_{n-c,m}(z)$$

and with a wedge metric  $dz_1^2 + \cdots + dz_{n-c}^2 + dr^2 + r^2 g^{TL}$ . Note that the negative gradient vector field  $-\nabla f_{\text{loc}}$  of the function  $f_{\text{loc}}$  with respect to the wedge metric is a vector field with a singularity of strong standard form at 0 in the sense of Definition 3.4.

The model Witten Laplacian is the self-adjoint operator

(6.30) 
$$\Delta_W := \operatorname{Id}_{L^2(T^*\mathbb{R}^{n-c})} \otimes \Delta_{W,cL} + \Delta_{W,n-c,m} \otimes \operatorname{Id}_{L^2(T^*cL)},$$
 with domain

(6.31) 
$$\operatorname{dom}(\boldsymbol{\Delta}_{W,n-c,m}) \otimes \operatorname{dom}(\boldsymbol{\Delta}_{W,cL})$$

$$\subset \operatorname{L}^{2}(\Lambda^{\bullet}(T^{*}\mathbb{R}^{n-c})) \otimes \operatorname{L}^{2}(\Lambda^{\bullet}(T^{*}cL))$$

$$= \operatorname{L}^{2}(\Lambda^{\bullet}(T^{*}(\mathbb{R}^{n-c} \times cL))).$$

Again, we denote by  $\Delta_W^{(k)}$  the restriction of the model Witten Laplacian to k-forms. By [49, Section 4.4], the local calculation for intersection (co-)homology (recalled in (2.14)) and the Hodge theorem for the smooth manifold  $(L, g^{TL})$ ,

(6.32) 
$$\ker(\boldsymbol{\Delta}_{W}^{(k+m)}) \simeq IH^{k}(cL) \simeq \begin{cases} \operatorname{Harm}^{k}(L) & \text{for } 0 \leqslant k < \frac{c}{2}, \\ 0 & \text{else.} \end{cases}$$

For later use (see the proof of Proposition 6.8), we explain the isomorphism (6.32) in more detail. Let  $\alpha^{n-c,m} \in \ker(\Delta_{W,n-c,m})$  denote the generator of  $\ker(\Delta_{W,n-c,m})$  defined in (6.17). Let  $\{\beta_i^k\}_i$  (resp.  $\{\gamma_i^k\}_i$ ) be the ON-basis of  $\operatorname{Harm}^k(L)$  (resp. of  $\ker(\Delta_{W,cL}^{(k)})$ ) described in the previous paragraph (see in particular (6.25)). For  $0 \leq k < c/2$ , an ON-basis of  $\ker(\Delta_{W}^{(k+m)})$  is given by

(6.33) 
$$\left\{\omega_i^{k+m} \coloneqq \alpha^{n-c,m} \otimes \gamma_i^k \mid i = 1, \dots, \dim H^k(L)\right\}_i.$$

For  $0 \le k < c/2$ , the isomorphism (6.32) is given by multiplication with  $e^{f_{\text{loc}}}$  followed by integration along the fibres of the unstable endpoint map for  $-\nabla f_{\text{loc}}$ ,

(6.34) 
$$\omega_i^{k+m} \longmapsto e^{f_{\text{loc}}} \omega_i^{k+m} \longmapsto \left( \int_{\mathbb{R}^m} e^{f_{n-c,m}} \alpha^{n-c,m} \right) \left( e^{f_{cL}} \gamma_i^k \right)_{r=0}$$

$$= \pi^{(2m+c-n)/4} \left( \Gamma(c/2-k)/2 \right)^{-1/2} \beta_i^k.$$

We denote by  $\Delta_W^{\perp}$  the restriction of  $\Delta_W$  to the orthogonal complement of  $\ker(\Delta_W)$ . We denote by N the number operator acting on sections of  $\Lambda^{\bullet}(T^*(\mathbb{R}^{n-c} \times cL))$  by multiplication by the form degree. By the results of the previous two paragraphs, for  $\Re(s) >> 0$ , the zeta function

$$(6.35) s \longmapsto \zeta(s) \coloneqq -\text{Tr}_{s} \left[ N \left( \mathbf{\Delta}_{W}^{\perp} \right)^{-s} \right],$$

is a well-defined holomorphic function. Moreover  $\zeta$  extends to a meromorphic function on the whole complex plane, which is holomorphic at s=0. Using (6.16), (6.20), (6.24) and the product formula for the analytic torsion [73, Theorem 2.5], we have

(6.36) 
$$\zeta'(0) = (-1)^m \zeta'_{cL}(0) + (-1)^m I \chi_{\overline{m}}(cL) \left(m - \frac{n-c}{2}\right) \ln \pi,$$

where  $I\chi_{\overline{m}}(cL) := \sum_{k=0}^{c} (-1)^k \dim IH_{\overline{m}}^k(cL)$  denotes the intersection Euler characteristic of cL.

# 6.2.2. Definition of the Bismut–Zhang torsion $\rho_{BZ}(X, \xi, g^{TL}, h_X)$

From now on, let X be an n-dimensional compact pseudomanifold with two strata  $\{Y, Z := X^{sm}\}$ , Y < Z. For simplicity we assume that the singular stratum Y is connected. We denote by L its link and by T its closed tubular neighbourhood. Set  $c := \operatorname{codim}_X Y$ .

Let  $\xi = \{\xi_Y, \xi_Z\}$  be a stratified anti-radial gradient-like vector field on X. For all  $x \in \text{Crit}(\xi_Y)$ , we fix a Riemannian metric  $g^{TL_x}$  on the smooth link manifold  $L_x$ . For  $x \in \text{Crit}(\xi_Y)$ , we denote by  $\Delta_{W,x}$  (resp. by  $\zeta_x$ ) the

model Witten Laplacian (resp. its torsion zeta function) defined in the last paragraph of Subsection 6.2.1. For  $x \in \operatorname{Crit}_m(\xi_Z)$ , we denote by  $\Delta_{W,x}$  (resp. by  $\zeta_x$ ) the smooth model Witten Laplacian for a smooth Morse critical point of index m on  $\mathbb{R}^n$  (resp. its torsion zeta function).

(a) Let  $(C^{\bullet}_{\overline{m}}(X,\xi), \partial^{\bullet})$  be the filtered Morse–Bott type cochain complex associated to the stratified anti-radial gradient-like vector field  $\xi$  and the lower middle perversity (see Definition 4.8) with associated graded complex  $G^{\bullet}_{m}$ ,  $m \in \mathbb{N}_{0}$  (see Section 4.2.2). For  $m \in \mathbb{N}_{0}$ , we define a finite dimensional cochain complex

$$\overline{G}_m^{\bullet} := H^{\bullet}(G_m^{\bullet})$$

$$(6.37) \qquad \simeq \left( \bigoplus_{x \in \operatorname{Crit}_m(\xi_Y)} IH_{\overline{m}}^{\bullet - m}(cL_x) \right) \oplus \left( \bigoplus_{x \in \operatorname{Crit}_m(\xi_Z)} H^{\bullet}(D_x^u, \partial D_x^u) \right),$$

with trivial coboundary operator. We equip the cohomology of the complex  $\overline{G}_m^{\bullet}$  with the Hilbert structure induced from the L<sup>2</sup>-metric on  $\bigoplus_{x \in \operatorname{Crit}_m(\xi)} \ker(\Delta_{W,x})$  via (6.16) resp. the first isomorphism in (6.32). We equip the finite complex  $\overline{G}_m^{\bullet}$  with a Hilbert structure, such that

(6.38) 
$$\rho(\overline{G}_m^{\bullet}) = -\sum_{x \in \operatorname{Crit}_m(\xi)} \frac{1}{2} \zeta_x'(0).$$

We have an up to homotopy unique cochain homotopy equivalence  $u: \overline{G}_m^{\bullet} \to G_m^{\bullet}$  with  $H^{\bullet}(u) = \mathrm{id}$ . This equips the graded complexes  $G_m^{\bullet}$ ,  $m \in \mathbb{N}_0$ , with simple structures.

(b) We equip the cohomology  $H^{\bullet}(C^{\bullet}_{\overline{m}}(X,\xi),\partial^{\bullet}) \simeq IH^{\bullet}_{\overline{m}}(X)$  with a Hilbert structure  $h_X$ .

As explained in Section 6.1.2, from the above data, we get a preferred simple structure  $(\overline{C}^{\bullet}, \overline{\partial}^{\bullet})$  on  $(C^{\bullet}_{\overline{m}}(X, \xi), \partial^{\bullet})$ .

DEFINITION 6.3. — Let  $h_X$  be a Hilbert structure on  $H^{\bullet}(C^{\bullet}_{\overline{m}}(X,\xi),\partial^{\bullet}) \simeq IH^{\bullet}_{\overline{m}}(X)$ . Moreover, let  $(C^{\bullet}_{\overline{m}}(X,\xi),\partial^{\bullet})$  be equipped with the preferred simple structure  $(\overline{C}^{\bullet},\overline{\partial}^{\bullet})$  as explained above. The Bismut–Zhang torsion of X is defined by

(6.39) 
$$\rho_{BZ}(X, g^{TL}, \xi, h_X) := \rho(C_{\overline{m}}^{\bullet}(X, \xi), \partial^{\bullet}) = \rho(\overline{C}^{\bullet}, \overline{\partial}^{\bullet}).$$

Remarks 6.4.

(a) The Bismut–Zhang torsion for the upper middle perversity  $\bar{n}$  is defined analogously, using the complex  $(C_{\bar{n}}^{\bullet}(X,\xi),\partial^{\bullet})$  and the model Witten Laplacian from Remark 6.2. In case of a Witt space, the

Bismut–Zhang torsion for lower middle and upper middle perversity coincide, since the two ingredients used for their definition do coincide.

- (b) Let L be a stratified pseudomanifold equipped with an iterated conical Riemannian metric  $g^{TL}$ . The model Witten Laplacian on cL resp. on  $\mathbb{R}^{n-c} \times cL$  has also been studied in this more general situation [49, Sections 4.3 and 4.4]. One can study the associated zeta function, compute it (in terms of the spectrum of the transverse Laplacian) and prove holomorphicity in 0. In Section 4, the Morse–Bott type complex  $(C^{\bullet}_{\overline{m}}(X,\xi),\partial^{\bullet})$  has been constructed for stratified pseudomanifolds of arbitrary depth. Hence both ingredients in the definition of the Bismut–Zhang torsion are available for stratified pseudomanifolds of arbitrary depth. The above definition of the Bismut–Zhang torsion can be generalised directly to the case of stratified pseudomanifolds of arbitrary depth.
- (c) By definition, the Bismut–Zhang torsion depends on the anti-radial gradient-like vector field  $\xi$  and on the metrics  $g^{TL_x}$ ,  $x \in \text{Crit}(\xi_Y)$ . However, using the homotopy principle in smooth Morse theory as in [9, Section XIV], one can prove that

(6.40) 
$$\rho_{BZ}(X, g^{TL}, \xi, h_X) = \rho_{BZ}(X, g^{TL}, \xi', h_X),$$

for a stratified anti-radial gradient-like vector field  $\xi'$  with  $\xi' = \xi$  on T. Moreover (6.40) also holds, if we perturb  $\xi$  to  $\xi' = -r\partial_r + \xi_Y$  in the tubular neighbourhood T.

- (d) The definition of the Bismut–Zhang torsion bears some similarity to the definition of the smooth Milnor torsion. Using (6.18), (6.20) one can prove that the singular points of  $\xi_Z$  do contribute to  $\rho_{BZ}$  as in the classical definition of the Milnor torsion. However, the contribution of the singular points of  $\xi_Y$  is given by the analytic torsion of the model Witten Laplacian.
- (e) Let  $(X, g^{TX})$  be a singular space with isolated conical singularities, i.e.  $X^{sm}$  carries a Riemannian metric  $g^{TX}$  which, near the singularities, is of the form (6.21). By work of Cheeger [17, Section 1 and Theorem 5.1] and of Cheeger–Goresky–MacPherson [18, Section 3.4], we have the following Hodge-de Rham theorem:

(6.41) 
$$IH_{\overline{m}}^{\bullet}(X) \simeq H_{(2)}^{\bullet}(X) \simeq \operatorname{Harm}_{(2)}^{\bullet}(X),$$

where  $H_{(2)}^{\bullet}(X)$  denotes the L<sup>2</sup>-cohomology of X and  $\operatorname{Harm}_{(2)}^{\bullet}(X)$  denotes the vector space of L<sup>2</sup>-harmonic forms on X. Let  $h_X$  be the metric on  $H_{(2)}^{\bullet}(X)$  induced from the L<sup>2</sup>-metric on  $\operatorname{Harm}_{(2)}^{\bullet}(X)$ 

via the isomorphism (6.41). We denote by  $|\ |_{\det H^{\bullet}_{(2)}(X)}^{RS}$  the Ray–Singer metric induced on the determinant line  $\det H^{\bullet}_{(2)}(X)$  from  $h_X$ . The Bismut–Zhang metric  $|\ |\ |_{\det H^{\bullet}_{(2)}(X)}^{\xi,g}$  defined in [51] relates as follows to the Bismut–Zhang torsion defined here:

(6.42) 
$$\rho^{BZ}(X, g^{TL}, \xi, h_X) = \pm \ln \left( \frac{\| \|_{\det H_{(2)}^{\bullet}(X)}^{\xi, g}}{\| \|_{\det H_{(2)}^{\bullet}(X)}^{RS}} \right).$$

Note that the general assumption in [51] is that X satisfies the Witt and a spectral Witt condition (see [51, Section 2.3]).

# 6.2.3. Gluing formula for the Bismut–Zhang torsion

The complexes  $(C^{\bullet}(Z,\xi_Z),\partial^{\bullet})$  and  $(C^{\bullet}_{\overline{m},\mathrm{loc}}(T,\xi),\partial^{\bullet})$  have been introduced at the beginning of Section 5. Adapting the construction of Section 6.2.2, we can equip the complex  $(C^{\bullet}_{\overline{m},\mathrm{loc}}(T,\xi),\partial^{\bullet})$  with a simple structure. We equip  $H^{\bullet}(C^{\bullet}_{\overline{m},\mathrm{loc}}(T,\xi),\partial^{\bullet}) \simeq IH^{\bullet}_{\overline{m}}(T)$  (see Corollary 5.4) with a Hilbert structure  $h_T$  and define the torsion  $\rho(C^{\bullet}_{\overline{m},\mathrm{loc}}(T,\xi),\partial^{\bullet})$  proceeding as in Section 6.2.2.

We have a short exact sequence of complexes (6.43)

$$0 \longrightarrow (C^{\bullet}(Z, \xi_Z), \partial^{\bullet}) \longrightarrow (C^{\bullet}_{\overline{m}}(X, \xi), \partial^{\bullet}) \longrightarrow (C^{\bullet}_{\overline{m}, loc}(T, \xi), \partial^{\bullet}) \longrightarrow 0.$$

Set  $M := \overline{X \setminus T}$ , which is a smooth manifold with boundary  $\partial M = \partial T$ . By smooth Morse theory, the Main Theorem 1 and Corollary 5.4 the long exact sequence in cohomology induced from (6.43) is isomorphic to

$$(6.44) \quad (\mathcal{L}^{\bullet}, \partial^{\bullet}) : \cdots \longrightarrow H^{\bullet}(M, \partial M) \longrightarrow IH_{\overline{m}}^{\bullet}(X) \longrightarrow IH_{\overline{m}}^{\bullet}(T) \longrightarrow \cdots.$$

We equip  $IH^{\bullet}_{\overline{m}}(X)$  (resp.  $H^{\bullet}(M, \partial M)$ ) with a metric  $h_X$  (resp.  $h_{(M, \partial M)}$ ). We denote by  $\rho(\mathcal{L}^{\bullet}, \partial^{\bullet})$  the torsion of the long exact sequence (6.44).

Proposition 6.5.

(6.45) 
$$\rho_{BZ}(X, g^{TL}, \xi, h_X) = \rho(C^{\bullet}_{\overline{m} \log}(T, \xi), \partial^{\bullet}) + \rho_R(M, \partial M, h_{(M, \partial M)}) + \rho(\mathcal{L}^{\bullet}, \partial^{\bullet}).$$

*Proof.* — The claim follows by applying the gluing formula for torsion to the exact sequence (6.43) and using Remark 6.4(d). The gluing formula for torsion is due to Milnor [59, Theorem 3.2] for finite Hilbert cochain complexes; here we use the extension of Milnor's result to cochain complexes with simple structures in [47, Section 2.6]).

# 6.3. The Bismut–Zhang torsion for pseudomanifolds with singular stratum of even codimension

Topological invariance of the Bismut–Zhang metric for oriented spaces with isolated conical singularities of even dimension has been proved already in [51, Section 8]. In Section 6.3.1 we recall some of the arguments of the proof for convenience of the reader—the arguments in the proof of Proposition 6.6 reappear in the proof of Proposition 6.8. In Section 6.3.2 we give a spectral sequence formula for the Bismut–Zhang torsion, for the case dim  $Y \geqslant 1$ . The results of this section will be used in the proof of the Main Theorem 2 in Section 6.4.2.

# 6.3.1. Vanishing of the Bismut–Zhang torsion for even dimensional spaces with isolated singularities

Let  $(L, g^{TL})$  be a smooth compact connected oriented Riemannian manifold of dimension c-1. From Corollary 5.4, the local calculation for intersection (co-)homology (recalled in (2.14)) and the Hodge theorem for the link manifold  $(L, g^{TL})$ ,

(6.46) 
$$H^{k}(C^{\bullet}_{\overline{m},\text{loc}}(cL, -r\partial_{r}), \partial^{\bullet}) \simeq IH^{k}_{\overline{m}}(cL) \simeq \begin{cases} H^{k}(L) & \text{for } k < c/2, \\ 0 & \text{else,} \end{cases}$$

$$\simeq \begin{cases} \operatorname{Harm}^{k}(L) & \text{for } k < c/2, \\ 0 & \text{else.} \end{cases}$$

We equip  $H^{\bullet}(C^{\bullet}_{\overline{m},\text{loc}}(cL,-r\partial_r),\partial^{\bullet})$  with the metric  $h_{\text{Harm}(L)}$  induced from the L<sup>2</sup>-metric on harmonic forms on  $(L,g^{TL})$  via the isomorphism (6.46).

Proposition 6.6. — Let  $(L, g^{TL})$  be a smooth compact connected oriented odd dimensional Riemannian manifold. Then,

(6.47) 
$$\rho(C_{\overline{m},\text{loc}}^{\bullet}(cL, -r\partial_r), \partial^{\bullet}) = \frac{1}{2}\rho_R(L, h_{\text{Harm}(L)}).$$

*Proof.* — We denote by  $h_{\ker(\Delta_{W,cL})}$  the metric induced on  $IH^{\bullet}_{\overline{m}}(cL)$  from the  $L^2$ -metric on forms on the cone cL (restricted to  $\ker(\Delta_{W,cL})$ ) via the first isomorphism in (6.24). By the definition of  $\rho(C^{\bullet}_{\overline{m},\text{loc}}(cL, -r\partial_r), \partial^{\bullet})$  in Section 6.2.3, we have

$$(6.48) \qquad \rho(C_{\overline{m},\text{loc}}^{\bullet}(cL, -r\partial_r), \partial^{\bullet}) = -\frac{1}{2}\zeta'_{cL}(0) - \sum_{0 \leqslant k < \frac{c}{2}} (-1)^k \llbracket id^k \rrbracket,$$

where we abbreviated

$$(6.49) \quad \llbracket \operatorname{id}^k \rrbracket \coloneqq \llbracket \operatorname{id}^k : (IH^k_{\overline{m}}(cL), h_{\operatorname{Harm}(L)}) \longrightarrow (IH^k_{\overline{m}}(cL), h_{\ker \Delta_{W,cL}}) \rrbracket,$$

Set  $b^k(L) := \dim H^k(L)$ ,  $k = 0, \ldots, c-1$ . Using (6.24), (6.26), for  $0 \le k < c/2$ ,

(6.50) 
$$[id^k] = \frac{1}{2}b^k(L)\ln\left(\frac{\Gamma(c/2-k)}{2}\right).$$

We denote by  $\theta_L(s) = -\text{Tr}_s[N(\Delta_L^{\perp})^{-s}]$  the torsion zeta function of the smooth link manifold L; here  $\Delta_L$  denotes the Laplace–Beltrami operator on the link manifold L. Comparing the explicit computation of the zeta function  $\zeta_{cL}$  in [52, Theorem I] with the computation of the analytic torsion of a truncated cone in [82] and using [36, Theorem 1.1],

(6.51) 
$$\zeta'_{cL}(0) = \frac{1}{2}\theta'_{L}(0) - \sum_{0 \le k \le \frac{c}{2}} (-1)^k b^k(L) \ln\left(\frac{\Gamma(c/2 - k)}{2}\right).$$

By the Cheeger-Müller theorem [16, 60],

(6.52) 
$$\rho_R(L, h_{\text{Harm}(L)}) = -\frac{1}{2}\theta'_L(0).$$

The claim of the proposition follows putting together (6.48)–(6.52).

Let now X be an even dimensional oriented space with isolated singularities, dim X = n = c. An even dimensional space with isolated singularities is a Witt space, hence the intersection cohomology with lower middle and upper middle perversity coincide (see [77, Section 3]) and Poincaré duality holds for  $IH^{\bullet}_{\overline{m}}(X)$  (see [30, Section 3.3]).

COROLLARY 6.7. — Let X be an even dimensional oriented space with isolated singularities,  $\xi$  a stratified anti-radial gradient-like vector field on X and  $\{g^{TL_x}, x \in \Sigma\}$  Riemannian metrics on the link manifolds. We equip  $IH^{\bullet}_{\overline{m}}(X)$  with a Hilbert structure  $h_X$  compatible with Poincaré duality. Then

(6.53) 
$$\rho_{BZ}(X, g^{TL}, \xi, h_X) = 0.$$

*Proof.* — The claim has been proved in [51, Section 8] and relies on Propositions 6.5 and 6.6.

6.3.2. Spectral sequence formula for the Bismut–Zhang torsion in case  $\dim Y \geqslant 1$ ,  $\operatorname{codim}_X Y$  even

In this section we assume that X is a compact oriented pseudomanifold with two strata  $\{Y, Z = X^{sm}\}$  with dim  $Y \ge 1$  and  $c = \operatorname{codim}_X Y$  even.

The assumptions and notations are as in the beginning of Section 6.2.2. Let  $\xi = \{\xi_Y, \xi_Z\}$  be a stratified anti-radial gradient-like vector field on X. We may moreover assume that  $\xi$  is of the form  $\xi = -r\partial_r + \xi_Y$  on T (see Remark 6.4(c)).

We denote by  $\mathcal{H}^k(L)$  (resp. by  $I\mathcal{H}^k_{\overline{m}}(cL)$ ) the flat vector bundle on Y with fibre  $H^k(L_x)$ ,  $x \in Y$  (resp.  $IH^k_{\overline{m}}(cL_x)$ ,  $x \in Y$ ). By [47, Lemma 5.2],  $\otimes_k \det(\mathcal{H}^k(L))^{\otimes (-1)^k}$  is unimodular. Since L is odd dimensional, using Poincaré duality on the link and the local calculation for intersection homology (see (6.24)), we have that  $\otimes_k \det(I\mathcal{H}^\bullet_{\overline{m}}(cL))^{\otimes (-1)^k}$  is unimodular as well. Hence, we can equip the flat vector bundles  $I\mathcal{H}^k(cL)$  (and hence  $\mathcal{H}^k_{\overline{m}}(L)$ ) with Hermitian metrics  $\overline{h}$ , such that the induced metric on  $\otimes_k \det(I\mathcal{H}^k(cL))^{\otimes (-1)^k}$  (resp. on  $\otimes_k \det(\mathcal{H}^\bullet_{\overline{m}}(L))^{\otimes (-1)^k}$ ) is flat.

The Morse–Thom–Smale complex  $(C^{\bullet}(Y, \xi_Y, I\mathcal{H}^k_{\overline{m}}(cL)), \partial^{\bullet})$  on Y associated to the vector field  $\xi_Y$  with values in the flat vector bundle  $I\mathcal{H}^k_{\overline{m}}(cL)$ , carries a metric induced from the metric on  $I\mathcal{H}^k_{\overline{m}}(cL)$ . Moreover we fix a Hilbert structure on  $H^{\bullet}(C^{\bullet}(Y, \xi_Y, I\mathcal{H}^k_{\overline{m}}(cL)), \partial^{\bullet}) \simeq H^{\bullet}(Y, I\mathcal{H}^k_{\overline{m}}(cL))$ . For the  $E_1$ -term resp. the  $E_2$ -term of the spectral sequence associated to the filtered chain complex  $(C^{\bullet}_{\overline{m}, loc}(T, \xi), \partial^{\bullet})$  we have, by the results in Section 5,

(6.54) 
$$U_1^{\bullet,k} : (C^{\bullet}(Y, \xi_Y, I\mathcal{H}_{\overline{m}}^k(cL)), \partial^{\bullet}) \simeq E_1^{\bullet,k}, \\ U_2^{\bullet,k} : H^{\bullet}(Y, I\mathcal{H}_{\overline{m}}^k(cL)) \simeq E_2^{\bullet,k}.$$

Set  $\rho_R(L, \overline{h}) := \rho_R(L, \overline{h}_x)$ , which does not depend on  $x \in Y$ , since the metric induced on  $\otimes_k \det(\mathcal{H}^k(L))^{\otimes (-1)^k}$  from  $\overline{h}$  is flat. We denote by  $\chi(Y)$  the Euler characteristic of the manifold Y.

Proposition 6.8. — Let  $c = \operatorname{codim}_X Y$  be even. Then

$$(6.55) \ \rho(C^{\bullet}_{\overline{m},\text{loc}}(T,\xi),\partial^{\bullet}) = \frac{1}{2}\chi(Y)\rho_{R}(L,\overline{h})$$

$$+ \sum_{k} (-1)^{k}\rho(C^{\bullet}(Y,\xi_{Y},I\mathcal{H}^{k}_{\overline{m}}(cL)),\partial^{\bullet})$$

$$+ \rho^{\geqslant 2}(C^{\bullet}_{\overline{m},\text{loc}}(T,\xi),\partial^{\bullet}) + \sum_{m,k} (-1)^{m+k} \llbracket U_{2}^{m,k} \rrbracket.$$

*Proof.* — In the sequel, we write  $(F_{\text{loc},m}^{\bullet}/F_{\text{loc},m+1}^{\bullet})$ ,  $m \in \mathbb{N}_0$ , for the graded complex associated to the filtered complex  $(C_{\overline{m},\text{loc}}^{\bullet}(T,\xi),\partial^{\bullet})$ . From the definition of  $\rho(C_{\overline{m},\text{loc}}^{\bullet}(T,\xi),\partial^{\bullet})$  in Section 6.2.3 and the transformation

formula [47, Section 2.6] we get

$$(6.56) \quad \rho(F_{\text{loc},m}^{\bullet}/F_{\text{loc},m+1}^{\bullet}) \\ = -\sum_{x \in \text{Crit}_{m}(\xi_{Y})} \left[ \frac{1}{2} \zeta_{x}'(0) + \sum_{0 \leqslant k < c/2} (-1)^{m+k} \llbracket \text{id}_{x}^{k} \rrbracket \right] \\ + \sum_{x \in \text{Crit}_{m}(\xi_{Y})} \left[ -1 \right]^{m+k} \llbracket U_{1}^{m,k} \rrbracket,$$

where we abbreviated

$$(6.57) \qquad \llbracket \operatorname{id}_{x}^{k} \rrbracket := \llbracket \operatorname{id}^{k} : (IH_{\overline{m}}^{k}(cL_{x}), \overline{h}_{x}) \longrightarrow (IH_{\overline{m}}^{k}(cL_{x}), h_{\ker \Delta_{W,x}}) \rrbracket.$$

From the proof of Proposition 6.6 (see in particular (6.51) and (6.52)) and the product formula (6.36),

$$\zeta_{x}'(0) = (-1)^{m} \zeta_{cL_{x}}'(0) + (-1)^{m} I \chi_{\overline{m}}(cL_{x}) \left( m - \frac{\dim Y}{2} \right) \ln \pi$$

$$= -(-1)^{m} \rho_{R}(L_{x}, h_{\operatorname{Harm}(L_{x})})$$

$$-(-1)^{m} \sum_{0 \leqslant k < c/2} (-1)^{k} b^{k}(L_{x}) \ln \left( \frac{\Gamma(c/2 - k)}{2} \right)$$

$$+ (-1)^{m} I \chi_{\overline{m}}(cL_{x}) \left( m - \frac{\dim Y}{2} \right) \ln \pi.$$

Using (6.32), (6.34), for  $x \in \text{Crit}_m(\xi_Y)$ ,  $0 \le k < \frac{c}{2}$ ,

(6.59) 
$$[\operatorname{id}_{x}^{k}] = [\operatorname{id}^{k} : (IH_{\overline{m}}^{k}(cL_{x}), \overline{h}_{x}) \longrightarrow (IH_{\overline{m}}^{k}(cL_{x}), h_{\operatorname{Harm}(L_{x})})]$$
$$+ b^{k}(L_{x}) \left(\frac{1}{2} \ln \left(\frac{\Gamma(c/2 - k)}{2}\right) - \frac{1}{2} \left(m - \frac{\dim Y}{2}\right) \ln \pi\right).$$

From (6.56)–(6.59) and the Poincaré–Hopf theorem for the vector field  $\xi_Y$  on Y,

$$(6.60) \quad \sum_{m} \rho(F_{\text{loc},m}^{\bullet}/F_{\text{loc},m+1}^{\bullet})$$

$$= \sum_{m} (-1)^{m} \sum_{x \in \text{Crit}_{m}(\xi_{Y})} \frac{1}{2} \rho_{R}(L_{x}, \overline{h}_{x}) + \sum_{m,k} (-1)^{m+k} \llbracket U_{1}^{m,k} \rrbracket$$

$$= \frac{1}{2} \chi(Y) \cdot \rho_{R}(L, \overline{h}) + \sum_{m,k} (-1)^{m+k} \llbracket U_{1}^{m,k} \rrbracket.$$

From (6.54) and the transformation formula [47, Section 2.6],

(6.61) 
$$\sum_{k} (-1)^{k} \rho(E_{1}^{\bullet,k}) = \sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi_{Y}, I\mathcal{H}_{\overline{m}}^{k}(cL)), \partial^{\bullet})$$
$$- \sum_{m,k} (-1)^{m+k} \llbracket U_{1}^{m,k} \rrbracket$$
$$+ \sum_{m,k} (-1)^{m+k} \llbracket U_{2}^{m,k} \rrbracket.$$

The claim of the proposition follows applying Theorem 6.1 to the filtered complex  $(C^{\bullet}_{\overline{m},loc}(T,\xi),\partial^{\bullet})$  and using (6.60) and (6.61).

COROLLARY 6.9. — Let  $c = \operatorname{codim}_X Y$  be even. Then the torsion  $\rho(C^{\bullet}_{\overline{m},\operatorname{loc}}(T,\xi),\partial^{\bullet})$  is independent of the choice of the vector field  $\xi_Y$  and the metrics  $g^{TL_x}$ ,  $x \in \operatorname{Crit}(\xi_Y)$ .

Proof. — Let  $\xi_Y$ ,  $\xi_Y'$  be two smooth Morse–Smale vector fields on Y and let  $\xi = -r\partial_r + \xi_Y$ ,  $\xi' = -r\partial_r + \xi_Y'$  be the corresponding stratified anti-radial gradient-like vector fields on T. Denote by  $U_2^{m,k}$  (resp.  $U_2^{lm,k}$ ) the isomorphism in (6.54) for the  $E_2$ -term of the spectral sequence associated to the filtered complex  $(C_{\overline{m},loc}^{\bullet}(T,\xi),\partial^{\bullet})$  (resp.  $(C_{\overline{m},loc}^{\bullet}(T,\xi'),\partial^{\bullet})$ ). Obviously, the first term in the formula (6.55) does not depend on the choice of the vector field  $\xi_Y$  and the choice of metrics  $g^{TL_x}$ ,  $x \in \text{Crit}(\xi_Y)$ . Since the Milnor torsion for a smooth manifold with coefficients in a unimodular flat vector bundle is a combinatorial invariant (see Section 6.1),

(6.62) 
$$\sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi_{Y}, I\mathcal{H}^{k}(cL)), \partial^{\bullet})$$

$$= \sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi'_{Y}, I\mathcal{H}^{k}(cL)), \partial^{\bullet}).$$

From the anomaly formula for the  $\rho^{\geqslant 2}$ -term in the formula for the torsion of a spectral sequence (see [47, Remark 4.3]),

(6.63) 
$$\rho^{\geqslant 2}(C^{\bullet}_{\overline{m},\text{loc}}(T,\xi),\partial^{\bullet}) + \sum_{m,k} (-1)^{m+k} \llbracket U_{2}^{m,k} \rrbracket$$
$$= \rho^{\geqslant 2}(C^{\bullet}_{\overline{m},\text{loc}}(T,\xi'),\partial^{\bullet}) + \sum_{m,k} (-1)^{m+k} \llbracket U_{2}'^{m,k} \rrbracket.$$

The claim follows from (6.55), (6.62) and (6.63).

# 6.4. Comparison between the Bismut–Zhang and the intersection Reidemeister torsion for pseudomanifolds with singular stratum of even codimension

In Section 6.4.1 we recall the definition of intersection Reidemeister torsion given in [19], and show that the perturbed complex constructed in Section 5 computes the intersection Reidemeister torsion of T. We prove Main Theorem 2 in Section 6.4.2.

### 6.4.1. Intersection Reidemeister torsion

Let X be a stratified pseudomanifold as defined in Section 2.2,  $\bar{p}$  a perversity in the sense of Goresky and MacPherson.

Let K be a triangulation of X compatible with the stratification, i.e. the closure of the  $X_k$ 's are subcomplexes of K. We assume moreover, that the triangulation is flaglike with respect to the stratification, i.e. for any simplex  $\sigma$ , the intersection of  $\sigma$  with the closure of an  $X_k$  is a single face of  $\sigma$ . An easy way to achieve a flaglike triangulation is to take the first barycentric subdivision of a given compatible triangulation.

We consider the complex  $(IC^{\bar{p},K}_{\bullet}(X,\mathbb{Z}),\partial_{\bullet})$  of intersection chains of perversity  $\bar{p}$  on X, which are simplicial with respect to K. The complex  $(IC^{\bar{p},K}_{\bullet}(X,\mathbb{Z}),\partial_{\bullet})$  is a complex of finitely generated free abelian groups. Moreover  $H_{\bullet}(IC^{\bar{p},K}_{\bullet}(X,\mathbb{Z}),\partial_{\bullet}) \simeq IH^{\bar{p}}_{\bullet}(X,\mathbb{Z})$  (see [33]), and hence by [34, Theorem 8.1],  $H_{\bullet}(IC^{\bar{p},K}_{\bullet}(X,\mathbb{Z})\otimes_{\mathbb{Z}}\mathbb{R},\partial_{\bullet}\otimes_{\mathbb{Z}}\mathbb{R}) \simeq IH^{\bar{\bullet}}_{\bullet}(X)$ .

Let us denote by  $(\mathcal{D}^{\bullet}_{\overline{p},K}(X), \partial^{\bullet})$  the complex dual to the complex  $(IC^{\overline{p},K}_{\bullet}(X,\mathbb{Z}) \otimes_{\mathbb{Z}} \mathbb{R}, \partial_{\bullet} \otimes_{\mathbb{Z}} \mathbb{R})$ . Then

$$(6.64) H^{\bullet}(\mathcal{D}_{\bar{r},K}^{\bullet}(X), \partial^{\bullet}) \simeq IH_{\bar{r}}^{\bullet}(X).$$

The complex  $(\mathcal{D}_{\bar{p},K}^{\bullet}(X), \partial^{\bullet})$  carries a natural Hilbert structure induced from its integer structure. Let us fix a Hilbert structure  $h_X$  on its cohomology  $H^{\bullet}(\mathcal{D}_{\bar{p},K}^{\bullet}(X), \partial^{\bullet}) \simeq IH_{\bar{p}}^{\bullet}(X)$ . The intersection Reidemeister torsion with perversity  $\bar{p}$  of X is defined as

(6.65) 
$$\rho_{IR}^{\bar{p}}(X, h_X) := \rho(\mathcal{D}_{\bar{p}, K}^{\bullet}(X), \partial^{\bullet}).$$

Definition (6.65) is dual to the definition of A. Dar [19, Definition 2.5], which is formulated in intersection homology.

If the Hilbert structure  $h_X$  on  $H^{\bullet}(\mathcal{D}_{\bar{p},K}^{\bullet}(X), \partial^{\bullet}) \simeq IH_{\bar{p}}^{\bullet}(X)$  is also induced from the integer structure of intersection cohomology, we have by [16, Example 1.3] (see also [19, Example 2.1.9]):

(6.66) 
$$\rho_{IR}^{\bar{p}}(X, h_X) = \sum_{k} (-1)^{k+1} \ln |IH_{\bar{p}}^k(X, \mathbb{Z})_{\text{tors}}|,$$

where  $IH_{\bar{p}}^{\bullet}(X,\mathbb{Z})_{\text{tors}}$  denotes the torsion subgroup of  $IH_{\bar{p}}^{\bullet}(X,\mathbb{Z})$ . Using (6.66) one can show, that the definition (6.65) is independent of the chosen flaglike triangulation.

Let  $(C^{\bullet}_{\overline{p},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet})$  be the perturbed complex introduced in Section 5 with the Hilbert structure induced from its integer structure. Let  $h_T$  be a Hilbert structure on  $H^{\bullet}(C^{\bullet}_{\overline{p},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet}) \simeq IH^{\bullet}_{\overline{p}}(T)$ . Using (5.28), (6.66) (for T instead of X) and applying Cheeger's result in [16, Example 1.3] also to the complex  $(C^{\bullet}_{\overline{p},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet})$ , we conclude

Proposition 6.10. — Under the above assumptions we have

(6.67) 
$$\rho_{IR}^{\bar{p}}(T, h_T) = \rho(C_{\bar{p}, \text{pert}}^{\bullet}(T, \xi, \eta), \partial^{\bullet}).$$

In particular, in case of an isolated singularity,

(6.68) 
$$\rho_{IR}^{\bar{p}}(cL, h_{cL}) = \rho(C_{\bar{p}, pert}^{\bullet}(cL, -r\partial_r, \eta), \partial^{\bullet}).$$

In the following we are only interested in the case where  $\bar{p} = \overline{m}$  is the lower middle perversity.

Let L be a smooth compact connected oriented manifold. Using the isomorphism (6.32), we equip  $IH^{\bullet}_{\overline{m}}(cL)$  with the metric  $h_{\mathrm{Harm}(L)}$  induced from the metric on harmonic forms on the link manifold L. Using the Poincaré duality for the intersection Reidemeister torsion [19, Theorem 2.8] and the combinatorial gluing formula (applied to the exact sequence of intersection chain complexes associated to the pair (cL, L)), one can prove the following

Proposition 6.11. — Let L be a smooth compact connected oriented odd dimensional manifold. Then

(6.69) 
$$\rho_{IR}^{\overline{m}}(cL, h_{\operatorname{Harm}(L)}) = \frac{1}{2}\rho_R(L, h_{\operatorname{Harm}(L)}).$$

#### 6.4.2. Proof of Main Theorem 2

Case 1. — Let X be an even dimensional oriented space with isolated conical singularities. The first claim of Main Theorem 2 follows directly from Corollary 6.7. Let  $h_X$  be a Hilbert structure on  $IH^{\bullet}_{\overline{m}}(X)$  compatible with Poincaré duality. By the duality for the intersection Reidemeister torsion (see [19, Theorem 2.8]) and Corollary 6.7, we get

(6.70) 
$$\rho_{BZ}(X, g^{TL}, \xi, h_X) = \rho_{IR}^{\overline{m}}(X, h_X) = 0.$$

Case 2. — Let now X be a space with two strata, dim  $Y \ge 1$  and  $c = \operatorname{codim}_X Y$  even. The fact, that  $\rho_{BZ}(X, g^{TL}, \xi, h_X)$  is independent on the stratified anti-radial gradient-like vector field  $\xi$  and the Riemannian metrics

 $g^{TL_x}$ ,  $x \in \text{Crit}(\xi_Y)$ , follows from Corollary 6.9, topological invariance of the smooth Milnor torsion, and the gluing formula for the Bismut–Zhang torsion, Proposition 6.5. This shows the first claim of the Main Theorem 2.

Using the gluing formula for torsion [59, Theorem 3.2] one can prove a gluing formula for the intersection Reidemeister torsion. Using the gluing formula for the Bismut–Zhang torsion in Proposition 6.5 as well as Proposition 6.10, the proof of the equality of intersection Reidemeister and Bismut–Zhang torsion is now reduced to the proof of the following proposition:

Proposition 6.12. — In case  $c = \operatorname{codim}_X Y$  even, we have

(6.71) 
$$\rho(C_{\overline{m},\mathrm{pert}}^{\bullet}(T,\xi,\eta),\partial^{\bullet}) = \rho(C_{\overline{m},\mathrm{loc}}^{\bullet}(T,\xi),\partial^{\bullet}).$$

*Proof.* — We proceed similarly to the proof of Proposition 6.8 and establish a spectral sequence formula for  $\rho(C^{\bullet}_{\overline{m},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet})$ . The complex  $(C^{\bullet}_{\overline{m},\mathrm{pert}}(T,\xi,\eta),\partial^{\bullet})$  is filtered by the index of critical points of  $\xi_Y$  in Y. For the associated graded complex we have (compare (5.22)),

(6.72) 
$$G_{\xi,\eta,m}^{\bullet} := F_m(C_{\overline{m},\mathrm{pert}}^{\bullet}(T,\xi,\eta),\partial^{\bullet})/F_{m+1}(C_{\overline{m},\mathrm{pert}}^{\bullet}(T,\xi,\eta),\partial^{\bullet})$$

$$\simeq \bigoplus_{x \in \mathrm{Crit}_m(\xi_Y)} C_{\overline{m},\mathrm{pert}}^{\bullet-m}(cL_x,-r\partial_r,\eta).$$

As in Section 6.3.2 we denote by  $I\mathcal{H}^k_{\overline{m}}(cL)$  the local system on Y with fibre  $IH^k(cL_x)$ ,  $x \in Y$ . We again equip the flat vector bundles  $I\mathcal{H}^k(cL)$  (and hence  $\mathcal{H}^k_{\overline{m}}(L)$ ) with Hermitian metrics  $\overline{h}$ , such that the induced metric on  $\otimes_k \det(I\mathcal{H}^k(cL))^{\otimes (-1)^k}$  (resp. on  $\otimes_k \det(\mathcal{H}^{\bullet}_{\overline{m}}(L))^{\otimes (-1)^k}$ ) is flat; as already explained, this is possible since c is even. The Morse–Thom–Smale complex  $(C^{\bullet}(Y, \xi_Y, I\mathcal{H}^k_{\overline{m}}(cL)), \partial^{\bullet})$  carries the metric induced from  $\overline{h}$ . Moreover we fix a Hilbert structure on  $H^{\bullet}(C^{\bullet}(Y, \xi_Y, I\mathcal{H}^k_{\overline{m}}(cL)), \partial^{\bullet}) \simeq H^{\bullet}(Y, I\mathcal{H}^k_{\overline{m}}(cL))$ . From (6.72) (see also the proof of Proposition 5.2), we deduce for the  $E_1$ - resp. the  $E_2$ -term in the associated spectral sequence, which we denote by  $E_1^{\prime \bullet, k}$  resp.  $E_2^{\prime \bullet, k}$ ,

(6.73) 
$$V_1^{\bullet,k}: (C^{\bullet}(Y,\xi_Y,I\mathcal{H}_{\overline{m}}^k(cL)),\partial^{\bullet}) \simeq E_1^{\prime\bullet,k}, \\ V_2^{\bullet,k}: H^{\bullet}(Y,I\mathcal{H}_{\overline{m}}^k(cL)) \simeq E_2^{\prime\bullet,k}.$$

From Proposition 6.11, (6.68), (6.72) and the transformation formula [47, (2.6)] we get

(6.74) 
$$\sum_{m} \rho(G_{\xi,\eta,m}^{\bullet}) = \sum_{m} (-1)^{m} \sum_{x \in \operatorname{Crit}_{m}(\xi_{Y})} \rho_{IR}^{\overline{m}}(cL_{x}, \overline{h}_{x}) + \sum_{m,k} (-1)^{m+k} \llbracket V_{1}^{m,k} \rrbracket$$

$$= \frac{1}{2} \chi(Y) \cdot \rho_{R}(L, \overline{h}) + \sum_{m,k} (-1)^{m+k} \llbracket V_{1}^{m,k} \rrbracket.$$

From (6.73) and the transformation formula [47, (2.6)] we get

(6.75) 
$$\sum_{k} (-1)^{k} \rho(E_{1}^{\prime \bullet, k}) = \sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi_{Y}, I\mathcal{H}_{\overline{m}}^{k}(cL)), \partial^{\bullet})$$
$$- \sum_{m,k} (-1)^{m+k} \llbracket V_{1}^{m,k} \rrbracket$$
$$+ \sum_{m,k} (-1)^{m+k} \llbracket V_{2}^{m,k} \rrbracket.$$

Applying Theorem 6.1 to the filtered complex  $(C_{\overline{m}, \text{pert}}^{\bullet}(T, \xi, \eta), \partial^{\bullet})$  and using (6.74) and (6.75),

$$(6.76) \quad \rho(C_{\overline{m}, \text{pert}}^{\bullet}(T, \xi, \eta), \partial^{\bullet})$$

$$= \sum_{m} \rho(G_{\xi, \eta, m}^{\bullet}) + \sum_{k} (-1)^{k} \rho(E_{1}^{\prime \bullet, k})$$

$$+ \rho^{\geqslant 2}(C_{\overline{m}, \text{pert}}^{\bullet}(T, \xi), \partial^{\bullet})$$

$$= \frac{1}{2} \chi(Y) \cdot \rho_{R}(L, \overline{h})$$

$$+ \sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi_{Y}, I\mathcal{H}_{\overline{m}}^{k}(cL)), \partial^{\bullet})$$

$$+ \rho^{\geqslant 2}(C_{\overline{m}, \text{pert}}^{\bullet}(T, \xi, \eta), \partial^{\bullet}) + \sum_{m, k} (-1)^{m+k} \llbracket V_{2}^{m, k} \rrbracket.$$

From the anomaly formula for the  $\rho^{\geqslant 2}$ -term (see [47, Remark 4.3]), we have

$$(6.77) \quad \rho^{\geqslant 2}(C^{\bullet}_{\overline{m},\text{loc}}(T,\xi),\partial^{\bullet}) + \sum_{m,k} (-1)^{m+k} \llbracket U_2^{m,k} \rrbracket$$

$$= \rho^{\geqslant 2}(C^{\bullet}_{\overline{m},\text{pert}}(T,\xi),\partial^{\bullet}) + \sum_{m,k} (-1)^{m+k} \llbracket V_2^{m,k} \rrbracket.$$

From Proposition 6.8 and (6.76) and (6.77) we get

(6.78) 
$$\rho(C_{\overline{m}, pert}^{\bullet}(T, \xi, \eta), \partial^{\bullet})$$

$$= \frac{1}{2}\chi(Y) \cdot \rho_{R}(L, \overline{h})$$

$$+ \sum_{k} (-1)^{k} \rho(C^{\bullet}(Y, \xi_{Y}, I\mathcal{H}_{\overline{m}}^{k}(cL)), \partial^{\bullet})$$

$$+ \rho^{\geqslant 2}(C_{\overline{m}, loc}^{\bullet}(T, \xi), \partial^{\bullet}) + \sum_{m, k} (-1)^{m+k} \llbracket U_{2}^{m, k} \rrbracket$$

$$= \rho(C_{\overline{m}, loc}^{\bullet}(T, \xi), \partial^{\bullet}).$$

Remark 6.13. — By combinatorial invariance of the intersection Reidemeister torsion, the second statement of the Main Theorem 2 implies the first. However, here we have also given an independent proof of the first statement.

# BIBLIOGRAPHY

- P. Albin, "On the Hodge theory of stratified spaces", in Hodge theory and L<sup>2</sup>analysis, Advanced Lectures in Mathematics, vol. 39, International Press, 2017,
  p. 1-78.
- [2] P. Albin, F. Rochon & D. Sher, "Analytic torsion and R-torsion of Witt representations on manifolds with cusps", Duke Math. J. 167 (2018), no. 10, p. 1883-1950.
- [3] ——, Resolvent, heat kernel, and torsion under degeneration to fibered cusps, Memoirs of the American Mathematical Society, vol. 1314, American Mathematical Society, 2021.
- [4] ——, "A Cheeger-Müller theorem for manifolds with wedge singularities", Anal. PDE 15 (2022), no. 3, p. 567-642.
- [5] J. A. ÁLVAREZ LÓPEZ & M. CALAZA, "Witten's perturbation on strata", Asian J. Math. 21 (2017), no. 1, p. 47-125.
- [6] D. M. Austin & P. J. Braam, "Morse–Bott theory and equivariant cohomology", in *The Floer memorial volume*, Progress in Mathematics, vol. 133, Birkhäuser, 1995, p. 123-183.
- [7] A. BANYAGA & D. E. HURTUBISE, "Morse-Bott homology", Trans. Am. Math. Soc. 362 (2010), no. 8, p. 3997-4043.
- [8] N. Bergeron & A. Venkatesh, "The asymptotic growth of torsion homology for arithmetic groups", J. Inst. Math. Jussieu 12 (2013), no. 2, p. 391-447.
- [9] J.-M. BISMUT & W. ZHANG, An extension of a theorem by Cheeger and Müller, Astérisque, vol. 205, Société Mathématique de France, 1992, with an appendix by François Laudenbach, 235 pages.
- [10] ——, "Milnor and Ray-Singer metrics on the equivariant determinant of a flat vector bundle", Geom. Funct. Anal. 4 (1994), no. 2, p. 136-212.
- [11] J. P. Brasselet, G. Hector & M. Saralegi, "Théorème de de Rham pour les variétés stratifiées", Ann. Global Anal. Geom. 9 (1991), no. 3, p. 211-243.
- [12] J. BRÜNING & X. MA, "An anomaly formula for Ray-Singer metrics on manifolds with boundary", Geom. Funct. Anal. 16 (2006), no. 4, p. 767-837.

- [13] ——, "On the gluing formula for the analytic torsion", Math. Z. 273 (2013), no. 3-4, p. 1085-1117.
- [14] J.-L. BRYLINSKI, "Equivariant intersection cohomology", in Kazhdan-Lusztig theory and related topics. Proceedings of an AMS special session, held May 19-20, 1989 at the University of Chicago, Lake Shore Campus, Chicago, IL, USA, Contemporary Mathematics, vol. 139, American Mathematical Society, 1992, p. 5-32.
- [15] F. CALEGARI & A. VENKATESH, A torsion Jacquet-Langlands correspondence, Astérisque, vol. 409, Société Mathématique de France, 2019, x+226 pages.
- [16] J. CHEEGER, "Analytic torsion and the heat equation", Ann. Math. 109 (1979), no. 2, p. 259-322.
- [17] ——, "On the Hodge theory of Riemannian pseudomanifolds", in Geometry of the Laplace operator (Proc. Sympos. Pure Math., Univ. Hawaii, Honolulu, Hawaii, 1979), Proceedings of Symposia in Pure Mathematics, vol. XXXVI, American Mathematical Society, 1980, p. 91-146.
- [18] J. CHEEGER, M. GORESKY & R. MACPHERSON, "L<sup>2</sup>-cohomology and intersection homology of singular algebraic varieties", in Seminar on Differential Geometry, Annals of Mathematics Studies, vol. 102, Princeton University Press, 1982, p. 303-340.
- [19] A. DAR, "Intersection R-torsion and analytic torsion for pseudomanifolds", Math. Z. 194 (1987), no. 2, p. 193-216.
- [20] C. Debord & J.-M. Lescure, "K-duality for stratified pseudomanifolds", Geom. Topol. 13 (2009), no. 1, p. 49-86.
- [21] C. Debord, J.-M. Lescure & F. Rochon, "Opérateurs pseudodifférentiels sur les variétés à coins fibrés", Ann. Inst. Fourier 65 (2015), no. 4, p. 1799-1880.
- [22] A. Floer, "Witten's complex and infinite-dimensional Morse theory", J. Differ. Geom. 30 (1989), no. 1, p. 207-221.
- [23] W. FRANZ, "Über die Torsion einer Überdeckung", J. Reine Angew. Math. 173 (1935), p. 245-254.
- [24] U. FRAUENFELDER, "The Arnold-Givental conjecture and moment Floer homology", Int. Math. Res. Not. (2004), no. 42, p. 2179-2269.
- [25] D. S. Freed, "Reidemeister torsion, spectral sequences, and Brieskorn spheres", J. Reine Angew. Math. 429 (1992), p. 75-89.
- [26] G. FRIEDMAN, "Polynomial invariants of non-locally flat knots", PhD Thesis, New York University, 2001.
- [27] ——, "Stratified fibrations and the intersection homology of the regular neighborhoods of bottom strata", Topology Appl. 134 (2003), no. 2, p. 69-109.
- [28] —, "Intersection homology of stratified fibrations and neighborhoods", Adv. Math. 215 (2007), no. 1, p. 24-65.
- [29] K. Fukaya, "Floer homology of connected sum of homology 3-spheres", Topology 35 (1996), no. 1, p. 89-136.
- [30] M. GORESKY & R. MACPHERSON, "Intersection homology theory", Topology 19 (1980), no. 2, p. 135-162.
- [31] ——, "Intersection homology. II", Invent. Math. 72 (1983), no. 1, p. 77-129.
- [32] —, "Stratified Morse theory", in Singularities, Part 1 (Arcata, Calif., 1981), Proceedings of Symposia in Pure Mathematics, vol. 40, American Mathematical Society, 1983, p. 517-533.
- [33] —, "Simplicial Intersection Homology", Appendix to R. MacPherson & K. Vilonen, "Elementary construction of perverse sheaves", *Invent. Math.* 84, p. 403-435, 1986.
- [34] M. GORESKY & P. SIEGEL, "Linking pairings on singular spaces", Comment. Math. Helv. 58 (1983), p. 96-110.

- [35] L. HARTMANN & M. SPREAFICO, "The analytic torsion of a cone over a sphere", J. Math. Pures Appl. 93 (2010), no. 4, p. 408-435.
- [36] ——, "The analytic torsion of a cone over an odd dimensional manifold", J. Geom. Phys. 61 (2011), no. 3, p. 624-657.
- [37] ——, "Intersection torsion and analytic torsion of spaces with conical singularities", https://arxiv.org/abs/2001.07801, 2020.
- [38] B. Helffer & J. Sjöstrand, "Puits multiples en mécanique semi-classique. IV. Étude du complexe de Witten", Commun. Partial Differ. Equations 10 (1985), no. 3, p. 245-340.
- [39] M. HUTCHINGS, "Floer homology of families. I", Algebr. Geom. Topol. 8 (2008), no. 1, p. 435-492.
- [40] F. KIRWAN & J. WOOLF, An introduction to intersection homology theory, second ed., Chapman & Hall/CRC, 2006, xiv+229 pages.
- [41] J. LATSCHEV, "Gradient flows of Morse-Bott functions", Math. Ann. 318 (2000), no. 4, p. 731-759.
- [42] F. LAUDENBACH, "On the Thom-Smale complex", Appendix to [9], 1992.
- [43] M. LESCH, "Determinate of regular singular Sturm-Liouville operators", Math. Nachr. 194 (1998), p. 139-170.
- [44] ——, "A gluing formula for the analytic torsion on singular spaces", Anal. PDE **6** (2013), no. 1, p. 221-256.
- [45] J. LOTT & M. ROTHENBERG, "Analytic torsion for group actions", J. Differ. Geom. 34 (1991), no. 2, p. 431-481.
- [46] W. LÜCK, "Analytic and topological torsion for manifolds with boundary and symmetry", J. Differ. Geom. 37 (1993), no. 2, p. 263-322.
- [47] W. LÜCK, T. SCHICK & T. THIELMANN, "Torsion and fibrations", J. Reine Angew. Math. 498 (1998), p. 1-33.
- [48] U. Ludwig, "Morse-Smale-Witten complex for gradient-like vector fields on stratified spaces", in Singularity theory, World Scientific, 2007, p. 683-713.
- [49] ——, "Comparison between two complexes on a singular space", J. Reine Angew. Math. 724 (2017), p. 1-52.
- [50] —, "A complex in Morse theory computing intersection homology", Ann. Inst. Fourier 67 (2017), no. 1, p. 197-236.
- [51] ——, "An extension of a theorem by Cheeger and Müller to spaces with isolated conical singularities", Duke Math. J. 169 (2020), no. 13, p. 2501-2570.
- [52] ——, "An index formula for the intersection Euler characteristic of an infinite cone", Math. Z. 296 (2020), no. 1-2, p. 99-126.
- [53] J. N. MATHER, Notes on Topological Stability, Mimeographed Notes, Harvard, 1970.
- [54] ——, "Notes on topological stability", Bull. Am. Math. Soc. 49 (2012), no. 4, p. 475-506.
- [55] J. MATZ & W. MÜLLER, "Analytic torsion of arithmetic quotients of the symmetric space SL(n, ℝ)/SO(n)", Geom. Funct. Anal. 27 (2017), no. 6, p. 1378-1449.
- [56] ——, "Approximation of  $L^2$ -analytic torsion for arithmetic quotients of the symmetric space  $SL(n, \mathbb{R})/SO(n)$ ", J. Inst. Math. Jussieu 19 (2020), no. 2, p. 307-350.
- [57] R. MAZZEO & B. VERTMAN, "Analytic torsion on manifolds with edges", Adv. Math. 231 (2012), no. 2, p. 1000-1040.
- [58] J. MCCLEARY, User's guide to spectral sequences, Mathematics Lecture Series, vol. 12, Publish or Perish Inc., 1985, xiv+423 pages.
- [59] J. W. MILNOR, "Whitehead torsion", Bull. Am. Math. Soc. 72 (1966), p. 358-426.
- [60] W. MÜLLER, "Analytic torsion and R-torsion of Riemannian manifolds", Adv. Math. 28 (1978), no. 3, p. 233-305.

- [61] ——, "Analytic torsion and R-torsion for unimodular representations", J. Am. Math. Soc. 6 (1993), no. 3, p. 721-753.
- [62] W. MÜLLER & J. PFAFF, "Analytic torsion of complete hyperbolic manifolds of finite volume", J. Funct. Anal. 263 (2012), no. 9, p. 2615-2675.
- [63] ——, "The analytic torsion and its asymptotic behaviour for sequences of hyperbolic manifolds of finite volume", J. Funct. Anal. 267 (2014), no. 8, p. 2731-2786.
- [64] W. MÜLLER & F. ROCHON, "Analytic torsion and Reidemeister torsion of hyperbolic manifolds with cusps", Geom. Funct. Anal. 30 (2020), no. 3, p. 910-954.
- [65] ——, "Exponential growth of torsion in the cohomology of arithmetic hyperbolic manifolds", Math. Z. 298 (2021), no. 1-2, p. 79-106.
- [66] W. MÜLLER & B. VERTMAN, "The metric anomaly of analytic torsion on manifolds with conical singularities", Commun. Partial Differ. Equations 39 (2014), no. 1, p. 146-191.
- [67] A. OANCEA, "La suite spectrale de Leray-Serre en homologie de Floer des variétés symplectiques compactes à bord de type contact", PhD Thesis, Université Paris Sud, 2003.
- [68] J. Pfaff, "Analytic torsion versus Reidemeister torsion on hyperbolic 3-manifolds with cusps", Math. Z. 277 (2014), no. 3-4, p. 953-974.
- [69] ——, "Exponential growth of homological torsion for towers of congruence subgroups of Bianchi groups", Ann. Global Anal. Geom. 45 (2014), no. 4, p. 267-285.
- [70] ——, "A gluing formula for the analytic torsion on hyperbolic manifolds with cusps", J. Inst. Math. Jussieu 16 (2017), no. 4, p. 673-743.
- [71] A. DU PLESSIS & T. WALL, The geometry of topological stability, London Mathematical Society Monographs. New Series, vol. 9, Clarendon Press, 1995, Oxford Science Publications.
- [72] J. RAIMBAULT, "Asymptotics of analytic torsion for hyperbolic three-manifolds", Comment. Math. Helv. 94 (2019), no. 3, p. 459-531.
- [73] D. B. RAY & I. M. SINGER, "R-torsion and the Laplacian on Riemannian manifolds", Adv. Math. 7 (1971), p. 145-210.
- [74] K. REIDEMEISTER, "Homotopieringe und Linsenräume", Abh. Math. Semin. Univ. Hamb. 11 (1935), no. 1, p. 102-109.
- [75] M.-H. Schwartz, "Classes caractéristiques definies par une stratification d'une variété analytique complexe", C. R. Acad. Sci. Paris 260 (1965), p. 3262-3264, 3535-3537.
- [76] S. Shen, "Laplacien hypoelliptique, torsion analytique, et théorème de Cheeger-Müller", J. Funct. Anal. 270 (2016), no. 8, p. 2817-2999.
- [77] P. Siegel, "Witt spaces: A geometric cycle theory for KO-homology at odd primes", Am. J. Math. 105 (1983), p. 1067-1105.
- [78] S. SMALE, "On gradient dynamical systems", Ann. Math. 74 (1961), p. 199-206.
- [79] R. THOM, "Sur une partition en cellules associée à une fonction sur une variété", C. R. Acad. Sci. Paris 228 (1949), p. 973-975.
- [80] —, "Ensembles et morphismes stratifiés", Bull. Am. Math. Soc. 75 (1969), p. 240-284.
- [81] A. Verona, Stratified mappings structure and triangulability, Lecture Notes in Mathematics, vol. 1102, Springer, 1984, ix+160 pages.
- [82] B. VERTMAN, "Analytic torsion of a bounded generalized cone", Commun. Math. Phys. 290 (2009), no. 3, p. 813-860.
- [83] S. M. Vishik, "Generalized Ray-Singer conjecture. I. A manifold with a smooth boundary", Commun. Math. Phys. 167 (1995), no. 1, p. 1-102.
- [84] E. WITTEN, "Supersymmetry and Morse theory", J. Differ. Geom. 17 (1982), no. 4, p. 661-692.

Manuscrit reçu le 29 septembre 2020, révisé le 22 septembre 2022, accepté le 28 novembre 2022.

Ursula LUDWIG University of Münster, Mathematics Münster, Einsteinstrasse 62, 48149 Münster (Germany) ursula.ludwig@uni-muenster.de